

Serving a Piece of National Nostalgia				
Female Foreign Workers in the Multiculturazing Hospitality Sector in Japan				
MDP in East Asian Studies/Faculty of Social Sciences				
Master's thesis				
Author(s):				
Pauliina Peippo				
26.5.2023 Turku				
T WING				

The originality of this thesis has been checked in accordance with the University of Turku quality

assurance system using the Turnitin Originality Check service.

Master's thesis

Subject: MDP in East Asian Studies

Author(s): Pauliina Peippo

Title: Serving a Piece of National Nostalgia: Female Foreign Workers in the Multiculturazing

Hospitality Sector in Japan

Supervisor(s): Dr. Yoko Demelius **Number of pages**: 112 pages

Date: 26.5.2023

This thesis will examine Japan's national nostalgia and how its political narrative manifests in the country's multiculturalizing hospitality industry through the experiences of foreign female workers. I apply the theoretical framework of "national nostalgia" and the inductive method of thematic analysis to the primary data collected from semi-structured interviews to analyze the data generated from six women who have worked in Japan's hospitality industry. I put particular emphasis on the concepts of "uniqueness," "hospitality," and "care work"—all of which symbolize Japan's hospitality concept, omotenashi, characterized by its nationalist self-image. Due to the lack of scholarship on foreign female workers in the hospitality sector and how the nationalistic discourse of Nihonjinron—theories of Japaneseness—affects the integration of foreigners, this study aims to find answers to how the national nostalgia imbued narrative of "cultural homogeneity and uniquity" influences the perceptions of "Japaneseness" and how this correlates with the obstacles to building a multicultural society. Furthermore, as the service-intensive hospitality concept is charged with the gendered notion of care work, I shall explore how the female gender role manifests in the hospitality sector and how social roles influence foreign female workers in Japan. This study's starting point originates in the researcher's own internship-turned-fieldwork experience of working at a traditional Japanese inn (ryokan) that belongs to the female-dominated service sector. Through the comparison of my own field experience and interview data, the research findings suggest that the national-nostalgia narratives employed by the country's leading elite construct an extremely gendered working domain that affects foreign female workers via the enhanced female-male gender binary. In addition, this national nostalgia-based narrative also serves as a significant obstacle hindering Japan's aspirations to transform into multicultural society (tabunka kyōsei shakai) which contradicts with the nation's growing need of foreign labor.

Key words: Japan, national nostalgia, foreign workers, women, the hospitality industry

Table of contents

1	Intr	oduction—"I felt nostalgic towards Japan"	5
	1.1	Research Questions	7
,	1.2	Thesis Structure	8
•	1.3	"National Nostalgia" as a Theoretical Framework: Yearning for the Past tl	nat
ı	Never	Was	9
2	Вас	kground on Multiculturalism in Japan—"I Didn't Know There Are So)
Ma	any F	oreigners in Japan"	12
2	2.1	Theory and Meaning of <i>Tabunka Kyōsei</i>	12
2	2.2	The Beginning of Multiculturalism in Japan	14
2	2.3	Criticism of <i>Tabunka Kyōsei</i>	16
2	2.4	Reasons for the Increase in Foreign Population in Japan	19
3	Lite	rature Review on the Gendered Labor in Japan—"Not a Service	
Inc	dustr	y Problem"	22
;	3.1	Women as Embodiment of Hospitality and Care	23
;	3.2	The "Corporate Warrior": Japan and the Breadwinner Model	25
;	3.3	Homemaker, Valued Employee, or Both? Women's Double Burden and th	е
(Gend	ered Labor	27
;	3.4	The Research Gap	30
4	Met	hodology—Research Approach, Design, and Primary Data	32
4	4.1	Methodological Concerns	32
	4.1.1	Defining "the Field": A Traditional Inn and a Gendered Domain	32
	4.1.2	The Original Plan and Adjustments	34
	4.1.3	The Researcher's Own Experiences of Working at a Traditional Ryokan	37
4	4.2	The Technicalities: Conducting Research in the Hospitality Industry	39
	4.2.	Accessing the Field and Study Concerns	39
	4.2.2	The Data Collection Method	41
	4.2.3	The Data Coding and Analysis	44
4	4.3	Methodological Discussion: The Researcher's Positionality	45
5	An	Analysis of "Unique" Japaneseness—Through National-Nostalgia-	
Tiı	nted	Glasses	50

5.1	Japan's Soft Power and "Cool Japan"	50	
5.2	Nihonjinron and "Unique" Society	55	
5.3	Obstacles for the Realization of <i>Tabunka Kyōsei</i>	62	
6 An	Analysis of the Gendered Hospitality Industry—"Ryokan Is the		
Embo	diment of Japanese Culture"	69	
6.1	Omotenashi as the embodiment of Japanese culture	69	
6.2	Service Sector as a Gendered Domain	76	
7 Cc	nclusion—"Women and Men Do Not Have the Same Expectations"	86	
References			
Appendices			
Appendix 1 Consent Form for Interview and Research Cooperation			
Appendix 2 Interview Questions			

1 Introduction—"I felt nostalgic towards Japan"

In this study, I will examine Japan's national nostalgia and multiculturalizing hospitality industry through the experiences of foreign female workers. Due to the ongoing globalization, commodities and people move faster and freer than ever before. This in turn has enhanced the immigration from country to country, and Japan is on its way step by step transforming into a more multicultural society. This change is prompted and speeded by the enormous demographic changes and challenges; Japan is becoming a super-aged society with around 28% of its population being 65 or over and projecting the numbers to reach 40% by 2055, and the vast and rapidly growing amount of elderly population who also live longer will become a burden on the pension and medical care systems (Chiavacci and Hommerich 2017, 20; Kang 2017; Kingston 2019), while at the same time Japanese women give birth to fewer children with an average fertility rate of 1.3 (Brinton and Nagase 2017, 340; Kang 2017; Graburn, Ertl and Tierney 2008). In response to these demographic realities, Sakanaka Hidenori, the former head of the Tokyo Immigration Bureau, states that Japan has two options: to either maintain its current economic capacity and power via foreign labor or to keep tight control over immigration (Willis and Murphy-Shigetmatsu 2008, 312).

Traditionally the Japanese immigration policies have been based on the remark that Japan has never been an "immigration country" (Chung 2010; Flowers 2012; Shiobara 2020). However, even though the Japanese government was against importing labor, Japan's labor shortage worsened so much in the early 1980s that the national government had to reluctantly yield and reconsider their immigration strategies as an important agenda (Graburn, Ertl and Tierney 2008; Haines, Yamanaka and Yamashita 2012, 93; Liu-Farrer 2011). The term "immigrant" (移民 *imin*) is rarely used to describe permanent residents, naturalized citizens, or the long-term residents in the context of immigration policy, in statements of political elite or even in mundane conversations, thus being absent in both policy and practise (Flowers 2012). Instead, in The Immigration Control and Refugee Recognition Act (ICRRA) policies refer to "foreign nationals" (外国人 *gaikokujin*), and their visa status deems their period of stay and, in extension, what kind of jobs they can acquire (Flowers 2012).

Despite national government's negative perception towards immigration, there already resides 2.7 million foreign residents in Japan (Shiobara, Kawabata and Matthews 2020, 118; Statistics Bureau of Japan 2021b). However, while the influx of foreign residents has the potential to transform homogenous societies into more multicultural ones, transnational

migration can also reinforce the ethnic identities and nationalist debates (Tsuda 2003 in Graburn, Ertl and Tierney 2008). Indeed, it seems that Japan's increasing multiculturalization has given rise to debates over Japanese identity and even the *tabunka kyōsei*—theory advocating Japanese citizens to live harmoniously with many cultures—related policies have enhanced the ethnic borders between the "ethnic Japanese" and "foreign other" (e.g., Burgess 2012; Lie 2001; Willis and Murphy-Shigematsu 2008; Shiobara 2020).

At the same time while Japan has been reluctant to accept a foreign workforce and to counter its demographic crisis, and increasing the previously low percentage of women in the labor force has been seen as a viable option in the eyes of the Japanese national government (Kang 2017; Paillard-Borg and Holmgren 2016; Steinberg and Nakane 2012, 1). Japan has succeeded in rising women's labor force participation rate (Kingston 2019; Miura 2012, 170), but this positive trend is contradicting the reality of Japan as a country built upon conservative gender roles and with the deep-rooted binary of masculine and feminine domains (e.g., Chapman 2020; Dasgupta 2015; Marshall 2017). In principle, women's working opportunities are more diverse than ever in Japan's history, but the underlying gender role expectations and the society which is built on women as the cornerstone of the welfare system nudges women to be employed in certain female-dominated fields with low salary and precarious work without strong job security (Miura 2012; Ogasawara 2016; Ueno 2021; See more in Chapter 3). Japanese gender gap is huge as female workers are underrepresented in many fields and unable to rise to top positions (Hara 2022; 36; Nemoto 2016; Yamaguchi 2019). In addition, Japan's is in the 116th place the World's Economic Forum's Gender Gap Index among 146 countries (The World Economic Forum 2022).

The point of departure for this research was the paradox that I personally experienced while working in the Japanese hospitality sector, in a traditional Japanese *ryokan* (旅館), during my university exchange in the summer 2019. While working there, I could not but notice the prominently gendered setting the *ryokan* embodied: all the cleaning and serving tasks were conducted by women, and the manager of the service, *okami* (女将), is also a woman. The majority of my colleagues working at the front desk greeting customers were women, and a remarkable amount of them were foreign women who presented to the domestic and international guests the "authentic" and traditional Japanese culture and hospitality, *omotenashi* (おもてなし, See more in Chapter 6.1). This is not surprising as many post-industrial countries have moved to the phase where they require a foreign workforce to counter the demographic changes due to shrinking fertility rates and the growing size of the

elderly population which causes an acute lack of workers in certain fields. Moreover, as the Japanese national government has a long history of emphasizing the homogeneity and monoethnicity of the Japanese nation (See Befu 2001; Chapman 2006; Lee, Murphy-Shigematsu, and Befu 2006; Lie 2001), the evident presence of multiculturalism through foreign presence which contradicts these claims piqued my interest. These details that I noticed inspired me to set forth to understand Japanese gendered labor and multiculturalism's multifaceted nuances.

By applying the concept of "national nostalgia" (Smeekes, Wildschut and Sedikides 2021), I will analyze how Japan's national nostalgia narrative manifests in multiculturalizing society and how it reproduces and enhances conservative gender roles via the experiences of foreign female workers working in Japan's hospitality sector. The data collection method utilized in this thesis is the qualitative semi-structured online interviews conducted among six foreign female workers, and the data was analyzed via thematic analysis. This study aims to contribute to the following four areas: 1) provide a contemporary picture of the Japanese gendered hospitality sector foreign female workers; 2) examine the influence of the national nostalgia narrative in the policymaking and integration of foreign immigrants in an East Asian context; 3) demonstrate how nationalist and obsolete *Nihonjinron* (日本人論) ideologies still manifest in the society and shape the perception of Japanese identity in the eyes of Japanese and foreigners; and 4) illustrate the power relations between foreigners and "the mainstream Japanese" that hinder integration and realization of a multicultural society (See more in Chapter 3.4).

1.1 Research Questions

This research focuses to gain insight on the gendered hospitality sector, foreign women workers and Japanese multiculturalism via three main questions:

- 1. How does the gendered societal roles affect foreign female workers in Japan?
- 2. How does the gender roles manifest in the hospitality sector?
- 3. To what extent Japan's national nostalgia affects the narrative of "Japaneseness" and how it correlates with the obstacles to realize multicultural society?

To attain the most multifaceted comprehensive picture as possible and receive a wide variety of data for the analysis, I am combining semi-structured online interviews and my own

previous fieldwork experiences in a traditional *ryokan* in the *Kansai* area. In similar vein to Ali's (2014; 2015) theories of researcher's positionality, I perceive that only by understanding my own "insiderness" or "outsiderness" positionality in this study and via integrating those perceptions to the joint analysis of interview data (primary data) and fieldwork (secondary data), I am able to distance myself physically as well as psychologically from the phenomenon and gain objective results (Moore 2012). Because my primary data (interviews) and secondary data (fieldwork) are inherently qualitative, my analysis approach will also be focusing on the emphasis on the qualitative representation of the phenomena occurring in Japan's hospitality sector. Furthermore, by employing the concept of "national nostalgia" as a theoretical framework for this study, I am able to understand more deeply the underlying causality relations between multiple interchangeable factors such as societal change, Japanese government agenda, gendered labor market and resistance to multiculturalizing society.

1.2 Thesis Structure

This thesis is divided into seven chapters. The structure of this thesis will be as follows: The First chapter introduces the topic and theme of this study and discusses the theoretical framework and the research questions. Chapter 2 covers the background of Japanese multiculturalism (tabunka kyōsei), the current situation of tabunka kyōsei in Japan and the criticism tabunka kyōsei policies have received. Chapter 3 presents the literature to understand the Japanese gendered society and labor market and illustrates how women became associated with the care work and domestic sphere. This is necessary to understand the research gap and the departure point of this thesis. The Chapter 4 explains the relevant methodology used for this study and considers the chosen methodological approaches as well as the challenges and reasoning behind them. The fifth chapter is the empirical chapter that analyzes Japan's national nostalgia and its manifestations via Japan's soft power, the Nihonjinron debate and how the self-essentialization prevents Japan's multiculturalization. Chapter 6 will explore the same national nostalgia narrative in Japan's hospitality industry and how it affects foreign female workers in Japan. Both of these empirical chapters include extensive discussion of the findings based on my collected data. Chapter 7 concludes this thesis with a summary of the results, identifies the limitations of this research and suggest potential future research possibilities.

1.3 "National Nostalgia" as a Theoretical Framework: Yearning for the Past that Never Was

To analyze my research data and findings, I will employ the concept of "national nostalgia" as a theoretical framework for my study. Social psychologist Smeekes together with Sedikides and Wildschut in their article "Longing for the 'good old days' of our country: National nostalgia as a new master-frame of populist radical right parties" (2021) explored the rhetoric of national nostalgia that populist radical right-wing parties (PRRP) exploit to justify arguments of loss of the ethnically and culturally homogeneous community and tested their hypothesis on native Dutch citizens to assess whether this rhetoric affected perceptions towards nativist ideology such as strengthened anti-immigration attitudes. In recent years, many other scholars have noticed and written extensively on nostalgia's effects on national identity in relation to ingroup (native population) and outgroup (foreign immigrants) relations (e.g., Smeekes and Verkuyten 2015; Smeekes and Jetten, 2019). Albeit I anchor my theoretical framework on Smeekes, Wildschut and Sedikides' article, I will apply other researcher's assessments as well in order to build a richer theoretical framework. Despite the national nostalgia being used commonly to examine radical politic parties in Western countries (e.g., Behler et al. 2021; Elgenius and Rydgren 2022; Smeekes and Verkuyten 2015), I perceive that this concept can be applied to the Japanese context as a national nostalgia-based nationalistic rhetoric of Western right-wing parties shares many similarities with Japan's ruling conservative Liberal Democratic Party's utilized policies and goals to defend the "homogenous Japan".

Nostalgia is defined as "predominantly positive yearning for the past" (Smeekes, Wildschut and Sedikides 2021). Nostalgia can be felt both individually and collectively (Smeekes, Wildschut and Sedikides 2021; see also Behler et al. 2021). In other words, a person can experience nostalgia on the basis of an autobiographic past, but also for objects, periods or shared experiences and events in the past, including the national past (Smeekes, Wildschut and Sedikides 2021). In contrast to individually felt nostalgia, group-based nostalgia can also be experienced in the guise of longing for a past that individuals have not experienced personally but has been bequeathed through collective memory (Smeekes, Wildschut and Sedikides 2021; see also Behler et al. 2021; Smeekes and Jetten 2019). In social psychological convention, national nostalgia is specified to be form of a collective nostalgia based on national group membership that manages intragroup and intergroup attitudes and behaviors of the members of the group (Smeekes, Wildschut and Sedikides 2021). Personal as

well as collective nostalgia is argued to be a coping mechanism which aids in repelling threats to self-continuity (Smeekes, Wildschut and Sedikides 2021). For instance, national nostalgia is perceived to boost the positive attitude towards the national in-group while affecting negatively the perception of the immigrant out-group (Smeekes, Wildschut and Sedikides 2021). After all, national nostalgia-based collectively nurtured national identity and sense of belonging depending on the shared past innately tend to form social categorizations that emphasize the distinctive differences between individuals (Smeekes, Wildschut and Sedikides 2021; see also Smeekes and Verkuyten 2015). In terms of multiculturalizing nations, there is a tendency to highlight a common origin by constructing an identity that separates residents into two distinct groups: the "old-timers" (ethnic majority) and the "newcomers" (immigrants) (Smeekes, Wildschut and Sedikides 2021; see also Behler et al. 2021; Smeekes and Jetten 2019). In this setting, this type of binary opposition can lead to an exclusionist comprehension of national identity and anti-immigration views, framing foreign residents as a threat to the society and culture (Smeekes, Wildschut and Sedikides 2021; see also Behler et al. 2021; Elgenius and Rydgren 2022; Smeekes and Jetten 2019).

National nostalgia is framed by and in sync with contemporary political aspirations and purposes (Elgenius and Rydgren 2022; Smeekes and Jetten 2019). Similarly, nostalgia as a rhetorical strategy has been used by both the right and the left-wing political forces to create their own narrative of multiculturalism (Grainge 1999, 621). Conservative parties perceive immigration to be one of the most important threats to a nation's identity continuation, and in many countries have adopted national nostalgia as rhetoric to counter this threat by painting a picture of a past glorious golden age characterized by ethnic homogeneity and social order which is lost due to multiculturalizing society (Elgenius and Rydgren 2022; Smeekes and Jetten 2019). Modern societies are multiculturalizing via increased mobility of goods and people, and these rapid societal transformations are argued to enhance feelings of nostalgia due to changed societal order (Smeekes and Jetten 2019). Indeed, some scholars define nostalgia fittingly as "yearning for the past in response to a loss" or "absence of continuity in the present" (Behler et al. 2021; Grainge 1999, 631). Therefore, during turbulent times, uncertain conditions and major changes such as demographic and economic changes, nostalgia as a political rhetoric is commonly utilized as a pillar of continuity (Elgenius and Rydgren 2022; Grainge 1999, 621). Even the government itself can revert to the temporal national nostalgia narrative of a homogenous past to protect the national identity from becoming more heterogenous (Elgenius and Rydgren 2022; Smeekes and Jetten 2019).

Interestingly, the narrated golden age might not have ever been real (Smeekes, Wildschut and Sedikides 2021; see also Behler et al. 2021).

To return to the discussion on Japan, regardless of national nostalgia predominantly being employed in the context of Western multiculturalism and conservative party rhetoric, Japan's leading political party LDP has utilized very similar political rhetoric to further its goals. The Japanese government has supported countless times rhetoric of "homogenous Japan" which is new to the issue of immigration and multiculturalism despite Japan's imperialistic multicultural past (e.g., Chung 2010; Flowers 2012; Lie 2001). Lie (2001) and Befu (2001) theorize that Nihonjinron (日本人論), the theories of Japaneseness, enthralled Japanese society after it lost its national symbols following the defeat in the Second World War. Smeekes argues that collective nostalgia may fix the splintered lost social identity by mending it with a new narrative of continuity (Smeekes, Wildschut and Sedikides 2021). Nihonjinron highlight Japan being ethnically homogenic country with a single language, excusing Japan's lacking integration attempts (Befu 2001; Lie 2001; Chung 2010; Flowers 2012). This reflects well with national nostalgia's perception of homogeneity and belonging; if one does not fit the ethnicity requirement, integration is impossible and exclusion imminent (Smeekes, Wildschut and Sedikides 2021). For this reason, this study aims to fill the existing research gap of the potential of national nostalgia as an analytical frame due to the scarcity of scholarship of collective nostalgia, especially on national nostalgia (Behler et al. 2021), in an East Asian context.

2 Background on Multiculturalism in Japan—"I Didn't Know There Are So Many Foreigners in Japan"

The objective of this chapter is to present the necessary literature and context to discuss the nuances of multiculturalism in Japan. Currently the world is internationalizing. This change has resulted in a notable rise in percentage of foreign residents in countries that have previously been considered "monoethnic" (Chung 2010; Liu-Farrer, 2011; Lie 2001). By this convention, Japan has a strong government-backed master narrative of cultural homogeneity and belief in racial monoethnicity (e.g., Burgess 2007; Burgess 2012; Chapman 2006; Lie 2001). Japanese ethnicity and nationality are intrinsically linked, causing the definition of Japaneseness "to be ambivalent, contradictory and inconsistent", and the notion of monoethnicity is used against any threats to "the homogenised culture of Japan" (Burgess 2012; Chapman 2006). As a result of this advocacy, Japanese citizens tend to divide individuals residing in Japan into two contrasting binary groups: Japanese (日本人 nihonjin) and non-Japanese (外国人 gaikokujin) (Lie 2001). However, Japan has never been monoethnic and is home to diverse populations consisting of many ethnicities, minorities, and indigenous people. (Burgess 2007; Lie 2001; Morris-Suzuki 2015b, 194). This attribute alone makes Japan a multicultural society. However, belatedly migration and foreign residents have become an issue in Japanese society, and to ease the tensions caused by cultural conflict, the Japanese central government has started to use the buzzword tabunka kyōsei, loosely translated as "multicultural coexistence" to encourage integration and cultural exchange (e.g., Burgess 2012; Chapman 2006; Demelius 2020). In this chapter, I will first define what is tabunka kyōsei and how it came to academic and political discussion. Next, I present the criticism tabunka kyōsei has received. Lastly, I will discuss the context and reasons for increased multiculturalism in Japan.

2.1 Theory and Meaning of Tabunka Kyōsei

As previously noted, *tabunka kyōsei* (多文化共生), loosely translated as "many cultures living together" (Demelius 2020) or generally translated as "multicultural" (Burgess 2007; Burgess, 2012), is a term which Japan's Ministry of Internal Affairs and Communications (総務省

_

¹ National nostalgia is used for the exact same purpose, see 1.3.

Sōmushō) embraced in 2006 to introduce their program for promoting The Multicultural Plan (多文化共生推進プログラム Tabunka kyōsei suishin puroguramu) (Demelius 2020; Nakamatsu 2014). In the plan the tabunka kyōsei is defined as "people of different nationalities or ethnicities respecting each other's differences and living on equal terms with Japanese as members of a local community" (Demelius 2020; Shiobara 2020). Burgess (2012) argues that in theory, the term tabunka kyōsei should portray foreign residents and Japanese as "equal partners" coexisting harmoniously together. In contrast to the central government's framed discourse of homogeneity, Burgess (2007) points out that the political idea of a "multicultural Japan" is framed so that it dismisses the popular myth of "homogenous Japan" and actually draws factual demographic and economic data to support the argument that further migration in inevitable.

Compared to other multiculturalization-related concepts, tabunka is not as visible in print media as terms such as kokusaika (国際化 internationalization) and ibunka (異文化 different cultures) (Burgess 2012). The term kyōsei in the coexistence discourse seems to have appeared only recently to replace kokusaika (Burgess 2012; Chapman 2006). Kyōsei consist of characters of 'together' and 'life', and it means "living alike together in a particular place" (Burgess 2012). Burgess (2012) notes kyōsei is used by the dominant group (Japanese) to describe their relations with the subordinate group which is the foreigners, and the way the term is used in discourse underlines the majority group's distinctiveness and separateness compared to the minority group. Moreover, it constructs a two-tier hierarchy of citizenship which upholds the Japanese/foreigner dichotomy and denies the non-Japanese access to power (Burgess 2012). Originally the term kyōsei was used in biology to refer to coexistence between species, and later meant Japanese-Asian relations, and finally in the mid-1990s kyōsei transformed to refer to "Japanese and foreigners living harmoniously together within Japan" (Takezawa 2008 in Burgess 2012). *Tabunka kyōsei* should not be mixed up with tabunka shugi (多文化主義, "multiculturalism") which is preferred by sociologists because it has more neutral connotation compared to tabunka kyōsei (Graburn, Ertl and Tierney 2008). Together with tabunka kyōsei, tabunka shugi is interpreted as a policy which supports multiculturalism (Graburn, Ertl and Tierney 2008).

During the postwar period, diversity and multiculturalism in Japan was based on nationality opposed to ethnicity (Chung 2010). Even though terms *ibunka* and *kokusaika* are still used by local governments, *tabunka kyōsei* is slowly replacing them (Shiobara 2020).

Multiculturalism's root is in conveying the importance of maintaining equality and cultural

diversity, and that in a truly multicultural society, co-existence, tolerance, and mutual respect would be perceived to bring benefits to the nation (Burgess 2007). Burgess (2007) notes that multiculturalism is not necessarily seen as supporting immigration but can be a belief that values it. According to Flowers (2012), the term *tabunka kyōsei* was meant to broaden and deepen superficial engagement with foreigners beyond the "consumption of the exotic". She also argues that *tabunka kyōsei* policies have become a valuable part of the discussion about diversity and difference in Japan, but it is uncertain whether policies such as The Multicultural Plan results in increased engagement among different cultures. Lastly, Morris-Suzuki notes (Morris-Suzuki 1998 in Burgess 2007) that if the meaning of culture includes values such as "possession of the same knowledge, values, and experiences", then every national society can be defined as multicultural. Many academics agree with Morris-Suzuki's notion of cultural variation and highlight the uncontroversial truth that multiculturalism clearly exists in Japan (Burgess 2007; Lie 2001; Willis and Murphy-Shigematsu 2008, 221).

2.2 The Beginning of Multiculturalism in Japan

One precondition for birth of multicultural communities is migration (Flowers 2012). Nakamatsu (2014) points out that the public policies on immigration and immigrants in many countries are usually guided by principles like segregation, assimilation or multiculturalism and reflect many demographic, historical, political and social changes and conditions. Lie (2001) traces the beginnings of multiethnic Japan by citing that the fall of *Tokugawa* shogunate gave rise to multiple minorities. First, the collapsed status hierarchies still left the former burakumin² as outcasts of the society, following by the Meiji state annexing Ezochi (Hokkaidō) and Ryūkyū, thus transforming Ainu and Okinawans into minority groups (Lie 2001). Furthermore, as a consequence of Japan's Asian conquest, the influx of Taiwanese, Koreans and people from Japan's other colonized countries settled in Japan creating even more multicultural and multi-ethnic Japan through massive movement of people (Lie 2001). Some of this ethnic heterogeneity was reduced after the collapse of Japan's colonial empire but it would be misleading to consider Japan ethnically homogenous during the postwar

² Burakumin (部落民) minority consists of people who are descended from pre-Meiji castes which practiced occupations that were considered "dirty" like executioners or butcherers.

period until the arrival of the new wave of foreign workers who enhanced the foreign presence in Japan (Lie 2001).

Japan is one of the very few non-Western countries who have accepted a large influx of migrant workers since the 1970s, and during the 1980s the labor needs of Japanese factories and construction sites were supplemented by foreign laborers especially from Asia and South America (Graburn, Ertl and Tierney 2008; Kim and Oh 2010). At that time there was a threefold increase in the number of foreigners entering Japan from 1970 until 1988 (Lie 2001). In 1984 under the leadership of Prime Minister Nakasone Yasuhiro, Japan pledged itself to transform into an international country (Chapman 2006; Flowers 2012). At that time, the national government introduced the term kokusaika and launched several policies to help Japan internationalize domestically as well as internationally (Chapman 2006; Flowers 2012). Although not directly related, multiculturalism is seen as a continuation to kokusaika (e.g., Burgess 2007; Chapman 2006). The Multicultural Japan boom started from mid-1990s, when the term tabunka kyōsei first appeared in 1993 to printed media which spurred the word "multicultural" to become a common term in writing in Japan (Burgess 2007; Nakamatsu 2014). As a result, the academics *Nihonjinron* genre (日本人論 Theories of Japaneseness), which highlights Japan's "monoehtnicity", was critically examined for the first time and it was admitted that social variation and diversity had a long history in Japanese society (Burgess 2007; Lie 2001).

Tabunka kyōsei got the official definition in the Ministry of Internal Affairs and Communications' 2006 Multicultural Plan, where tabunka kyōsei shakai (多文化共生社会 multicultural coexistence society) was described as a society "where people of different nationalities and ethnic groups live together as members of one community, in mutual acceptance of each other's cultural differences and making efforts towards building an equal relationship" (Bugess 2007; Nakamatsu 2014). The Multicultural Plan emphasized four themes: "communication support, livelihood support, multicultural community building and improvement of systems for implementation" and advocated foreign residents' equality and rights for the same government services that the local population had. (Nakamatsu 2014). Previously many multicultural policies were aimed at the oldcomer immigrants such as Zainichi Koreans,³ but when the new immigration wave rapidly increased the number of

³ Zainichi Koreans (在日韓国人) are the ethnic Korean minority in Japan who are descendants of Korean immigrants prior to 1945.

foreigners in Japan, the government shifted its policy-focus to support the newcomer foreign residents and interethnic cooperation (Nakamatsu 2014; Shiobara 2020). To encourage Japanese and foreign residents' interethnic interactions, the local governments have established international exchange offices which organize cultural activities, events, and festivals (Graburn, Ertl and Tierney 2008). In general, multicultural coexistence is implemented through various local assistance services and cultural projects organized by local governments (Flowers 2012; Graburn, Ertl and Tierney 2008). The cities with large condensation of foreign populations offer foreign workers different kinds of consultation services (ranging from legal to employment issues), language and translation services, information handbooks and pamphlets, health insurance and emergency medical coverage and public housing to name a few.

The use of the term "multicultural" was to challenge the conventional view of Japanese society as being monocultural and homogenous (Burgess 2007). One indication for diversity in Japanese society is the fact that more than 2.7 million "foreigners", about 2% of the total populace, reside in Japan (Statistics Bureau of Japan 2021b; Shiobara, Kawabata and Matthews 2020, 118; Takahashi 2020, 131). In addition, one out of 50 babies are of mixed ethnicity in Japan, which also reveals that homogenous Japan is just a myth (Shiobara, Kawabata and Matthews 2020, 118). Recently, *tabunka kyōsei* has transformed into an important part of policy discussion about difference and diversity in Japan because the current policies set very narrow parameters for "culture" that excludes sexual minorities and other similar groups who might belong to a broader definition of culture (Flowers 2012). Thus, Flowers argues that in *tabunka kyōsei* discourse and policies, "multiethnic" would be a more appropriate description as they currently focus on "foreigners" and not on the external diversity found inside Japan's borders such as the presence of *Zainichi* Korean residents and *burakumin*.

2.3 Criticism of Tabunka Kyōsei

Multicultural coexistence society has been criticized throughout the years. One of its weaknesses is that the definition given by the Ministry of Internal Affairs and Communications is highly abstract and idealistic, and the roles of the participants are undefined (Demelius 2020; Shiobara 2020). Nakamatsu (2014) agrees that the phrase *tabunka kyōsei shakai* seems to be unfamiliar at a grassroots level among the minorities it is supposed

to empower. Furthermore, kyōsei is often presented to be a state of harmonious existence that will be achieved without a question (Chapman 2006). The Multicultural Plan is pointed out not reflecting the realities that non-Japanese residents face in Japanese society (Shiobara 2020). In addition, because it is assumed that difference comes from outside Japan, migration is the focus of society's development of place-based tabunka kyōsei policies (Flowers 2012). This conception leaves almost assimilated groups and minorities outside of policymaking (Flowers 2012). Tabunka kyōsei is criticized to put emphasis on the binary opposition of Japanese and foreigners, and in the process enhancing and enforcing the belief of homogenous Japan (Lie 2001, 48; Burgess 2007). When tabunka kyōsei highlights the foreign—mainstream opposition, non-Japanese are seen inferior compared to mainstream Japanese because they have not become self-reliant, and in turn undermines the autonomy and rights of foreign residents and transforms them to vulnerable people in need of support from Japanese government and Japanese people (Shiobara 2020). Moreover, the constitutional guarantee of equal treatment of foreigners with Japanese nationals does not guarantee equality nor individual human rights as the constitution is acknowledged as very equivocal. (Lee, Murphy-Shigematsu and Befu 2006, 2–3; Shiobara 2020). One reason for the lack of equality is perceived to be the lack of sanctions and penalty provisions of the international human rights laws that Japan has ratified, thus letting the human rights violations go without consequences (Lee, Murphy-Shigematsu and Befu 2006, 103).

In addition, the central government delegates most of the responsibility of arranging the smaller compartmentalized projects and implementation of the multicultural policies to local governments under the "the umbrella of local internationalisation" (Nakamatsu 2014). Reason for this is that the Japanese government perceives that local municipalities are responsible for the welfare of their residents, including both native and foreign residents (Graburn, Ertl and Tierney 2008). However, in this way the central government tries to avoid the conflicts which would ensue if foreign workers would not be embraced properly as the national government itself has not taken enough measures to facilitate the social integration of foreign workers (Graburn, Ertl and Tierney 2008; Iwabuchi, Kim and Hsia 2016, 57). In spite of the Japanese government's several adjustments to *tabunka kyōsei* policies and events leaning towards multicultural communication, without proper guidelines the local government's attempts to accommodate local multiculturalism focuses on shallow linguistic and cultural assistance which in turn encourage superficial toleration of the foreign other (e.g., Burgess 2012; Demelius 2020; Liu-Farrer 2020; Munia 2015). Only in the face of notably increased

concentration of foreign population—such as in the case of areas with acute demand of foreign workforce to cover the industrial needs⁴—local municipalities are forced to address the problems of foreigners and provide diverse services (Green 2021; Haines, Yamanaka and Yamashita 2012, 98; Noguchi 2007). However, in the case of municipalities with small accumulation of foreigners, those local governments have been reluctant and inactive in assisting foreign residents (Graburn, Ertl and Tierney 2008).

Additionally, many tabunka kyōsei projects are shallow and superficial multiculturalism and tabunka kyōsei policies overemphasize the importance of intercultural communication (Iwabuchi 2010; Shiobara 2020). The tabunka kyōsei projects arranged by the local communities are merely cosmetic, narrowly focused and situational internalization and might spread and enhance superficial stereotypes of foreign cultures, usually through the "3F (food, fashion, festival)" (Demelius 2020; Shiobara 2020). Flowers (2012) agrees that there is certain "consumer/migrant" relationship between Japanese and the ethnic other. For example, the leisure industry participants at tourism travel seminars in Japan have been requested by the Japanese government to wear "ethnic" dress for receptions, which also denies ethnic other's own subjectivity. However, it is noted that even the local government officials view the policy as unrealistic in principle and assimilationist in practice" and deem that social cohesion achieved through assimilation would prevent segregation (Nakamatsu 2014). The policy's culture-orientated approach results in a paradox where any discrimination or injustice towards foreigners is perceived to be not socially constructed problems but personal matters that can be solved by Japanese and foreign residents themselves (Shiobara 2020).

In contrast, the way to achieve equal civil and political rights with the Japanese population is perceived to be naturalization (Lee, Murphy-Shigematsu and Befu 2006, 100). This in my opinion is related to the government's assimilationist stance. The Japanese national government keeps tight control on immigration and divides the people into beneficial foreigners who are provided with public services and unbeneficial foreigners who are excluded from society (Shiobara 2020). The government presumed that the strict border control ensures the harmonious coexistence because it prevents the "illegal or imposter" people entering who would threaten the national security, but in reality, the tabunka kyōsei policies enhance the control over foreigners (Shiobara 2020). Furthermore, the official

⁴ Many of unskilled labor are concentrated in the vicinity of manufacturing institutions and factories, for example Nikkeijin (日系人, people who are descendants of Japanese emigrates and have later returned to Japan) working in automobile factories (e.g., Roth 2017, 335; Tsuda 1999, 687).

tabunka kyōsei discourse legitimizes social integration policies of the local and national governments and forces the foreign residents to assimilate into Japanese society while at the same time the same society is neglecting and disowning the cultural diversity it deems undesirable (Morris-Suzuki 2015a; Shiobara, Kawabata and Matthews 2020). It is criticized that Japanese society's approval of diversity is decided by person's usefulness from the viewpoint of economic rationality (Liu-Farrer 2020; Shiobara, Kawabata and Matthews 2020). It is also implied that individuals who have assimilated better—such as persons behaving as Japanese or speaking the language fluently—are seen as more desirable and less as a threat to the Japanese culture and thus welcomed as "law-abiding, locally functioning and well-adjusted residents" (Nakamatsu 2014; Shiobara, Kawabata and Matthews 2020, 20).

2.4 Reasons for the Increase in Foreign Population in Japan

Nation-states becoming multiethnic through three mechanisms, which are state-making, colonialization, and migration (Lie 2001). The more prosperous and powerful the nation-state is, the more multiethnic it will be in general (Lie 2001). As briefly discussed above, the central government of Japan utilized internationalization policies starting from the 1980s to increase immigration (e.g., Chapman 2006; Flowers 2012; Willis and Murphy-Shigematsu 2008). As implied earlier, the Japanese government has been against immigration or previously preferred an assimilationist approach to immigration (Nakamatsu 2014; Shiobara 2020), but it has had to change the absolute attitude towards immigration due to the prominent demographical changes occurring in Japan. Kim and Oh (2011) point out that Japan passed the threshold of "aged society" in 1994, and according to United Nation's reports, Japan has now transformed into super-aged society with more than 20% of the populace being 65 years old and older (United Nations 2017). At the same time, the birth-rate in 2018 was 1.3 children per woman in Japan (The World Bank 2020).

Furthermore, Japan's demand for low-wage service labor and manual work became a pressing issue in the late 1980s because Japanese nationals were not willing to take the so-called "3K jobs" (*kitsui, kitanai and kiken*, translated as difficult, dirty, and dangerous) due to the demanding working conditions, long hours of work and low salary (Lie 2001). In addition to these demographical changes and the shrinking labor force, the composition of the labor force itself has changed. The number of women in the labor force nearly doubled between the 1960s and 1980s to alleviate the dire need for workers. By internationalization and accepting

foreigners, Japan tries to alleviate the societal problems caused by an aging population, low birth-rates and a dropping labor force (Demelius 2020; Kingston 2019). However, Japan is still estimated to need around 600.000 foreign workers annually if it wants to keep its current labor force ratio (Graburn, Ertl and Tierney 2008). Due to the amends made to the Immigration Control and Refugee Recognition Act in 1990, now thirty years since the revision there are approximately 2.47 million foreign nationals residing inside Japan's borders (Shiobara, Kawabata and Matthews 2020, 249). The number of foreigners is only 2% of the total Japanese population, but significant as 6.1% of registered marriages in Japan are international, thus Japan will continue to become even more multicultural through the mixing of ethnicities (Kim and Oh, 2011). All these changes contribute to Japanese demographical changes and force Japan to evaluate its stance towards immigration and how they will arrange multiculturalization in Japan.

In the past, even though Japan is seen as hesitant in immigration policies, the government has not restricted the inflow of foreign workers even in the Revised Immigration Act in 1989 (Lie 2001). However, in 1992 the national government outlined the Basic Plan of Immigration Control which made a distinction between skilled and unskilled workers to control the inflow of economically beneficial foreign workforce (Lie 2001). Later, Japan's immigration policies seem to be influenced by demographic and economic changes and the fear towards ethnic conflicts and terrorism (Lie 2001). In contrast, in the Basic Plan for Immigration Control 2005 edition perceived a possibility for allowing medium-skilled workers and emphasized foreigners' right to the social security system and a comfortable living environment. Following those plans, the 2010 plan suggested raising the range of acceptable workers substantially and underlining the kyōsei shakai policies which were introduced in the Multicultural Coexistence Plan in 2006 (Nakamatsu 2014). One of the pioneers of tabunka kyōsei model is Sakanaka Hidenori who is the former head of the Tokyo Immigration Bureau. Sakanaka has been a vocal and influential commentator on immigration issues ever since the 1970s, and especially in the 2010s he advocates for the expansion of the immigration population from the current level of little over two million to around 10 or even 20 million (Morris-Suzuki 2015a).

Despite the various criticism *tabunka kyōsei* policies have received, Japan is step by step transforming into a more multicultural society. Among OECD member states, Japan has the fourth highest annual influx of foreign residents (Takahashi 2020, 130). Another indication for this is that recently Japanese government has implemented several adjustments to mold

itself into more enticing destination of labor migration by easing the previously tightly restricted immigration. The national government launched in April 2023 new visa status "Highly skilled professional" (HSP) for researchers and engineers to lure more highly skilled professionals (Ministry of Foreign Affairs, 15 May 2023). On the other hand, even though highly skilled foreign workers have been accepted the government has refused to recognize them as immigrants (Shiobara 2020). Furthermore, Japanese national government drafted new immigration policy which recognized manual labor also as a "specified skills" in April 2019 (Kondoh 2020; Takahashi 2020, 130). Under the terms of this new "specified skills" category, the foreign workers are proposed to find employment in 12 fields and those with background in nursing or construction sector are eligible to extended their stay for longer term (Ryall, 26 April 2023). In addition, after being criticized for terrible working conditions, serious exploitation and various other abuses, the notorious Technical Intern Training Program (技能 実習制度, Ginō Jisshū Seido) which exploited low skilled workforce was abolished in 2023 (Morita 2023; Tauchi, 10 April 2023). These previously exploited individuals are now allowed to enter Japan with actual visas and work compensations. However, only time will tell whether these new amends to Japan's immigration policy and the presence of foreigners will make Japan to move from toleration to embracing its own increasing diversity and reality as an "immigration country".

3 Literature Review on the Gendered Labor in Japan—"Not a Service Industry Problem"

This section focuses on Japanese gendered labor and strong gender norms promoted by the Japanese government. A strong relationship between the gender roles and the type of work woman are required to conduct surfaced during the study; not only the society, rather the labor market itself is extremely gendered and built on conservative gender roles (e.g., Chapman 2006; Dasgupta 2015; Marshall 2017). In her study about women's labor force behaviour in Japan, Yukiko Abe (2013) notes that women's participation in the labor force lingers behind many other developed countries. This chapter will explore how gender binary manifests in every facet of Japanese society, including national identity, welfare system and labor market. According to the Statistics Bureau of Japan's (統計局 Tōkeikyoku)⁵ provided statistics from 2021, Japan's labor force participation rate in the "15 years old and over" category was 62.1% (Statistics Bureau of Japan 2022, 124). If observed by gender, the rates were 71.3% for male and 53.5% for female (Statistics Bureau of Japan 2022, 124). Traditionally women's low labor force participation has been believed to be the remnant of patriarchal values and conservative society (e.g., Morris-Suzuki 2015; Nemoto 2016). In addition, Japan's labor market is not gender neutral (Broadbent 2003, 26; Ogasawara 2016, 127; Ueno 2021, 9). Men dominate the politics and legislative positions (Assmann 2014, 11; Dalton 2015, 122) as well as many other fields that are considered "masculine" (Assmann 2014, 9; Broadbent 2003, 26; Parreñas 2012, 272). Furthermore, the female employment is highest especially in the "medical, health care and welfare" sector, followed by the category labelled as "accommodations, eating and drinking services" and "living-related, personal and amusement services" (The Statistics Bureau of Japan 2022, 128). All these categories fall under the tertiary industry, amounting 50.9% of total employed women (The Statistics Bureau of Japan 2022, 128).

Based on the above statistics, there seems to be a pattern connecting the type of work women are conducting in Japan and conservative gender roles which can be summarized as "care work". The most prominent employment types for women heavily involve caring; for example, the most prominent categories for female workers are "service workers" and "clerical workers", 68.5% and 60.2% respectively (Statistics Bureau of Japan 2022, 130). It is

⁵ This bureau is subordinate to the Ministry of Internal Affairs and Communications (総務省 Sōmushō, also known as MIC).

-

also important to note that number of women as non-regular staff members is 53.6% (Statistics Bureau of Japan 2022, 131). This is a drastic difference compared to the ratio of men of which only 21,8 percent are non-regular workers (Statistics Bureau of Japan 2022, 131). Based on OECD's 2017 statistics, Japan's gender wage gap is 22.1% which means that women earn only eighty percent of what a man employee earns in Japan (OECD 2021). Approaching from this angle, Japan can be claimed to be a gendered society.

Why is the gender binary so persistent in Japan and what are the reasons for it? The answer lies in the way labor markets have been structured in the postwar era Japan, but also in the traditional view of suitable domains of men and women. The objective of this chapter is to give a better understanding of the history and causal relations behind Japanese society and the enforced gender roles. First the connection between female occupations, hospitality sector and hegemonic gender roles will be explored. Next, it will tackle the multifaceted connections between the gender norms, politics and labor legislation via the breadwinner model which indirectly maintains the conservative agenda of gendered divisions advocated by the national government. Finally, societal obstacles which prevent attaining gender equality and elimination of the gendered division of labor in Japan are considered.

3.1 Women as Embodiment of Hospitality and Care

The gendered division of labor in Japan mainly identifies women as domestic workers and caretakers (Broadbent 2003, 42). In a similar vein, the female-dominated hospitality industry in Japan which belongs to the service sector is plagued by the same stereotypes and expectations. As the above statistical examples highlight, women workers work prominently in the accommodation, eating and drinking services (Statistics Bureau of Japan 2022, 128). The vast majority of Japanese accommodation services consist of traditional inns (旅館 ryokan) which are considered to be an extension of domestic services, and the hospitality provided there incorporates closely the aspect of cozy and relaxed atmosphere served by women staff (Alalsheikh 2016, 36; Alaslsheikh and Sato 2016, 126). As serving is considered

_

⁶ In Japan, workers are offered two types of employment contracts. Regular workers enjoy many benefits and stable employment, whereas non-regular workers lack any type of work security and may suffer abuse like attendees to Japan's notorious the technical intern program (TITP) have experienced. Especially women have been marginalized in the labor force and are disproportionately represented, however, also foreigners suffer from this marginalized status (e.g., Bélanger et al. 2011; Bishop 2005; Vogt 2015).

nurturant and women embodiments of care and politeness (Yoshida 2001, 361), women are perceived to be well-suited in this type of work based on their gender (Broadbent 2003, 15, 19; Nemoto 2016, 222; Lee 2010, 647). Unlike hotels were service can be also provided by men, especially in the context of traditional Japanese inns women embody the traditional values and serve customers (Yoshida 2001, 362). Furthermore, the inn proprietress, called okami, is always a woman and usually the wife of the owner (Yoshida 2001, 362). This echoes the gendered expectations of Japanese women as managers of household and by proxy, protectors of tradition and continuation associated with feminine duty (Koyama 2013, 184; Morris-Suzuki, 2015b). In addition, serving food, cleaning, and taking care of the guests reminds of household tasks and responsibilities women already have at home, and these typical female jobs in Japan that are perceived to require less skills (Broadbent 2003, 15, 19; Nemoto 2016, 222; Lee 2010, 647). Thus, the Japanese service sector must be considered via the underlying assumption of women as caretakers providing nurturing labor which reminds domestic chores and is perceived as low skilled (Broadbent 2003, 42; Parreñas 2012). Based on these ideals, the work performed by females is built on gender ideologies which are constructed on stereotypical view of women as daughters, mothers, and wives (Broadbent 2003, 15, 19; Nemoto 2016, 222). The fact that typically ryokan owner's wife is the okami and workers are called "onee-san" (お姉さん sister) underline this fact.

Moreover, it can be argued that the stereotypical conception of men and women's professions is still very prominent in Japan because many of the occupations women are holding are tied to the lens of masculinity or femininity, thus emphasizing their proper gender roles (Broadbent 2003, 15, 19; Nemoto 2016, 222; Lee 2010, 647). The prior mentioned gender stereotypes are pervasive in Japan, and according to gender essentialism, women are associated with adjectives such as "emotional, irrational, and unprofessional" and hierarchically deemed suitable to be men's assistants (Nemoto 2016, 222). Broadbent (2003) agrees that the paid work done by women has been equated as "temporary or auxiliary assistants to the male colleagues" by the employers (Broadbent 2003, 15). It is noted that Japan's conservative Liberal Democratic Party and the social elite utilize the legislation to keep women's labor cheap and that they rely on certain conformity, only weakly prompting employers to hire more women (Nemoto 2016, 219, 221). Women have limited opportunities

_

⁷ The ideology of "Good Wife, Wise Mother" (良妻賢母 *ryōsai kenbo*) coined by Nakamura Masanao in 1875 set the foundations of moral and gender expectations towards Japanese women as revered mothers (Marshall 2017; Sechiyama 2013, 77). In addition, feminist movements in Japan have employed "motherhood" to define Japanese women throughout Japan's modern history (Eto 2005, 316, 325; Morris-Suzuki 2015b; Muta 2009).

for upward labor mobility and are poorly paid without proper job security (Broadbent 2003, 19; Nemoto 2016, 218). Thus, women's labor whether it is full-time or part-time is seen "as peripheral and unskilled" (Broadbent 2003, 5; Zhou 2015, 115). In addition, Japanese women themselves are reinforcing the pattern as when they choose career, women often choose the so-called "female occupations" like customer service, producing notable gender-based job segregation in Japan instead of STEM jobs (Zhou 2015, 114–115). In the female-dominated fields, the work performed by women are seen as extension of their domestic roles as mother, wife or daughter and the work they conduct resemble domestic chores as they are perceived to be innately build for those type of roles which are nurturing (Broadbent 2003, 15, 19; Nemoto 2016, 222; Lee 2010, 647). For example, there seems to be a gender bias against men in childcare industry, many deeming it being a feminine profession (Osaki, 15 Nov. 2020).

3.2 The "Corporate Warrior": Japan and the Breadwinner Model

Japan's profound tendency for care-based binary at labor market and conservative gender roles are not centuries-old tradition but a deliberate contemporary construction of the leading elite. Post-war Japan enjoyed the benefits of economic miracle with seemingly limitless economic growth, followed by abundance of working opportunities (e.g., Tipton 2015, 220; Hatase and Matsubayashi 2019; Townsend 2013, 498). Japan's unemployment rates were remarkably low in the 1970s and 1980s, only at 2%, and because it supported the robust economic growth, Japan was considered as a "model state" by the West (Tsutsui 2020). The economic miracle is closely tied to gender roles and how they were advocated in Japanese societal structures. Japanese left political parties would have wanted to develop Japan into a welfare state, but the leading Liberal Democratic Party (LDP) wanted to keep the high and stable employment, support restricted state welfare provisions as well as retain the small government. To achieve this, the LDP decided to externalize the welfare to the private corporations and households (Miura 2012; Schoppa 2006). Thus, the conservative LDP fused the economy with welfare provisions provided by women which allowed the exponential

⁸ When hiring foreign workers, Japanese companies require high enough level of Japanese language skills. Thus, foreign male workers do not seem to be advantageous in general, but the international trend of women working in the perceived "feminine fields" highlight this issue as these fields are seen as unskilled and precarious work compared to STEM fields that men are more likely to be employed at. (See for more Adachi 2013; Ikkatai et al. 2020). In addition, overwhelming amount of the highly skilled foreign workers seem to hold visa in "engineering" or "specialist in humanities/international services" (75.6%) (Krishnan, 28 Nov. 2019; Ministry of Health, Labor and Welfare 2018).

economic growth but also forced women to be the uncompensated caretakers in the domestic sphere (Broadbent 2003, 47; Chiavacci and Hommerich 2017, 123; Miura 2012, 48).

Until the bursting of Japan's economic bubble, another feature that enabled Japanese economy's exponential rise was the alleged Japanese-style employment system which was connected to the idea of the system being unique compared to other countries (e.g., Befu 2001, 1; Ohtsu and Imanari 2015, xvii; Tsutsui 2020). In the center of this triad model which supported Japanese economic growth was the breadwinner white-collared salaryman and his full-time income dependent housewife (專業主婦 sengyō shufu) (Dasgupta 2015, 9; Goldstein-Gidoni 2019). In the sengyō shufu model, women were expected to carry out the housekeeping and the provide family's welfare by caring for children and elderly relatives (Aronsson 2015, 38–39; Broadbent 2003, 90; Lee 2010, 647; Mosedale 2011, 131). This was especially important during the 1970s when the state welfare and social services were reduced and government actively encouraged women to work as fulltime family supporters (Broadbent 2003, 91; Goldstein-Gidoni 2019; Nemoto 2016, 5). With this combination, the economic prosperity was ensured through the husband and wife's distinct gender roles stemming from the earlier hegemonic family model and perceived ideal gender roles.

As illustrated above, the national government has reproduced an image of home managing wives as the primary caretaker. The national government not only expects women to support their families and take care of children and the elderly, rather they are also expected to expand their care in the service sector in various roles such as elderly carer, nurse, or teacher (Adachi 2013, 640; Aronsson 2015, 38; Broadbent, 2003, 42). This is not just a governmental plan but echoes a certain value system more deeply rooted in Japanese society. This ideology can be summarized with the proverb "men have job, women have household" (男は仕事、女は家庭 otoko wa shigoto, onna wa katei) (Broadbent 2003, 5; Koyama 2013, 184). A fundamental attribute to this binary model is the gendered ideology associating masculinity with public sphere and production whereas femininity was associated with the household and consumption (Koyama 2013, 184; Morris-Suzuki, 2015b). The husband of the family who receives benefits such as lifetime employment and seniority-based salary and contributes to society directly with his loyalty and long working hours while the wife compliments him by freeing the husband of other responsibilities such as childrearing and household management

⁹ The same mix of Confucian and Western gender values dictated also the framework of hegemonic ideal nuclear family, *koseki* and *ie* systems (E.g., Chapman 2020; Sugimoto 2014, 163–164; White 2018).

(Dasgupta 2011, 11; Mosedale 2011, 131; Zhou 2015, 119). However, this made the wives dependent on the income of their husband, despite the fact that symbolic value and respect was added to the role of the housewife (Goldstein-Gidoni 2019; Nemoto 2016, 44). Despite the effects of prominent women's liberation movements and the economic struggles caused by the oil crisis in 1973, the conservative government together with labor unions sought to strengthen the Japanese-style employment structure which heightens women's role in the domestic sphere (Ehara 2013, 97; see more Eto 2005; Shigematsu 2012.).

The gendered breadwinner discourse is tightly embedded to the socio-political and economic ideology of the postwar Japanese state partnership between the ruling Liberal Democratic Party (LDP), bureaucracy and private industries which provided the foundation for Japan's political economy until the economic collapse (Dasgupta 2011, 10). Furthermore, this kind of structure which was thought to ensure economic success was attained not only through a dedicated paid workforce but together through the uncompensated work that women provided in the domestic sector (Dasgupta 2011, 11; Miura 2012; Schoppa 2006). Nevertheless, if Japan wants to keep its economic growth and cope with the demographic changes such as aging population and falling childbirths, this type of binary division based on different gender-based responsibilities is not viable anymore and the society needs to encourage more women to combine work and children (Assmann 2014, 6; Marshall 2017).

3.3 Homemaker, Valued Employee, or Both? Women's Double Burden and the Gendered Labor

Cabinet Office's survey from spring 2023 divulges that over 80 percent of respondent think that women are burdened by childcare and household (JIJI, 14 March 2023). Kawaguchi and Lee (2017) argue that classic, strong gender norms slow down the institutional changes based on the division of labor within households. These same cultural norms contribute to men's participation in household work despite the fact that women even when full-time worker conduct almost all the household chores (Kawaguchi and Lee 2017; Tsutsui 2020). Compared to other post-industrial societies, Japanese men's participation to household chores and child rearing is low (Brinton, Mun and Hertog 2021; Brinton and Nagase 2017, 360; Mizukoshi, Kohlbacher and Schimkowsky 2016). Ministry of Internal Affairs and Communications' survey on "Time Use and Leisure Activities" conducted in 2021 reveal that husbands used around 51 minutes to household chores whereas wives' weekly average was 3 hours 24

minutes (Statistics Bureau of Japan 2021a). The burden of fulfilling various expectations and combining the work and family has resulted in the practice of women working until marriage and quitting workforce upon childbearing (Ehara 2013, 97; Yu 2009; Zhou 2015, 106). Not all women leave the labor force, but the burden to solely take care of the child due to husband's fulltime work obligations and the possibility of inflexible career influence some to leave work force until children are older (Sugimoto 2014, 174; Yu 2009).

In an attempt to revitalize Japan's economy and alleviate labor shortage, the former Prime Minister Abe Shinzo and his administration launched the economic reforms "Abenomics", of which one core pillar was the "Womenomics" that aimed to increase women in leadership position, to increase childcare facilities, encourage men to assume more active parenting role and to achieve better work life balance by reducing the infamously long working hours of Japanese companies (Matsui et al. 2014, 7; Nagase 2018; Kang 2017). Of women aged 15 or over, the labor force participation rate amounts only 53.5% compared to 71.3% of men in 2021 (Statistics Bureau of Japan 2022, 124). A survey by the Intelligence HITO Research Institute (2015) indicated that three in four young women would rather not be promoted due to negative aspects like bigger responsibilities, longer working hours and lack of female role models (Moriyasu, 14 May 2015). On the other hand, more and more young Japanese women yearn to become fulltime housewives (Sony Life Insurance Company, 1 Nov. 2022). 11 Fulltime housewife stubbornly continues to receive cultural appreciation despite leading Liberal Political Party attempts to lure more women to combine family and work (Marshall 2014; Nemoto 2016). Also, it is important to note that Japanese government's recent efforts in implementing gender equality is based on utilitarian approach to make use of women to counter the demographic changes and not on feminist ideals (Assmann 2014, 5). This emphasis on women's various roles implies that the gender norms not only affect how Japanese labor division and how labor market are constructed, even the society is dependent on the system built on heteronormative traditional gender roles and expectations (e.g., Miura 2012; Ogasawara 2016; Ueno 2021).

_

¹⁰ Japan is an extreme case of gender-based division of household labor, but in Western countries the statistics are only slightly better. In Finland, the average of women's daily household chores was 3 hours and 15 minutes compared to men's 2 hours and 38 minutes. In the European Union, 91% of women with children took part in household tasks whereas only 30% of men did chores on daily basis. (Tilastokeskus 2022; EIGE 2021).

¹¹ This survey is conducted by a respected private institution, and bear in mind that life insurance companies usually have quite accurate data as they collect it for marketing and not for shaping public opinion.

However, the current strict division, government's family policies and low wages sustain the continuation of gendered division of labor (Aronsson 2015, 34; Goldstein-Gidoni 2019). One other social construction which encourages women to stay at home is the tax policy known as "one-million-yen wall", which separates households to primary and secondary earners: the full-time husband and parttime worker wife (Marshall 2017; Kodama and Yokoyama 2018). The state-corporate relationship has also still preserved some tax exemptions which favor single-earning households with welfare benefits such as housing benefit, family allowance and wife's pension. Based on the primary earners income levels, this policy discourages secondary earners who are usually the wives who have given up their career after childbirth to restrict their working hours in order to not to earn more than one million yen unless they end up to higher tax stratum and losing the benefits (Marshall 2017; Miura, 14 Dec. 2022). The conservative dualistic employment system which leads men into long-term employment and women into short-term employment track is argued to be constantly imposed, resulting in women's shorter careers and return to the labor force as parttime workers (Aronsson 2015, 68). It is noted that the base for this kind of dualistic system has long vanished despite the continued support of various institutions dubbing it as "Japanese tradition" (Yu 2009). Under Japan's recessionary economy, double-income households would be a considerable asset (Aronsson 2015, 34).

Japanese government has partly awoken to the reality of importance of female laborers by enacting the "Equal Employment Opportunity Law" in 1985 to boost employment equality between genders (Abe 2011; Ducanes and Abella 2008). Prior to the law together with the breadwinner model the labor market undermined by gender discrimination in education and work opportunities (Ducanes and Abella 2008). Regardless of the labor market opportunities and policies allowing more women to enter workforce, Zhou (2015) asserts that the Confucian culture with deeply rooted patriarchal values still embedded in Japanese contemporary society. In Japan, the national consciousness of suitable gender specific role divisions between men and women is claimed to be more influential than in other developed countries (Zhou 2015, 113). For instance, Goldstein-Gidoni (2019) points out that managers seem to still expect total devotion and support long working hours and prefer women to primarily stay home managing the households. Direct result of this is the fact that women are employed in great numbers in perceived feminine care work related fields such as healthcare and service industry (Statistical Bureau of Japan 2022, 130; Parreñas 2012, 272; Zhou 2015, 115).

Furthermore, despite the implementation of Act on Promotion of Gender Equality in the

Political Field to increase the number of women in the political sphere, the 2019 election saw the number of women representatives fall from 47 to 45 in Lower House Elections and after Kishida Cabinet's reshuffle only two women got nominated as ministers (KYODO, 1 Nov. 2021; 11 Aug. 2022). Thus, the structural stereotypical gender roles of breadwinner husband and homemaker wife remain strong in Japan and are reflected in career paths as well as unequal employment opportunities (Mizukoshi, Kohlbacher and Schimkowsky 2016).

3.4 The Research Gap

The background analysis and literature review have revealed certain gaps in the existing literature that this study aims to contribute to. Firstly, there seems to be a lack of research on foreign women in Japan's service industry. For example, much of the previous research about foreign women focuses on female labor working in the entertainment and sex industry such as hostess clubs (e.g., Aoyama 2009; Kamise 2013) or the foreign female workers in the elderly care industry (e.g., Efendi et al. 2022; Świtek 2016). However, even though these women share some common realities and problems while residing in Japan, the workers in the service industry, like the hospitality business, have a different context. For example, they do not bear the same stigma nor racism, but struggle with inclusion and diverse expectations on other fronts.

Secondly, there is a scarcity of scholarly work on the influence of national nostalgia narratives and policy making in an East Asian context. Most of the studies about the national nostalgia narrative are conducted in Western context (e.g., Smeekes, Wildschut and Sedikides 2021; Behler et al. 2021; Elgenius and Rydgren 2022; Smeekes and Jetten 2019). Even though the Japanese leading elite's narrative shares many similarities with Western populist party rhetoric, it is important to expand from the Eurocentric focus. In addition, as national nostalgia rhetoric's tendency to highlight the homogeneity, monoethnicity, and collective identity serve as an obstacle to foreign residents' integration and creates hierarchical structures, it is vital to study because even if Japan is not perceived as a traditional destination for labor migration, it still is a rising country of immigration (Burgess 2012; Chung 2010; Takahashi 2020, 130).

Thirdly, *Nihonjinron* is rightfully deemed fictionalized self-essentialism, obsolete, and not based on proper academic theory (e.g., Befu 2001; Sugimoto 1999). However, its

contemporary effects on Japanese self-essentialization via government policies, perceptions of Japanese national identity, and its role as an invigorator of cultural and ethnic barriers between the ingroup (Japanese) and outgroup (foreigners) has received less focus. For instance, there is a clear indication that *Nihonjinron* has had a profound impact on shaping Japanese postwar identity (Befu 2001, 101) which cannot be overlooked. The lingering effects of *Nihonjinron* ideology are visible in various facets of Japanese society, and still affect the Japanese people's idea of race, cultural differences as well as perceptions of foreigners (Hambleton 2011; Yamamoto 2015). Newer, more discreet manifestation of this phenomenon is cultural policies such as "Cool Japan" (Hambleton 2011).

Finally, considering all the above-mentioned features of Japanese society and work life, there seem to be unequal power relations in the culture and the society itself. According to the results of the literature review, there is power relations between foreigners and "the mainstream Japanese". Due to Japan's conservative society, women have to balance several roles and fill various expectations and ideologies associated with domestic life and work. Ali (2014) states in her study about empowerment and power relations in Northern Pakistan that women are constantly forced to alter their lives by "balancing various expectations and passions associated with home and work". She notes that the unequal power dynamics stems from multiple factors such as cultural values, societal ideologies, and social and political structures (Ali 2014). Considering all these points, I hope to fill the existing research gap.

4 Methodology—Research Approach, Design, and Primary Data

In this chapter, the methodology, research approach, research design and primary data will be discussed. In my research, the initial research plan that involved travelling had to be changed due to the outbreak of the COVID-19 pandemic. First, this chapter will present the original research plan and illustrate how it has evolved and changed. Secondly, the research approach and other relevant technicalities are considered from various angles and the characteristic of the data is discussed. Lastly, the methodological discussion will include some considerations for researcher's "insiderness" or "outsiderness", which is crucial in understanding the fieldwork, data analysis and the context of this study.

4.1 Methodological Concerns

4.1.1 Defining "the Field": A Traditional Inn and a Gendered Domain

The idea for this thesis developed after my own experience working at a traditional Japanese inn (旅館, ryokan) in summer 2019 as a foreign worker and gained first-hand opportunity to experience the traditional Japanese work environment. The ryokan was located in the Kansai region 12 (関西) and was owned by a single family who also owned two other hotel complexes in the same area. The area where the ryokan resides is one of the oldest hot spring spots (温泉 onsen), and the hot springs there are believed to have healing prospects. Throughout history, the area has been popular among the high-class members of society, making the area very special and a famous tourist destination because of its long history and preserved traditional atmosphere. As the ryokan's history extends to 800 years prior, the company seems to try to ensure the high quality of service and targets affluent and relatively affluent guests who can appreciate its unique position. All three of the hospitality complexes which belong to the company have different qualities and services; the oldest building has 20 traditional Japanese rooms on four floors and three different room types which can house 2–5 guests each; the modern hotel-style inn has around 10 rooms with Western luxury hotel interior; and the last inn has traditional Japanese rooms suitable for elderly people and those individuals who have

¹² The *Kansai* region is located in the Western part of Japan's main island $Honsh\bar{u}$ (本州). The former capital Kyoto is also in the *Kansai* area, and *Kansai* is seen to be the host of a long history of tradition and culture.

troubles with the traditional Japanese architecture.¹³ By a lucky coincidence, besides luxury *ryokan*, I also reached an informant who worked at a hotel resort belonging to a hotel chain. The hotel in question is rather luxurious and holds certain prestige as a provider of customer service. I also got hold of two part-timers, one who worked in an earthquake memorial hall in *Kansai* and another who worked in a big retail chain store in Tokyo. All of these establishments are considered to be part of the service sector and a female-dominated field (See Chapter 3). I define my field as Japan's dynamic, multiculturalizing, and gendered hospitality sector. Due to the fact that this study's emphasis is on the hospitality sector and that I did not have a chance to visit other establishments personally, I refrain from describing them in detail and focus on contextualizing the concept of *ryokan*.

Ryokan originates from Japan and is a unique thing in Japan. Ryokans focus on customer service much like hotels. However, a ryokan differs from a hotel in the sense of its customer service style. In most cases, Japanese traditional inns might go very far to show the traditional spirit of Japanese hospitality (おもてなし omotenashi). Omotenashi is the dominant service concept in the Japanese-style hospitality. Ryokans are the main providers of omotenashi, making it their selling point (Alalsheikh 2016, 62; Morishita 2015, 157). Omotenashi manifests conservative ideals and history that is related to the Japanese mindset and philosophy to selflessly care wholeheartedly for customers; it might be argued that they go all out to make customers' experiences unforgettable. Ryokan belongs to the hospitality sector and employs a significant number of women. What piqued my interest was the fact that it had a great number of women working there, of which many were foreigners just like myself. As around two-thirds of the ryokan staff were women, it can be deduced that the service industry, which a hospitality business like ryokan represents, is a very gendered domain (See Chapter 6).

In particular, women's working conditions and rights differ from those of men's and women's engagement in care-related positions is taken for granted. Another gender-related issue is that the wages for care-related work are significantly lower than in other sectors. Furthermore, besides working and contributing to society, women are also expected to manage other fields related to domestic work and childrearing in Japan (e.g., Lee 2010; Bishop 2002; Kohara and Bipasha 2021). Overall, there also seems to be a lack of studies about foreign women who

¹³ In traditional Japanese Houses, it is common to sleep and eat on the *tatami* mat floor (畳), and there are no elevators so climbing many stairs is inevitable and et cetera.

engage in the service industry as many of the current research focuses on female labor in the entertainment and sex industry such as working at hostess clubs (e.g., Aoyama 2009; Kamise 2013). These issues, together with the troubles of implementing multiculturalism and integrating foreign residents in Japan, make this study topical and relevant to pursue. Thus, I consider my study to be situated in Japan's gendered service industry and the current context of internationalising Japan.

4.1.2 The Original Plan and Adjustments

In the beginning stages of this research, I had planned to utilize participant observation together with one-to-one open-ended interviews among non-Japanese female workers in Japan's service sector. Through participant observation, I hoped to situate myself in the working environment of these women, who would be my study subjects, because I expected that an ethnographic study would help me access experiential perspectives of foreign forwards in the local context. This could allow me to compare the data because I would have seen and experienced similar circumstances myself. Furthermore, participating in their everyday lives could cultivate our relations and ease any possible power relations that might have formed due to the rigid researcher–researched dichotomy (See 4.3). Participant observation also offers a natural form of exchange between individuals as researchers share the same tasks with the research participants as equals. It could provide more spontaneous data as participants conduct their normal daily duties without feeling reservation because of the familiar environment, because getting data from marginalized foreign workers could have been difficult simply by interviewing.

I considered my initial experience of working at the ryokan as my first "fieldwork experience", but as my first "fieldwork" happened before the official start of this research, I wanted to conduct another solely for data acquisition in mind. However, the outbreak of the novel coronavirus pandemic made it difficult to travel and enter Japan for the second fieldwork opportunity and the method to acquire data had to be switched solely to conduct semi-structured online interviews. I decided to concentrate on quality over quantity in data collection as I wanted to understand more deeply the realities of foreign female workers in Japan. Qualitative research entails analyzing and collecting non-numerical data such as text, video, or audio data to understand concepts, research subjects' experiences or their opinions (Gibson and Brown 2009). Qualitative research enables amassing comprehensive insights

about an issue or generating new ideas for the research itself. I perceived that interview as a method would suit my purpose best because an interview innately provides qualitative and extensive data and is flexible to organize. In addition, interviews are also an insightful way to observe the verbal and non-verbal behaviours of the participants. Interviews are split into three main categories which are structured, semi-structured and unstructured (Robson 2002, 70). I deemed a semi-structured interview being the most convenient approach for my research. Semi-structured interviews as consist of specified key themes which are formulated to key questions (Gibson and Brown 2009). He lists the strength of the semi-structured model being the research flexibility; researchers try to shape the pre-defined interests into topics and if necessary, change the order of questions or even research themes based on the natural flow of the conversation rather than pressuring the interviewee to fit their ideas into pre-defined order of question (Gibson and Brown 2009) Semi-structured interviews are conducted with one respondent at a time and includes a blend of open- and closed-ended question together with possible follow-up questions (Adams 2015). Via an open-ended interview situation, it is easier to access the "experimental" or "insightful" perspective of the informants as they offer their own narrative, thus in my opinion being an adequate method for obtaining quality data. As my research themes and topics are sensitive in certain cases because of Japan's history of exploiting foreign workers, semi-structured interview seems like an excellent way for me to create a relaxed and safe atmosphere to discuss difficult topics.

Another beneficial trait of a semi-structured interview is the opportunity for the less formal distinction between researcher and research subject (Gibson and Brown 2009). Holstein and Gubrium (1995, in Gibson and Brown 2009) point out that the less formality creates a "climate of mutual disclosure" between researcher and research subject by allowing the interviewee to have a significant involvement in how the conversation develops, thus creating "the creative interview". Gibson (2009) adds that this way removes the interactional barriers making the researcher an "objective outsider". I perceive this as an asset as my interviewee can freely share his or her own narrative in situations they deem safe and confidential because of the less formal atmosphere and conversational style. Furthermore, Adams (2015) reminds that one-on-one semi-structured interviews are at their strongest when the researcher needs to ask possibly probing questions on topics that respondents are reserved to answer among peers in group interviews. The interviewer can also observe the body language and the reactions of the interviewed persons. In addition, in a situation where the informants would not be fluent in the interview language, talking might be easier for them in comparison to a written format.

Although, compared to other methods the semi-structured interviews can prove to be more labor intensive and time-consuming and researchers need to be well-versed and sensitive about relevant substantive issues (Adams 2015).

Despite my fears that the changes in data acquisition methods would cause me problems, the changes were not too drastic and time-consuming, and the new situation helped me methodologically more than if I had followed the original plan. I was concerned about the possible fieldwork and all the technicalities; if the researcher does not build an applicable research plan before entering the field, the fieldwork might not bring any remarkable findings or data. For example, I would have to plan accurately the whole timeline for the fieldwork to be efficient and successful, contact the ryokan, ask their permission to work there once more and possibly need to answer their questions and assure any concerns that my presence and observation would bring forth. In addition, this kind of ethnographical participant observation would have required me to stay in Japan for months to access the relevant data, and in that case, I would have had to apply for a research travel grant to cover the cost of travel and living, not to talk about the travel in the *Kansai* area which would have been expensive if I could not have stayed in the vicinity of my field.¹⁴ Furthermore, the new online approach provided me with the possibility to focus on research participants' narratives, concentrate on collecting data and nurture the ties with my informants because I did not need to worry about other trivialities which would restrict my time. These aspects would not have come across as multi-faceted via a face-to-face approach.

Consequently, the forced reconsideration of my methods increased my data collection efficiency so that I did not have to travel between possible fields, curtailed the time needed for my data collecting from months to a couple of interview sessions and flexibility in conducting the research in general. However, it must be acknowledged that the new online method does not replace the importance of a real fieldwork experience and the density of data it provides. In contrast, participant observation would have provided me with the opportunity to perceive them in the context of Japanese work life and all the social hierarchies and small nuances in behaviour which is now lost to me due to the change of method. There are numerous limitations in this new method, and it will not provide me with complete data which I would have acquired through utilizing both the participant observation and the semi-structured

¹⁴ One way trip from the nearest big city to the tourist location where the *ryokan* is situated is around 1000¥ which is around 10€.

interviews. Collecting richer data and materials through fieldwork in hospitality industry is an opportunity for a future endeavour.

4.1.3 The Researcher's Own Experiences of Working at a Traditional *Ryokan*

Despite being unable to conduct the second fieldwork due to the effects of corona pandemic, I was surprised by how much my previous internship experience from the summer of 2019 helped me to understand my informants. In retrospect, because I had gone through similar types of work, I could effortlessly parallel their thoughts with my own experiences. For example, many of the interviewed persons who worked in the hospitality sector put emphasis on the elegance or high-classness of their workplace. I must agree with this notion as I noticed the same things at the time I was working at the *ryokan*. Furthermore, many of the co-workers I had a chance to chat with underlined that the service must be perfect to satisfy the guests' experience. One of the permanent Japanese staff members who was referred to be the expert in *omotenashi*—who I later learned is the *okami* who manages *omotenashi* of the *ryokan* even took time from her shift to teach me clearer Japanese articulation when welcoming the guests to the hotel and to teach me how to walk, kneel or bow in the right angle while standing or on my knees. She showed me details such as how to fold umbrella correctly (which I never seemed to do correctly despite many tries) or to properly set things so they would be at peak beauty. She instructed me that had to be mindful of my posture and keep your left hand on top of my right hand while bowing as right as the dominant hand is also used in fights, so it is "an act of submission" to show your unthreatening posture to welcome customers in. I was also coached to kneel delicately and show respect when caring for the shoes the customers took off at the entrance, or if I was carrying guests' luggage to their rooms, I had to set the luggage down carefully. Every time a guest arrived, we had to go wait for their car to arrive and bow and wish them welcome; we also had to reprise it when guests left by bowing while their car drove away, and we were only allowed to return inside when the car passengers couldn't see us anymore. Also, I was told that you cannot run at work or look that you are in a hurry, so if a guest is walking by you need to stop everything you are doing and greet them with a bow.

I also remember how many of these meticulously conducted tasks resembled household chores; together with a Korean girl I had to sweep the entrance of any leaves; I had to use a sticker roll to clean the onsen area of any strings of hair; and to clean and serve the dishes and

carry meals to the guest rooms. As discussed in Chapter 3, much of the work conducted at the service sector reminds me of the work women are expected to provide in the household domain as they are associated as caregivers. One of the foreign workers whom I did not manage to interview for my thesis but with whom I spent a lot of time and shifts together commented many times about this aspect of meticulousness. She said that she feels that the work is meaningless or pointless and too repetitive and meticulous. As an example, she cited that the customer will not notice if their slippers are one centimetre off or the fold on the curtains is not perfectly folded to a symmetrical position. She was ethnically Chinese and had previously worked in Europe where the service was done "laxer", so she did not understand the Japanese mindset to overdo it. This experience helped her to make up her mind to quit working at the ryokan and move back home. She told me about her decision in secret and made me to promise that I would not tell others of her decision before she had settled the official procedures with the company's CEO. She also mentioned another reason for leaving Japan being that the salary is too low compared to the effort she had to put up at work. This resulted in a lot of frustration on her part as she wasn't satisfied with her work and wanted a change in atmosphere.

During the internship, I lived in a two-story dorm house which was located within walking distance from the ryokan. Three of my interviewed persons have lived in the same dorm and it was reserved solely for the female workers of the ryokan. By the time of the interview, two of the research subjects had already moved away. The *ryokan* also had at least one dormitory reserved for the male workers, but I never got to know its location. This kind of strict split between genders was also common in the exchange universities I attended which I found fascinating. As discussed in Chapter 3, the gendered domains show in all layers of Japanese society, even in the labor market, so it isn't particularly unusual. I shared a room with another trainee, but the other rooms accommodated one worker each. At the time I lived there the dorm only hosted foreign workers; the Japanese workers lived in their own apartment. The dormitory residents shared information related to the housing via a group chat which was managed by a Japanese contact person. This contact person, Hanako (pseudonym), was responsible for the property and housing, so she informed the residents of any relevant information like coming maintenance or the cleaning days for the joint areas. I remember Hanako living nearby as I had to message her one time to ask someone to come to fix my room's stove. Living at the dorm was a very communal experience; sometimes my neighbours knocked on my door to ask if I wanted to eat dinner together or go sightseeing on our free

day. The communality between *ryokan* workers was also imminent in the custom to wish your colleagues thank you for hard work, "otsukaresama desu" (お疲れ様です), every time you walked past them or left work.

4.2 The Technicalities: Conducting Research in the Hospitality Industry

4.2.1 Accessing the Field and Study Concerns

The informant group, the female foreign workers in Japan's service industry, was chosen after carefully considering its relevance to this study, and the informants were narrowed down after pinpointing the research focus. Ali (2014) points out that even a single woman's empirical reality can unveil deep and complex understanding of changes and power dynamics occurring at society's micro levels. Following this logic, in this research I aim for quality instead of quantity and concentrate on fewer informants in order to collect comprehensive data. The informant group was chosen for two reasons. First, it was partly decided based on the ease of data acquisition through my previously established relations in the field. I was interested in including informants from diverse backgrounds and perspectives which I deemed that was possible only while having informants whose age, occupation, nationality, experiences et cetera varied. Also, I expected them to potentially be able to answer the research questions and thus give me access to valuable data. Including informants from various backgrounds gives me a broader and in-depth perspective on my focus group (foreign women), but by any means I do not want to generalize them to present all of the foreign female workers in Japan.

At first, I was worried about conducting interviews among service industry workers as Japan has a history of exploiting and abuse of foreign workers which has also drawn immense media attention. Following this logic, some of the foreign female workers at the hospitality industry I interviewed could have experienced this delicate and vulnerable situation. In addition, as some of the interview questions were very private, I worried that they would put my interviewees in an unpleasant situation where they would be hesitant to answer them for fear of the information spreading and affecting their situation. As for the research subjects, my first goal was to access at least four informants, though I acknowledged that five would be the most ideal number for a qualitative study. I contacted four of my ex-co-workers of which two readily resumed contact with me. I was concerned whether my two work colleagues who served as my first informants would provide me with enough information or forward me to

new contacts. In the end, my fears were uncalled for, and I could reach six informants which was more than I had anticipated or planned to have. I was very surprised that they were all of different nationalities, but around the same age and legal status. Four of the interviewees previously knew me, but with the rest of them I created lasting connections so that we can still stay in touch; we even promised to meet after the pandemic receded.

However, as my requirements for informants were female foreign workers in Japan, I feel like the pandemic situation had resulted in fewer people working in Japan, which in turn reduced the sample size. In addition, my informants themselves might have known more foreigners who have worked in Japan, but I was under the impression that they themselves were making some combing out to choose the acquaintances who would most likely agree to take part in my research. If I used more time and effort, I predict that I could have gotten even more contacts. In the end, I decided against it due to time restrictions and the vast amount of data that I had already received. I had trouble including the current gathered data in my thesis to give it justice, and with an even larger sample, it would be nigh impossible to achieve.

Overall, of the six foreign female workers, two had already left Japan and one returned home briefly after the interviews. All had experience in female-dominated service sector work. Two of the informants came from Europe and four of them were Asian by ethnicity. Five of them had worked in the *Kansai* region and one in Tokyo metropolitan area. I was positively surprised by how reciprocal our exchange was and how openly my informants wanted to share information with me about their work; they also spoke their minds on issues they had had while living and working in Japan. I sometimes had to press for more detailed answers, but all in all, I was very delighted with how readily they provided detailed information and honest opinions to me or showed me the way forward for further study subjects. It was also surprising that most of the interviewees thanked me for the possibility to share their own stories with me. One of the most important aspects is the trust between the researcher and the research participants. This includes that the data will be handled accordingly and that I value their confidentiality and willingness to provide me with rich data. Ensuring the confidentiality of the informants in this study, pseudonyms for the informants will be used.

Table 1: Participant Information

Enkhtuya	29	Mongolian	Ryokan	Full-time	Five years working visa	Yes
Minjun	27	South Korean	Ryokan	Full-time	Three years working visa	No
Meiling	22	Taiwanese	Ryokan	Full-time	Working visa	No
Lina	31	Lithuanian	Hotel	Full-time	Working visa (Humanities)	Yes
Amy	21	Malaysian	Museum	Part-time	Student visa with working permission	Yes
Maria	25	Finnish	Retail store	Part-time	Student visa with working permission	No

4.2.2 The Data Collection Method

As stated earlier, I approached the data collecting method via semi-structured interviews and prepared a couple of discussion themes beforehand; it would be left to my study subjects' own judgement how much they wanted to share or if they wanted to refrain from answering to certain themes. I decided that my research sample is the foreign female workers because of my own previous experience of working at Japan's service sector. I did not delimit the age, marital status, or other similar factors of the research subjects as I wanted to acquire multiple perspectives and as rich data as possible. According to Merriam-Webster Dictionary, sampling is defined to be "the act, process, or technique of selecting a suitable sample". Following this trail of thought, it can be partly argued that the initial sampling was achieved by utilizing purposive sampling. Suen, Huang and Lee (2014) explain that this technique is

-

¹⁵ Merriam-Webster Dictionary, "Sample."

utilized by researchers to "carefully select subjects based on study purpose with the expectation that each participant will provide unique and rich information of value to the study". However, it is important to highlight that this method is extremely vulnerable to creating biases because the researcher tries to artificially interfere with the process. Because my subjects share common traits, such as being foreigners in Japan and having work experience in the service sector, it could be argued that I have on some parts employed this method. However, my sampling also includes aspects of convenience sampling. According to Etikan, Musa and Alkassim (2016, 2), the researcher may also choose subjects who are conveniently accessible and thus the most likely to be included in the study. Due to the pandemic situation and lockdowns, conducting online interviews was most convenient for me because of the geographic distance separating me from my potential research participants. Additionally, the fact that I had previous beneficial connections to the field served not only as a prerequisite for my study, but arguably suited me methodologically the best in light of convenience sampling as I had existing accessible entry points to my field. Additionally, I had intended to employ the snowball sampling method. This snowball sampling relies on the current informants who among their social circles will introduce the researcher to new potential research subjects (Emerson 2015, 166). This method is deemed to help a researcher gather the desired number of research participants (Emerson 2015, 166). It is noteworthy that both the convenience and snowball sampling might end up targeting participants from the same geographical area, similar ethnic background or socioeconomical status (Emerson 2015, 166). This might serve as a shortcoming for specific type of research; however, as I specifically wanted to reach female foreign workers in Japan's hospitality sector, luckily this aspect did not hinder my acquisition of extensive data.

Considering the previously mentioned points, I relied on my existing contacts with the staff of the *ryokan* to access my informants. An informant I have no personal ties with might be more inclined to refuse attending an interview with an unknown "prying" researcher. In this regard I considered myself privileged to already have ties to the field. To gather the necessary information, I messaged my ex-co-workers through a social media messaging app called LINE and began collecting the contact details, some of which I had lost. Once I was able to connect with two of my foreign colleagues via the *ryokan*'s Japanese HR specialist, then I proceeded to strike up mundane conversations with them for several months. The reason for such conversations revolved around my concern that my sudden request would make my contacts feel awkward or that I would exploit our relationship to gain sensitive knowledge;

thus, I deemed it necessary to rekindle the relations before approaching them with an idea of interview. My next step was to request their consent as my informants and to schedule appointments for multiple interviews. All the important documents were passed through email with additional information sent over LINE or other preferred messenger applications. After the interviews, I decided to use snowball sampling method and ask these two initial informants whether they know anyone else who has worked in Japan and whether they would be willing to participate in this study. With this method, I could access two new informants. The rest of the study subjects whom I decided to include in my study were my friends. Our status as friends made it easier to ask their cooperation directly. The interview participants were asked to sign the "Consent form for Interview and Research Cooperation" before the interviews.

During the time of the pandemic, I assumed that it is easier to schedule longer online appointments because people stay at home if they have free time. My deduction was somewhat accurate, and informants were mostly forced to stay at home because of restrictions. If the pandemic situation allowed physical working, I had to wait for some of the informants to schedule time for the interviews as they had busy work schedules and few days off. I conducted semi-structured interviews with six persons via ZOOM from September 2021 until November 2022. Adams (2015) points out that ideally the maximum length of a semi-structured interview is one hour as it would keep both interviewee and interviewer's fatigues at minimum. However, despite trying to keep the interview sessions around 30 minutes, the length of my interviews strongly depended on how long each participant wanted to talk about the interview topics. All of the participants were interviewed twice, and the length of the interviews varied from half an hour to almost two-hour. The interview participants could choose freely whether they would like to conduct the interviews in Japanese or English. Because most of the interviewed persons were not confident in the English language, most of the interviews were done in Japanese.

As I am not a native Japanese speaker, I asked permission to record the interview sessions as I was afraid that I might miss important details or misinterpret the data if there came up any terms I was unfamiliar with. This method proved very helpful as some of the interview topics had very specialized vocabulary and the discussions roamed quite freely, sometimes going in unexpected directions. Also, as the informants weren't native speakers either, it was a helpful way to cross-check the information and understand the discussions and context. I transcribed the interviews and translated the ones which were in Japanese into English by myself with

occasional help from my thesis supervisor. Gibson (in Gibson and Brown 2009) stresses that employing recording is decided based on the deemed analytical usefulness but without forgetting the ethical and practical issues involved. He adds that audio and video as data are inherently intrusive as they thoroughly record the practices or performances of research subjects (Gibson and Brown 2009). Generally, video data is harder to anonymize than audio data (Gibson and Brown 2009). To ensure confidentially, it is highly recommended to anonymize, store, and handle the data with utmost care.

4.2.3 The Data Coding and Analysis

Marshall and Rossman (in Gibson and Brown 2009) characterize analysis as involving generating generalized themes from which point of view researchers scrutinize the relationships between components of data. Gibbs (2007) notes that a more generic definition of analysis can be conceptualized as a structure implicating an interest with a reference to relationships and themes. I analyzed my data through thematic analysis, which involves searching for repeated patterns or themes within a data set (Braun and Clarke 2006, 86). Thematic analysis is argued to be a foundational method for qualitative research as it categorizes and explains data sets in rich descriptions, but frequently also interprets aspects of the research topic itself (Braun and Clarke 2006, 78–79).

The next step after the interviews was to transform the audio recordings into transcripts. The interviews conducted in Japanese had to be transcribed subsequently into English. Gibson (in Gibson and Brown 2009) reminds us that as a form of representation, transcripts must be examined as such. He continues that transcribing is not just writing down what participants said or how they acted but it also involves shaping analytic judgements about what to represent and how to do it. By this he means that scholars make judgements on which features such as action, interaction, or talk they focus on, thus deciding to put emphasis on a certain part in the analysis. To put it simply, data is the material gathered during the research when the transcript can be summarized as a representation of the research data (Gibson and Brown 2009). As warned by Adams (2015), making a transcript of multiple semi-structured interviews was an arduous task and required many hours of work. Nevertheless, despite its tediousness, it is also a brilliant way for a researcher to familiarize himself or herself with the collected data (Braun and Clarke 2006, 87).

However, what was even more time-consuming was the coding and analyzing the transcript. As previously mentioned, I employed thematic analysis to search for themes related to my research questions. Fortunately, because I had already sorted the interview questions into themes, it was easier to tackle forming smaller categories, sub-categories, or codes. Coding is a method used to recognize certain patterns in research data (Saldaña 2009, 3). I performed the initial coding manually, meaning that I coded my data on hard-copy printouts which are the recommended platform for first-time or small-scale studies (Saldaña 2009, 22), and later moved the findings to an Excel file for further analysis. Coding requires the scholar to identify and document passages of transcript text or other types of data to form a theoretical or descriptive theory (Gibbs 2007; Saldaña 2009, 8). In short, coding is method of categorizing or indexing research findings to create a framework of ideas about the topic (Gibbs 2007). It is crucial to consider that when a researcher categorizes the found patterns, they must be mindful of the fact that they might group patterns together not just based on the likeness, but because a shared common aspect, even though dissimilarities also belong to commonality (Saldaña 2009, 6). During the coding process, I opted to emphasize aspects linked with gender roles, care work, hospitality (omotenashi) and uniqueness. From the invented categories or codes, I started to create a hypothesis and eventually a solid theory (Saldaña 2009, 12).

4.3 Methodological Discussion: The Researcher's Positionality

The researcher needs to be mindful of his or her own positionality while conducting research. I perceive it to be one of the most important aspects of research and closely tied to the ethics of research. For example, researchers should always give voice to their chosen informant group and objectively present the findings. Thus, the researcher cannot abuse the innate power-relations stemming from the researcher-researched dichotomy. Ali (2014) explains that empowering informants is neither defiance of coercion or standard norms but instead it is about making the informants feel safe and secure within the specific situation. Furthermore, a non-exploitative researcher can produce empowering, objective, and useful research about their informant group (Pillow 2003 in Ali 2015). In this light I wanted to strive to create an atmosphere in my interviews in which my informants feel safe to share information with me so that I am not using my power to coerce information out of them. However, there are also occurrences where this traditional top-down power relations model where a powerful

researcher who exploits a powerless observation group is turned upside down. In Ali's example, one of her informants used power over her by continuously rescheduling interview appointments and thus withholding precious information needed for the research. Therefore, power relations consist of numerous factors, and it is suggested to "look beyond the binary constructions of researched/powerless and researcher/powerful model (Ali 2015). This paper tries to contribute to the discussion of the researcher's positionality in a way in which the research subjects and researcher were both outsiders of mainland culture and customs and the cross-cultural narrative was conducted online which is a context not used too often. However, the COVID-19 pandemic gave researchers push to experience this type of cross-cultural online research which might be used more in the future.

Philosopher Edmund Husserl defined the term "intersubjectivity" to refer to both conscious and unconscious feelings or thoughts between two subjects (Cooper-White 2014). Intersubjectivity is widely used in social sciences because its diverse utility such as "a cognitive agreement between individuals or groups" (Cooper-White 2014). Objectivity is ideal for the research, but it is nearly impossible to achieve because humans tend to have since what we understand, observe, and determine are always affected by our own life experiences and "that is necessary to maintain an attachment between the researcher and the study" (McKingley 2017, 37). For a non-native researcher it might be challenging to comprehend a culturally different field which they are not part of. To achieve objectivity and not hastily interpret data based on the researcher's own predetermined agenda, attitude or conception, the researcher can attempt to create objectivity by removing the prior knowledge and possible prejudices (McKingley 2017, 37). In order to be objective and not influence the researched social group, the researcher might opt for "invisibleness" (Moore 2012). Participating as an invisible individual in research seems tempting, but it rarely succeeds. For instance, a researcher entering the setting without prior introduction or consultation itself is an act of inducing authority, and people are never oblivious to the presence of an observer (Moore 2012). Moreover, in an interview as well as while observing, the researcher might accidentally enforce his or her stronger position by sitting behind the observed group where they cannot see their observer (Moore 2012). Also, in my opinion not taking any contact or not participating in the data gathering situation even in the simple way of asking questions can undermine the researcher's more powerful authority and status by ignoring those in a weaker position. Applying Ali's (2014) perspective of the researcher's positionality to my study, I would like to gain a better understanding of the particular power hierarchies that are

at play between foreigners and Japanese nationals as well as the context and micro-level changes that the foreign presence brings to the communities. She points out that even "a single woman's story" can give a more intricated and deeper understanding of "the power dynamics and the related changes at micro levels of the society" (Groot, Hodgetts and Leggat-Cook 2011 in Ali 2014). According to this logic, by interviewing even one female service industry worker I would already acquire some details about the community, implications of challenges experienced by female foreign workers, the reality of the foreign workers and the changes occurring at society's micro level.

In addition, it is also important to consider the power relations that occur in the field between the researcher and research participants. By this I mean researcher's "insiderness" or "outsiderness" in relation to the group he or she is researching. There are various methods to categorize an insider or outsider, but one common way is to classify an insider as "native" and an outsider as "non-native" or "foreigner" to the society he or she is researching. In the field of anthropology, this polarizing paradigm has been considered "disciplinary wisdom" (Narayan 2003, 285). According to this logic, the "regular" (outsider) researchers study others whose "alien cultural words" must be precisely known (Narayan 2003, 285). In addition, as an outsider a non-native researcher might have a harder time gaining entry to the field. As an insider, gaining entry to the field might be easier due to shared cultural and ethnic background, and previously especially in the field of anthropology those researchers who were deemed "native" were believed to provide an "insider's perspective" from the position of intimate affinity which was authentic and unproblematic (Narayan 2003, 286). However, lately these kinds of views have been challenged, with people proposing that native and nonnative polarization should be abandoned (Narayan 2003, 287; Lu and Hodge 2019). In addition, in the field of anthropological ethnography and traditional sociology, separating researcher into insider-outsider categories are declared to be too shallow and inadequate to truly give insight into the "multi-dimensional interactions in which field researchers engage" (e.g., Lincoln and Cannella 2009; Mullings 1999; Subreenduths and Rhee 2010 in Lu and Hidge 2019). As Narayan (2003, 285) argues, "culture is not homogenous, a society differentiated, and a professional identity that involves problematizing lived reality inevitably creates a distance". Because of the diverse society, the most professional native researcher cannot understand his or her native culture and society wholly (Aguilar 1981 in Narayan 2003, 295). Therefore, it cannot be assumed "nativeness" automatically deems research objective or unproblematic, and even a non-native researcher's narrative (perspective) can be

as valid and authentic. Nevertheless, even though the native-non-native dichotomy is outdated, in this chapter I will succinctly utilize the native-non-native dichotomy to discuss how my entry and methods to access the field differed from that of a native researcher and what kind of challenges or advantages my background gave me. Furthermore, I will also utilize the concept of positionality to "emphasize that the production and understanding of knowledge in research are shaped by and also shaping the way researchers see themselves and are seen by others, and the agentive role researchers play within varied social contexts and structures" (Lu and Hodge 2019).

The issues of researcher's positivity are unveiled most notably when researchers must rethink their approach or readjust their methodology while confronting tensions in the interactions with researched people (Lu and Hodge 2019). Ali (2014) encountered tension in the field and points out that even though as a researcher she shared cultural and ethnic aspects with her informants like language and ethnic background, thus considered native in the society, the small differences in background and opinions eventually transformed her to an outsider of the group. Furthermore, because of her indigenous background, the informants worried that their personal information would leak whereas if the researcher were considered a foreigner to the society, he or she would leave the country with the information (Ali 2014). According to this logic, I as a non-native researcher have an advantage compared to native researchers in the sense that I might be perceived trustworthy by my informants as I have fewer ties with society and will leave the field with my acquired knowledge and not abuse them by using the information against them. In addition, when I first acquainted myself with my informants during my internship period, I shared the aspect of "foreignness" with my informants because we all did not belong to mainstream Japanese society. However, while my informants shared cultural and ethnical foreignness in Japan, they also seemed to belong to their work community because of their job contracts and ties. In retrospect, I as an internship trainee felt like an outsider to the work community because of my different background (entering the workplace due to my university's social connections) and because I was not a permanent employee but left the community quite soon. Thus, I was an insider and outsider to the work community at the same time. However, it must be acknowledged that through continuous engagement within a relationship of reciprocity, relationships, and ties between a stranger (outsider) and the observed can develop and may differentiate from the initial status/position (Narayan 2003, 300). Taking these points into consideration, cross-cultural studies, which my research also represents, are a fortuitous and productive method for discussions about

positionality as participants from various diverging cultural and language backgrounds come into contact (Lu and Hodge 2019).

Another advantage or asset I bring to the field is my position as a woman. My informants and I were all female, which enabled easier interactions and nurturing ties due to shared (features in) womanhood. The informants seemed less hesitant and suspicious of my intentions, and they felt more comfortable with helping and sharing information with a younger woman. Nonetheless, even though in some cases the "insider" or the "native" status is a significant way to enter the field, it is noteworthy to realize that a researcher's positionality is not set in stone and the identity must be debated throughout the process (Ali 2015). Narayan (2003, 285–286) acknowledges that certain factors such as class, gender, and race might overcome the associated cultural identity related to insider or outsider position. She also asserts that the research's focus should be in the "quality of relations" with the people represented in the study, meaning that are they treated as means to an end to acquire information and underlying researcher's own opinion or are they perceived and accepted as individuals who have opinions, dilemmas, and potential criticism towards researcher's "professional enterprise" (Narayan 2003, 286). Furthermore, Narayan emphasizes that a non-native anthropologist can bring in different textures by nurturing the already-formed ties and can even become bicultural through years of continuous fieldwork (2003, 293–294). McKinley (2017, 44) confirms that being a Western (outsider) researcher in Japan is not an advantage, but also is not a disadvantage or limitation. Rather, he argues that by abandoning positivism and embracing constructionist theory – in other words, "to discover the ways that individuals and groups create their perceived reality" – supported by recognizing definable perspective gives Western researchers the much-needed objectivity (McKinley 2017, 44).

It is important to point out that in some cases of qualitative research, the researcher is already a member of the researched social group and thus a "native" or "insider" even before the study has begun (Moore, 2012). Consequently, it is argued that as qualitative research blurs the borders of the insider-outsider categories', so it would be better to study less by "paradigmatic positions" and more from "the physical and psychological distance from the phenomenon" (Moore 2012). In qualitative research, some common power imbalances which create ethical issues might stem from the exploitation of relationships while trying to pass own agenda and the abuse of the power imbalance incurring from the entry into the field (Moore 2012). These points need to be meticulously scrutinized to avoid them.

5 An Analysis of "Unique" Japaneseness—Through National-Nostalgia-Tinted Glasses

This chapter will discuss the research data and findings. This chapter is divided into three sections. Firstly, Japan's soft power and the "Cool Japan" strategy and its utilization in attracting foreign workers to Japan will be considered. Secondly, the "theories of Japaneseness" (日本人論 Nihonjinron); in spite of Nihonjinron having been proven obsolete (Befu 2001; Lie 2001), the innate contradictory nature (Burgess 2007; Sugimoto 1999) and nationalist connotations (Befu 2001; Burgess 2007; Lie 2001) of the theory, it has discreetly integrated itself to be a part of the modern Japanese society and still hold significant relevance in Japanese policymaking (e.g., Burgess 2007; Shani 2019, 1119; Sugimoto 1999; Taira 2019). Therefore, it is important to discuss Nihonjinron from the viewpoint of creating obstacles for integration and the realization of tabunka kyōsei society (多文化共生社会). Thirdly, the societal obstacles and reasons for lacking integration of foreigners in Japan are studied via the data obtained from the interviews.

5.1 Japan's Soft Power and "Cool Japan"

National politics and power are separate. However, contemporary power revolves around money and following this change the power has shifted to the global market (Valaskivi 2013). No country is an exception if they want to benefit from foreign opportunities (e.g., Ferrarini 2012, 127; Ozturk, Joiner and Cavusgil 2015, 120). During Japan's first labor shortages in the 1970s, Japan rejected the possibility to import foreign workers to counterweight the shortage, but contemporary Japanese society does not have that option anymore should the world's third largest economy want to meet its economic needs and counter the demographic crisis (Graburn, Ertl, and Tierney 2008; Morita 2018). In 1984, Japan joined the trend of globalization when Prime Minister Nakasone pledged in his speech that Japan would become an international country and introduce new *kokusaika* policies (e.g., Burgess 2012; Chapman 2006; Flowers 2012). *Kokusaika*, often simply translated as "internationalization", is a complex term, which has little to do with producing more international Japan and letting Japanese companies enter international markets. Rather it was launched to improve international comprehension of Japan abroad and to expose Japanese people to foreign cultures (Flowers 2012; Ishiwata 2011, 1609). This desire was intended to be achieved

through programs which would recruit more foreign graduates and exchange students to Japan and encourage Japanese nationals to travel abroad (Flowers 2012).

However, Japan's image and reputation were tarnished after the Second World War and Japan's identity had to be reconstructed to suit the new international order and make Japan more enticing (Iwabuchi 2015; Kowner and Demel 2015, 408; Nederveen Pieterse and Kim 2012, 169). The Japanese government and its institutions have gone to great lengths in trying to influence Japan's image abroad (Sugimoto 1999, 87). To achieve this goal, the Japanese government needed to alleviate the country's suffering image which had turned unfavorably hostile after the second World War (Iwabuchi 2015). Instead of "hard power" i.e., influencing other countries via forceful means, globalization offered Japan an option to turn to "soft power", i.e., a governmental strategy to influence others through cultural diplomacy (Agyeiwaah, Suntikul and Shan 2019; Iwabuchi 2015; Valaskivi 2013). The term soft power was coined by Joseph S. Nye, and it focuses on cultural exchange and promotion of preferred national image by exportation of various cultural commodities (Iwabuchi 2015). The prevailing cultural tools employed are listed as traditional culture, language education, intellectual exchange and people-to-people exchange programs. These tools act as medium to coerce other countries to be more sympathetic towards Japan through circulation of Japanese culture and values (Iwabuchi 2015; Valaskivi 2013). Japan is observed to have transformed into "a soft power superpower" with the ability to attract others with cultural resources as a medium (Bukh 2014, 462).

In order to transform into the so-called soft power superpower, Japan's Ministry of Foreign Affairs (外務省 *Gaimushō*) has engaged in two cultural diplomatic strategies: public diplomacy and soft power (Iwabuchi 2015). Public diplomacy is defined to increase the international understanding of Japan's disposition on issues and act directly on the international stage through effective publicity (Iwabuchi 2015). The Japanese government established numerous institutions to advertise Japanese media and culture, e.g., the Committee for Tourism Nation (2003) and the Council for the Promotion of International Exchange (2006) to name a few (Iwabuchi 2015). Following these events, the "Cool Japan" policy was formed to cover Japan's aim to shape perceptions of Japan in their specified areas of interest (Nye 1990 in Agyeiwaah, Suntikul and Shan 2019; Iwabuchi 2015; Valaskivi 2013). Even though none of the Japanese ministries have a systematic cultural policy, the Ministry of Economy, Trade and Industry (経済産業省 *Keizaisangyōshō*) proceeded to establish the "Cool Japan" promotion office in 2010, followed by the Cabinet Secretariat's

"the Council for the Promotion of Cool Japan" in 2013 which was assigned to spread international awareness of not only Japanese media content but also items such as cuisine, fashion and traditional Japanese culture (Agyeiwaah, Suntikul and Shan 2019; Iwabuchi 2015; Valaskivi 2013).

Lina from Lithuania, who works in a big hotel chain company in the *Kansai* area shares her first impression of Japan and Japanese people and why she feels attached to Japan, describing them in amendable terms which is in significant contrast to Japan's wartime military aggression:

Attached to home and living, it is (the feeling) not from people, it is from a place. It's like warm shrines, temples, it is really like another world, mythological land of the gods, so it is an interesting country. (Lina, 31, Lithuania)

It seems that the Japanese government has been successfully utilizing soft power to turn Japan's image to one of a more peace loving and harmonious society (Kowner and Demel 2015, 407–408). In Lina's mind Japan has transformed into a warm and mysterious land of supernatural entities and traditional culture. My informant Enkhtuya, who has worked in Japan's service sector for nine years by the time of the interview, recounts her own initial perceptions of Japan as a country:

I thought Japan is respectful and clean country. I think so even now but what else... There is a lot of manga and ninjas, and... There are izakaya, you can watch anime from television (...) (Enkhtuya, 29, Mongolia)

Without any doubt, one of the most popular aspects in the "Cool Japan" strategy is Japanese popular culture as Japanese anime and manga have enthralled international audiences. Joseph S. Nye theorizes popular culture divulging conceptions about a country's values and regards it to be exceptionally influential in shaping people's perceptions on norms, values and the understanding of a culture's different facets (Nye 1990 in Agyeiwaah, Suntikul and Shan 2019). Nye claims that Japan has the greatest soft power potential compared to any other Asian country (Akaha 2005, 67). Goldstein-Gidoni, who studied the consumption and production of Japanese culture on international markets, perceives that despite the fact that "Japanese culture" is principally manufactured in Japan for a local audience, the cultural products later travel abroad (Goldstein-Gidoni 2005, 174). It is noted, that despite Japanese cultural soft power like anime playing a crucial part in motivating tourists to visit Japan, the actual extent to which Japan's soft power influences the decision to travel to Japan has not had a lot of attention in research (Agyeiwaah, Suntikul and Shan 2019). However, in contrast

to the limitations of lacking empirical studies, at least two of my informants revealed that the image of Japan conveyed through Japanese pop culture was an important factor in deciding whether to come to Japan. Lina from Lithuania feels nostalgic about her first encounter with Japanese culture and how it influenced her decision:

In my case, during my childhood days (sic), my brother and I watched "Dragon Ball" and became interested in Japanese culture. After my studies at (sic) Lithuania and exchange year in [University One], I really felt nostalgic towards Japan. I badly wanted to come back so I chose to study again at [University Two] about (sic) Japanese culture (...) I was thinking "how can I come back, I miss that, I miss onsen and karaoke". (Lina, 31, Lithuania)

Lina's opinion is shared by Maria who worked in a big chain retail shop complex during her student exchange. Her motivation to apply to Japan was also her love for Japanese culture and ambition to become a Japanese popular culture specialist:

Yes, so I have been a fan of Japanese culture ever since the middle school times, like the gateway was pretty much Japanese manga, anime and video games. I have been a very intense fan, one of the reasons why I... in general, I applied to my current degree because I want to be a Japan popular culture researcher (laughs). So, ever since I started my studies at the university, my dream was to be able to go to Japan for student exchange. (Maria, 25, Finland)

Valaskivi points out that nation branding intends to recognize something essentially unique in a nation compared to others, but at the same time it cannot be too distinguishable of an unfamiliar feature for the audience not to perceive it as positive and "cool" (Valaskivi 2013). The familiar aspects of the Japanese environment and culture seemed to inspire my informants and what they did in their free time. Many of them also highlighted that some things they can only do in Japan. Minjun, a South Korean woman, said that in her opinion, because it is Japan, it is natural to read manga in Japan: "Because it was Japan, I read Japanese manga." (Minjun, 27, South Korea)

Lina continues:

Back in Lithuania they don't have karaoke (laughs) and onsen and some stuff which Japan has, so it's... some things I can only do in Japan. So now I am starting to appreciate this stuff even more. (...) (Lina, 31, Lithuania)

Maria:

I actually went to this Nightmare before Christmas [anniversary] concert thing, (...) it was a very, very special experience. (...) But also, the fact that such worldwide huge events were just being held in the city I was in, and I could

relatively easily get there, was really big because you don't get those kinds of things in Finland. (Maria, 25, Finland)

As presented above, Japan seems to have successfully influenced foreign perceptions of itself through cultural diplomacy (see Valaskivi 2013). Various Japanese ministries and institutions have made efforts to entice foreign tourists and labor through the interest which Japanese culture has raised abroad. In reality, these internationalization aims were not a proof of Japan's opening up but observed as a temporary movement and means to protect Japan's cultural borders through promoting Japanese culture and values abroad (Liddicoat 2007, 36– 37). This is in line with the fact that national nostalgia is related to protecting the cultural transmission of heritage during times of change like the new global world order (Grainge 1999, 621, 627). However, Japan's soft power vantage position to attract foreign labor is negated by the nation's restrictive immigration policy and cultural bigotry (Akaha 2005, 69). On a transnational level, battling low national self-esteem and the incapability to communicate the strengths of a nation branding are closely linked (Valaskivi 2013). The promotion of "unique Japan" seems to have somewhat been successful as several interview subjects recited that their free time and work was affected by the notion that "these are the things I can only do in Japan". Iwabuchi (2007, in Valaskivi 2013) notes that "brand nationalist" attempts to portray a pre-determined image to the international audience inevitably leads to hiding undesired and uncomfortable features that do not fit the envisioned picture. ¹⁶ Moreover, this is a prime example of the self-essentializing Japan has pursued; by highlighting the exceptionality of Japanese culture and creating a visible brand, at the same time national government is setting clear boundaries between insiders (Japanese) and outsiders (foreigners) and deepening the already existing deep binary (Befu 2001, 127; Burgess 2007; Burgess 2012; Chapman 2006; Ishiwata 2011; Willis and Murphy-Shigematsu 2008, 135). The Japanese government's systematic attempts at promoting and propagating its "unique" and unordinary culture internationally brought forth the genre of debates of "Japaneseness" called Nihonjinron (Befu 2001, 82). Nihonjinron, Japanese "uniqueness" and their relation to contemporary othering and marginalization of the foreigners residing in Japan will be detailed explored in the next part.

¹⁶ In Japan's case, one could argue it has developed national amnesia to its own multiculturalism during the Imperialist past and continuously avoiding addressing the "foreign presence" in Japan and the problems that government's monoethnic narrative brings (e.g., Demelius 2021; Lie 2001; Shiobara, Kawabata and Matthews 2020, 118).

5.2 Nihonjinron and "Unique" Society

Postwar Japan has drawn a lot of criticism from the West on its economic policies and political issues, the common accusations ranging from hoarding economic benefits to ethnocentrism and parochialism (Befu 2001, 1). Japan has accommodated some of the foreign demands but in turn also turned to cultural exceptionalism, arguing that the Western world does not recognize the Japanese uniqueness stemming from its distinctive history and culture (Befu 2001, 1). Much like national nostalgia's tendency to protect group identity via dividing nation to native and non-native groups (Behler et al. 2021; Smeekes, Wildschut and Sedikides 2021; Smeekes and Jetten 2019), this defense mechanism categorizes the West as a collective entity in response to Western allegations, building up a theory of national uniqueness surrounding Japan (Befu 2001; Liddicoat 2007, 37). Born from this convergence of cultures, Japanese mainstream agencies participated in debates on the so-called *Nihonjinron*, translated as "theories of Japaneseness" (Burgess 2007; Morris-Suzuki 2015a; Sugimoto 1999). As previously established, national nostalgia might emerge in a situation where realistic or symbolic threats for ingroup continuity posed by outgroups threaten the individual or group identity (e.g., Behler et al. 2021; Smeekes, Wildschut and Sedikides 2021; Smeekes and Jetten 2019), and Japan utilizing *Nihonjinron* to counter Western pressure is no exception (Lie 2001). Despite the name "theories of Japaneseness", Nihonjinron is often misunderstood as a scientific theory of scholarly origin even though it is created mainly for popular consumption and for the purpose of controlling the narrative of what Japanese culture includes on behalf of the Japanese institutions (Befu 2001, 3; Burgess 2012; Nederveen Pieterse and Kim 2012). Nihonjinron is demonstrated to be a long obsolete and non-scholarly ideology (Befu 2001; Sugimoto 1999). However, these debates have etched themselves into Japanese society and paved a path to resurgence of cultural nationalism among ordinary citizens due to growing immigration (Graburn, Ertl and Tierney 2008). Furthermore, even now the remnants of Nihonjinron beliefs contribute to the othering of the foreign presence in Japan, painting a strict binary of native and foreigner which still influences Japanese society's interactions with foreign people and cultures (Demelius 2020; 2021). For this reason, I deem it necessary to discuss Nihonjinron in the light of the cultural boundaries it constructs to divide Japan from the reality of its own multiculturalism.

It is difficult to define which components make a nation truly a nation. Political Scientist Benedict Anderson claims in his book Imagined Communities that nation and nationalism are not constructed around a particular identity, but a nation is a consciously constructed community created through the perception of people who envisions belonging to the same group despite not ever having a chance to know most of their fellow group members (Anderson 1991, 6–7). Anderson's theory focuses especially on the role of different media ¹⁷ which he perceived having had a vital role in creating nationalist pull to imagine a common nation together as a group (Anderson 1991, 6). Continuing with the same logic, Jeong theorizes a nation being an undefined polity with cultural representations that constantly need to ensure its binding force via self-narrating and creating cultural boundaries (Jeong 2022). Sometimes culture can act as this kind of cultural boundary and binding force. In Japan, the ideas of its own culture have developed together with debates on race which are fused seamlessly with the perception of culture (Morris-Suzuki 2015b, 78). Morris-Suzuki (2015b, 78) observes that not only the idea of race, but the perception of culture is still very influential in modern Japan and long-standing effects on the contemporary Japanese sense of identity. She continues that the word 'Japanese culture' (日本文化 Nihon bunka) has transformed in the 1900s to become a key concept in underlying Japanese uniqueness (Morris-Suzuki 2015b, 65). The concept of Nihon bunka as its non-reflection state has also been utilized by the Japanese government and mainstream Japanese when debating about "theories of Japaneseness" (Demelius 2020; Morris-Suzuki 2015b, 77).

Nihonjinron incorporates three almost interchangeably intertwined concepts: nationality, ethnicity and culture (Chapman 2006; Lie 2001; Sugimoto 1999, 82). In Japan, the race (ethnicity) and citizenship are so closely bound that they are hard for common Japanese to distinguish from each other (Chapman 2006; Willis and Murphy-Shigematsu 2008, 266). Lie supports the view and adds that citizenship (市民権 shiminken) is more unfamiliar concept whereas nationality (国籍 kokuseki) is envisioned as an extension of household registry (戸籍 koseki), thus a more familiar concept to the public. (Lie 2001). In addition, Nihonjinron embodies two main theories: that of Japan being ethnically homogenous nation (単一民族 tan'itsu minzoku) and the unique society possessing dominant group orientation pattern influencing nationals' behavior (e.g., Befu 2001; Burgess 2010; Liddicoat 2007; Lie 2001). Nihonjinron is presented to the public in the shape of ideologies, shared sentiments, media or

-

¹⁷ Many scholars have pointed out how the elite has produced the *Nihonjinron* genre for public consumption which has built the image of "Japaneseness" among the public.

¹⁸ Recently the city of Kumamoto received 2400 complaints after it planned to change the definition of the term "city resident" (*shimin*) to include foreigners because denizens thought it would allow foreigners to vote; this in turn is indicative of the general public's lack of awareness of what rights foreigners have in Japan, and which rights are tied to citizenship (*Kumamoto Nichinichi Shinbun*, 20 Jan. 2023).

even by commodities which are sponsored by corporations and the government (Jeong 2022). *Nihonjinron* is recognized to maintain the secondary nationalism and enable the reenhancement of Japanese national identity (Jeong 2022). Nonetheless, it cannot be dismissed that both ethnic identity and nationalism are not autonomous but deliberately constructed identities (Lie 2001; Befu 2001).

From the very beginning *Nihonjinron* has been a controversial ideology. Many scholars such as Harumi Befu have criticized *Nihonjinron* and delegitimized its claims as they simplify and create stereotypes of the Japanese nation (Befu 2001; Jeong 2022; Lie 2001). In the wake of *Nihonjinron*, many academics have published studies which try to counter *Nihonjinron* by representing Japan as a multicultural nation (Kowner and Demel 2015 404). As a consequence of rapid globalization, *Nihonjinron* and Japan's isolationist mindset have been somewhat weakened in order to pave way for Japan's ventures in transnational affairs (Sugimoto 1999, 87). Nevertheless, the Japanese government and official institutions utilize *Nihonjinron* as a major tenet to maintain the societal structures and to enhance the interests of conservative leadership (Burgess 2012; Jeong 2022; Nederveen Pieterse and Kim 2012, 169). These interests include the aim to influence the image of Japanese culture and society both domestically and internationally (Sugimoto 1999, 87).

In Japan, *Nihonjinron* has functioned as a part of the cultural system that governs representations of the Japanese people and society. Preservation or creation of unique cultural features might be influenced by the aspect of isolation. The Japanese exclusive and isolationist attitude is argued to stem from the mix of its geographical isolation as an island nation and a previous self-imposed isolationist policy which has resulted in Japan's unique homogeneous culture and habits (Itoh 1996; Lie 2001). This assertion is reasonable as Japan is not only a geographically isolated island country that has also experienced a historically relevant insular mindset but has also been blamed of a cultural closure (Goldstein-Gidoni 2005, 173). This view is supported by dominant Japanese historiography of Japan being isolated from all foreign contact (鎖国 *sakoku*) until Commodore Perry forced entry to Japan in 1853 (Lie 2001). In addition, this Japanese narrative sponsored by the Japanese government is evident in the official website of the EU-Japan Centre for Industrial Cooperation (July 2022) as it employs the same narrative of isolation creating uniqueness:

Japan is quite isolated in geographical terms and, as is often the case with island countries, many of Japan's cultural traits and its differences from overseas neighbors are attributed to this geographic isolation. In the case of Japan, the

country's cultural evolution is further explained by factors such as cultural homogeneity, overcrowding brought about by limited inhabitable land and intensive rice production. Anthropologists and other observers, both Japanese and foreign, often invoke its geographical conditions to explain Japan's uniqueness.

Despite Japan not ever being fully closed to cultural interactions and foreign cultural influences, the government institutions' mainstream narrative seems to lean towards the theory of cultural isolationism (Itoh 1996; Lie 2001). This isolation mentality or "island mentality" (島国根性 shimaguni konjō) has persisted as a myth in Japanese society and still affects Japanese public and private sectors (Itoh 1996). Interestingly, this logic is completely opposite to the multiethnicity embraced by the Japanese empire during Japanese imperial period (Oguma 1995; Kowner and Demel 2015, 408; Shiobara, Kawabata and Matthews 2020). Especially during the post-war period, the government's rhetoric shifted to underline a "peaceful and harmonious island nation without foreign other", a rhetoric which gathered support among Japanese citizens (Oguma 1995; Kowner and Demel 2015, 408). In reality, this argument is deliberately used to justify the government's conscious objectives to build barriers to preserve traditional attitudes and customs and not a result of Japan's unique grouporiented mentality (Itoh 1996). The reasoning which was exploited to justify Japan's multiethnic aspirations during Japan's imperial period was shaped to promote an opposing logic during contemporary times. For Enkhtuya, this perception of preserved unique culture due to geographical isolation is an obvious fact. She argues that because Japan is an island country, its culture and manners have remained distinct from other countries, making it a unique and one-of-a-kind society:

I think it is good to know Japanese manners and understand what kind of land Japan is. Japan is—do you know the word "island country" (島国 shimaguni)? Well, because everything around is surrounded by sea, there are unique characteristics to culture. (...) In trains you cannot speak in loud voice, inside bus you cannot speak in phone, these manners are basic manners but because I am Chinese, in China we talk in trains, call on bus, that is common but in Japan there are that kind of rules and manners. (Enkhtuya, 29, Mongolia)

According to *Nihonjinron*, Japanese spirit and customs are so unique that foreign residents are unable to comprehend because they are inherently ethnic attributes (Shani 2019, 1120; Sugimoto 1999). Approaching from that viewpoint, Enkhtuya's comment above highlights this claim as she is referring to the need of understanding and educating oneself of these aspects before coming to Japan. Cultural homogeneity and monolingualism are claimed to enable effective non-verbal communication and cooperation among Japanese citizens (Befu 2001, 25). Furthermore, Japaneseness is listed to include a unique thought process reflected in

language structures and patterns of non-verbal communication (Jeong 2022; Shani 2019, 1119). Japanese is portrayed as a uniquely difficult language, one that is hard even for the Japanese to understand (Gottlieb 2005, 4; Kowner and Demel 2015, 405). This arguably creates a cultural language barrier that complicates the full integration of the ethnic other or even disrupts Japanese collectivist-based harmony according to *Nihonjinron* apologists (Kowner and Demel 2015, 405). Enkhtuya agrees and stresses that without knowing the national language of the country, one cannot truly comprehend the Japanese mindset and culture:

Talking about coming to Japan, people around you only speak Japanese. I am happy that I studied Japanese after all before coming here. If I hadn't, I wouldn't be able to have conversations and, deep down, you can't understand the culture of the country, right? (Enkhtuya, 29, Mongolia)

Additionally, the most drastic *Nihonjinron* dogmas claim that because the language is hard for even native Japanese, then the non-native people are unable to grasp the true meaning of Japanese language and are incapable of ever mastering it (Gottlieb 2005, 5; Kowner and Demel 2015, 405). Lina recognizes the underlying still relevant stereotype among Japanese nationals that foreigners are not good at speaking Japanese due to these emic obstacles: "(...) foreign people can't speak Japanese so well maybe in that case I do not like that stereotype, so I want to talk as much as possible." (Lina, 31, Lithuania)

As many researchers agree, *Nihonjinron* is premised upon the assumption that the Japanese people are uniform in their thoughts and actions (Befu 2001; Iwabuchi 1998 in Jeong 2022). In turn, the uniform thought and actions, such as group-orientation, loyalty and harmony, has been deployed to the needs of conservative ruling elite by managing the society and how it is organized by desired appropriate values (Nederveen Pieterse and Kim 2012, 169). These values affect societal expectations, customs, and rules. Lina also talks about the impact of group-orientation, thoughtfulness towards others (思いやり *omoiyari*) and the strict rules which the Japanese are expected to follow in order not to disturb the harmony:

In Japan they have lot of rules, but in a way, I like these rules. Even on train (sic) you won't speak on phone (sic) because it is common knowledge that you cannot do that. (...) And I don't think that in Lithuania we have those kinds of rules, that for example on the escalator we would stand one side. (...) But it's sometimes double-sided rules what makes your life harder but at some time, these rules make simple because you just need to follow them (...). So, these kind of steps from Japanese people makes our life easier in a way. (Lina, 31, Lithuania)

Because of lingering Confucian values, Japanese society still values courtesy, thoughtfulness, and respect (Yamanaka 1986 in Goldstein-Gidoni 1999). Lina views that Japanese are very respectful towards everyone which is something she personally likes:

I do not think that in Lithuania they have the respect, the deep respect the Japanese have for people. (...) [So, because of the respect] I learned to thank my colleagues for hard work. It was new for me because you don't see that in Lithuania. It is nice cultural thing (...). (Lina, 31, Lithuania)

Respect and thoughtfulness towards others were not only mentioned by Enkhtuya (included in the quote about "island country" earlier in this segment) or Lina, but the stereotype of good mannered and kind Japanese was a perception shared by almost every one of my informants. Meiling narrates her insights about unique Japanese mannerisms and customs as follows:

Whenever you are, in train, eating at restaurant (sic), their manners are really good. They really try not to cause any inconvenience; Japan is that kind of country. ... [The] Japanese also speak really silently, that's why... you can hear each other [clearly], that cannot be done in my country. (Meiling, 22, Taiwan)

Lina is in Japan for the second time during the interview. She reveals that her perception about Japanese politeness and respect has shifted a little bit towards a more realistic mindset after her return. She talks about expectations and peer pressure, which force Japanese to act in a collective custom, even if they are unwilling:

All these people are very polite, and polite in every situation. But then I came for second time (sic), I saw that they are forced to be polite. Even if they don't want to, they are polite. (...) Last time when I went to Japan, one girl just told me that it's the way of Japanese people communicating. They need to put face and be polite person (sic). (Lina, 31, Lithuania)

Other manifestations of Japanese uniqueness narrated by *Nihonjinron* is the habit to distinguish Japanese "self" (内 *uchi*) from the foreign "other" (外 *soto*) (Burgess 2007; Jeong 2022; Shani 2019, 1119). Amy has experienced "othering" while living is Japan, as she narrates that she feels Japanese nationals keeping their distance from her and other foreigners: "Maybe after coming to Japan, my image about very kind Japan has changed. Now they are still kind, but I also get a little bit cold vibe, I mean from the Japanese." (Amy, 21, Malaysia)

There is no denying that *Nihonjinron*'s downside is the construction of generalized and exaggerated stereotypical representations of the Japanese and foreign dissimilarities which are proclaimed to influence how Japanese identify themselves. They also produce psychological stress, discomfort, and nervousness before encounters between native Japanese and

foreigners. This causes obstacles to communication and affects how the Japanese communicate with the ethnic other (Kowner and Demel 2015, 407). Amy further explains her stance:

Comparing Japanese friends to Malaysian friends, Japanese friends have a certain sense of distance. If it's a Malaysian friend, we can be super close and intimate, but if it's a Japanese friend, it doesn't matter how close you are, you cannot say you are that close. I don't understand a lot about practical and concrete things, [but] I think that maybe because our nationality is different it's like that. (Amy, 21, Malaysia)

Amy mentions once more the aspect of nationality, which is a central view in Japanese nationalism and *Nihonjinron* discourse. She views that nationality divides them and the gap between the different cultures cannot be overcome. This is very similar to the narrative of cultural borders *Nihonjinron* tries to paint about a "unique" Japan.

Moreover, Nederveen Pieterse and Kim point out that the concept of Japaneseness is fluid and fluctuating, as the Japanese identity has been perceived to change in a positive-negative binary stratum following historical and geopolitical changes in the economy (Nederveen Pieterse and Kim 2012, 169). In other words, during economic prosperity, the Japanese national identity and self-confidence have been speculated to be bolstered and cultural nationalism is thriving, whereas, during economically harder periods, perceptions about the foreign other are more lenient during economic downturns (Nederveen Pieterse and Kim 2012, 169; Kowner and Demel 2015, 408). Nihonjinron is allegedly the latest phase of attempts to form modern Japanese identity (Kowner and Demel 2015, 408). In this section it was evident that despite the nationalistic connotations and abstract conceptualization, remnants of *Nihonjinron* narrative are found in evolved form in contemporary society, which still affects how Japanese society dictates the binary of "us" and "them" (Burgess 2007; Ishiwata 2011), much like national nostalgia affects the same perception of a native ingroup and a foreign outgroup (e.g., Behler et al. 2021; Smeekes, Wildschut and Sedikides 2021). Befu also argues that, as women are absent from the official *Nihonjinron* debates, in this light, especially the presence of female migrants might help to challenge the "hegemonic" ideology of Japanese identity (Befu 2001, 44).

5.3 Obstacles for the Realization of *Tabunka Kyōsei*

Japan is simultaneously wrestling with two contradictory directions: the need to open up in order to achieve multiculturalism and the desire to stay an isolated and homogenous society (Willis and Murphy-Shigematsu 2008, 31). As discussed in Chapter 2, the Japanese government's realization of a tabunka kyōsei society in Japan is still a faraway reality, due to the ambiguity and the reluctance to clearly define tabunka kyōsei related policies (e.g., Burgess 2007; Chapman 2006; Chung 2010). Regardless of the acute demand of foreign nationals to maintain the world's third largest economy and to counter demographic changes, the efforts made by the Japanese government to lure more foreigners into Japan and integrate them to become part of the society are artificial and half-hearted attempts (Itoh 1996; Morris-Suzuki 2015a; Haines, Yamanaka and Yamashita 2012, 93). At the same time, the Japanese national government insists on Japan not being "an immigration country", 19 the Japanese nationals experience on a local level the reality of a more multiculturalizing society, meaning that the multiculturalizing Japan needs to be addressed on the local level and not the national policy level (Green 2021; Flowers 2012). The Japanese government continues to refrain from explicitly defining the frames of "multicultural coexistence" and pushes the responsibility of implementing the vague guidelines falls onto local governments and municipalities (Burgess 2007; Flowers 2012; Haines, Yamanaka and Yamashita 2012, 93). Some of the local municipalities have stepped up to implement multicultural programs to address the obstacles foreign residents face in their lives but, without clear guidelines, some local governments are unable to effectively implement lasting integration policies, or the policies remain limited (Chung 2010; Green 2021). In this section, I argue that, even though my informants were from more privileged backgrounds and did not seem to require official assistance from the national or local governments, the data still portrays major obstacles encountered in their everyday lives. Furthermore, based on my data, I perceive that, despite the increasing foreign presence in Japan, the local municipalities where my informants lived are not yet desperate enough to accommodate, advertise or organize local programs to alleviate the problems born from the clash of different cultures.

-

¹⁹ See Chapter 2.

Japan considers itself to be new to facing immigration as a country, excusing their slow progress in integrating foreign immigrants²⁰ with arguments of historic isolation and national homogeneity (Green 2021). Despite Japan's interest in achieving kokusaika to counter demographic changes, the internationalization is implied to be slowed by the sakoku mentality, as government implemented immigration restrictions and strict requirements to block integration and assimilation (Itoh 1996). Together with weak governmental leadership, the island mentality delays decision making structures, especially bureaucracy, and discourages decision making, tending to encourage problem avoidance and build obstacles for Japan's internationalization (Itoh 1996). Because of the vague tabunka kyōsei policy guidelines, only a slim majority of the biggest cities have adopted any kind of multicultural plan (Green 2021). It is hinted that especially municipalities which are facing economic and demographic concerns, such as a large elderly proportion among residents, are more likely to adopt tabunka kyōsei plans (Green 2021). In this notion, Japanese are speculated to suffer from "racial privilege", meaning that, unless the size of foreign population is notable enough, the dominant Japanese majority can live their lives without needing to dwell deeper on the issues of ethnic minorities or foreigners residing in Japan (Hammine and Rudolph 2022, 97). Together with the previously presented fact that *kokusaika* made Japan define Japaneseness and protect Japan's uniqueness by othering the foreign presence, arguments about racial privilege can be generalized to mean that policymakers are unwilling or even disinterested in furthering multiculturalism and foreign integration in Japan. In addition, these imagined perceptions shape national immigration policies that determine whether people are worthy of inclusion or not according to the desired image and qualities they bring to the nation and community. As such, imagined national identity relates directly to inclusion and exclusion (Bauder 2011, 9; Shiobara, Kawabata and Matthews 2020). Tabunka kyōsei aims for people with different nationalities to live harmoniously among each other, so it is often argued that, as a spiritual successor for kokusaika, the policy is othering foreigners by inclusion through superficial and stereotypical interactions and international events between different cultures (Burgess 2007; Chapman 2006).

The programs the government has promoted to enhance Japan's internationalization are defined to aim to increase foreigners' understanding of Japan, its culture and society as well

⁻

²⁰ The Japanese government avoids using term "immigrant" and opts to use other terms as immigration is seen permanent and they wish foreign workers to leave after their usefulness have vaned (e.g., Flowers 2012; Morris-Suzuki 2015a).

as expose Japanese citizens to foreign cultures at the same time (Flowers 2012). These aims consist of three distinctive sections; launching programs to recruit foreign graduates to Japan; luring in more exchange students to Japan; and getting Japanese to travel more abroad (Flowers 2012). Another measure is to promote multicultural interactions via different community events and to build a certain multicultural coexistent spirit through joint events (Demelius 2020; 2021; Flowers 2012). Five out of my six informants admitted to not knowing exactly whether their neighborhood had any special community activities including multicultural exchange programs or had never taken part in such events: "I have not taken part in those. (...) In the area I am living, there are no activities. (...) I haven't heard much of those." (Enkhtuya, 29, Mongolia)

Because of the lack of clear policy guidelines for multicultural coexistence, public services and governmental support offered to foreigners, the citizens of Japan have organized assistance groups for immigrants in need of public services from 1990s onwards (Haines, Yamanaka and Yamashita 2012, 92–93). Local governments hosting large immigrant populations also executed their own policies whose goal was to alleviate foreign residents' disadvantaged positions in fields such as education, language, health, and housing (Demelius 2020; Haines, Yamanaka and Yamashita 2012, 92–93). Some regional governments have addressed these issues and have started to offer various programs for foreigners, including radio programs in their native language, festivals and education but also providing opportunities for interaction between natives and foreigners (Graburn, Ertl and Tierney 2008). Social rights, such as public housing and social security, have also been guaranteed (Morris-Suzuki 2015a). In addition, local governments also provide multilingual brochures, magazines, and newsletters (Burgess 2007). Despite not having had to relied on these types of help, Minjun recounts the services she has heard local governments are offering to foreigners:

There are leaflets and those things, there seems to be Japanese classes for foreigners, people who are suffering and have a hard time are given counselling. When foreigners want to find a workplace, they also help then. They also have help for legal problems, those kinds of things. (Minjun, 27, South Korea)

Enkhtuya has heard about the possibility of getting social aid: "(...) for example, if you quit a job, if you lose job, you go to non-governmental office, they give you consulting, in addition (...) you can receive subsidiary aid. If you don't have a job, you are given unemployment allowance." (Enkhtuya, 29, Mongolia)

Tabunka kyōsei has also been criticized for being too superficial and idealistic, and for not depicting the actual realities foreigners encounter in Japanese society (Shiobara 2020). For example, numerous foreign residents in Japan are reported to experience discrimination in communities and from governmental institutions, notably in the areas such as education, employment, housing, and medical services (Burgess 2007; Demelius 2020; Lee, Murphy-Shigematsu and Befu 2006, 102–103). Blatant housing discrimination against foreigners is especially common in Japan (Lee, Murphy-Shigematsu and Befu 2006, 3). My informants did not mention ever experiencing housing discrimination, but many voiced their stress about the need for a Japanese guarantor:

That (Japanese guarantor) is important for Japanese (...) you ask a Japanese person as a guarantor. I think that is a little inconvenient. Foreigners must search among the Japanese they know and ask them to become one, it is no good if you have not got a guarantor. (Enkhtuya, 29, Mongolia)

Liu-Farrer argues that the importance of forming and maintaining social relations with Japanese nationals is essential for non-Japanese to progress in Japanese society (Liu-Farrer, 2011). Enkhtuya's above comment raises the valuable question about the importance of social connections with native, ethnically Japanese citizens. As previously explored, tabunka kyōsei ideology builds a binary opposition between Japanese and foreign residents and strengthens the notion of homogeneity already lingering from the *Nihonjinron* debate (Burgess 2007; Chapman 2006; Shiobara 2020), ending up excluding the foreign other (e.g., Befu 2001; Burgess 2012; Demelius 2020). Tabunka kyōsei divides the Japanese to uchi (内 inside a group) and foreigners to soto (外 outside a group) peripheries, underlying foreign residents' position as not being a part of the society because Japanese society is believed to consist of only Monoethnic Japanese (Lie 2001; Morris-Suzuki 2015a). Shiobara joins other academic critics and argues that tabunka kyōsei policies in Japan pledge, instead of ensuing human rights to the foreign sojourners, guarantees their self-reliance and support so that they can "live in Japan like Japanese" (Shiobara 2020). However, if you have not integrated, it is a stressful task to find a guarantor and, in a country that stresses the difference of native and foreign, it creates a lot of pressure, especially for those who do not plan to stay permanently. Not even one of my informants voiced definite plans of staying in Japan and did not address the notion of assimilation because of their temporary status. In addition, none of them had needed to rely on any governmental help while staying in Japan; they could live normal life among the Japanese like tabunka kyōsei policies dictated and thus could be considered relevantly privileged. However, I argue that, despite my informants' seemingly privileged

position, it does not mean they were integrated into society. When asked to describe what kind of relationships they had formed in Japan, Minjun said that, despite forming many friendships, they were mostly from company circles:

After I did jobhunting and came to Japan, in the end, my own relations were with my company's people. There were people of various ages, with people younger than me, people older than me, I could make friends with people of various ages. (Minjun, 27, South Korea)

Enkhtuya concurs that she has made friends with various nationalities mainly from her work circles: I have formed relationships which became friendship – well, almost all are workplace relationships, but also the restaurant I went to eat, that restaurant's owner became my acquaintance. (...) (Enkhtuya, 29, Mongolia)

The concepts of *uchi* and *soto* are still relevant in modern Japanese society (Lie 2001). As pointed out above, my informants avowed that they did not interact with persons outside of their own friend circles, and pointed out that Japanese nationals living close by did not press for interactions either. This trend was only interrupted by the occasional occurrences of receiving vegetables from landlords. Lina lived at communal housing which houses tighter community spirit, but, when she moved from *Kyūshū* to the *Kansai* area, she acknowledged that the atmosphere changed, and the smallest interactions between native Japanese disappeared:

I had one Japanese elderly lady near my apartment who was always coming, asking how we are doing, bringing some snacks and vegetables, so she was a nice lady. And I heard another student got some kind of cake hanged on his door. (...) Back in $Ky\bar{u}sh\bar{u}$, I was living in prefectural housing (...). So, I think maybe it is even prefecture's idea to bring community together, in a way. Here (*Kansai*) I cannot say a lot about my recent apartment because I did not interact with them that much. (Lina, 31, Lithuania)

Another manifestation that divides Japanese society and prevents multicultural coexistence and full integration is the problem of official language. Language is invoked to have the power to unite and exclude (Linton 2009). Language is power, as it is tied to politics (Joseph 2006, 1–3) and the role in forming and maintaining national identity (Joseph 2006, 147). As mentioned earlier, *Nihonjinron* concerns itself very much with the notion of language, as it claims Japan to be a culturally homogenous society with only purely ethnically Japanese persons who speak Japanese (Befu 2001; Lie 2001). If one cannot comprehend or speak Japanese, according to *Nihonjinron* principles they are not considered a part of Japanese society, because foreigners without Japanese blood are unable to ever understand Japanese

language or culture (Liddicoat 2007, 35). Furthermore, Amy, who studies at a local university in the *Kansai* area, points out how it is impossible to live in Japan without mastering the official language: "Japanese language is important (laughs). English or other languages are almost never used inside Japan." (Amy, 21, Malaysia)

In support of Amy's quote, the biggest limitation of the Japanese language is its marginal use on a worldwide scale, which restricts and complicates Japanese nationals' means of communication with other nationals remarkably (Akaha 2005, 69). Japan is noted to be a country with very low English proficiency; it is estimated that only 30% of Japanese speak any English at all, despite the positive image English language carries (Margolis, 26 May 2020). This highlights the need for Japanese language skill on every facet of Japanese society and puts pressure on non-Japanese residents to learn the language in order to lead their lives in Japan.

Japan is not really a country where people have a high proficiency in English, so it really matters whether you can speak Japanese or not. (...) It is not really a question where it has been useful. I think it would be pretty impossible to just live in Japan without knowing at least basic Japanese. (Minjun, 27, South Korea)

As Japan is the only spoken language in Japan, this might cause unequal and hierarchical power relations, which are employed through certain economic, political, and social practices and can be utilized as forces of discrimination and marginalization (Mohanty 2010, 132). For example, some languages entitle people by granting them access to resources which speakers of other languages are unable to attain, leading to a disadvantaged position or even marginalization (Mohanty 2010, 150). Also, Japan's insistence on a high level of Japanese language skill is narrated to hinder recruitment efforts for recruitment of caregivers (Kingston 2019; Morita 2018). On account of the fact that it would be in Japan's interest to maintain global competitiveness with foreign labor while its own population is shrinking, if the local governments want to build more international and accessible communities based on the harmonious coexistence, they need to address the linguistic needs of the growing number of foreign residents (Carroll 2010; Morita 2018).

However, academics note that outreach activities, like mentioned above, remain on low levels despite the heightened presence of foreign immigrants (Green 2021). In addition, the offered local integration initiatives are observed to be mainly information providing and service-based and only addresses issues as they occur, thus not being integrative or inclusive in nature (Green 2021). The efforts fail to transform foreign residents into active participants in local

communities, which the government's *Tabunka Kyōsei* Plan visions as its final goal (Green 2021). All of these points are visible from the data I received; besides the brief listing of provided services, none of my informants knew in depth what kind of services they were eligible to receive nor were they active participants in their respective communities. Instead of designing schemes on top of new schemes which have no lasting effect due to the expectation of assimilation, it is suggested that the Japanese government needs to contemplate how to build a welcoming atmosphere to make foreigners feel welcome to join (Kim and Streich 2020). Moreover, different types of foreign residents require different types of assistance from municipalities and local governments, and in the future, their special needs must be addressed better (Kim and Streich 2020). In addition, Japan's immigration policy emphasizes only specific economic aspects such as impending labor shortage or the immigrants' presence breaking the harmonious society (Willis and Murphy-Shigematsu 2008, 233–234). Policies that are drafted without the consideration that foreign migrants are also participants in society and likely future Japanese citizens widen the already deep gap between reality and policymaking (Willis and Murphy-Shigematsu 2008, 234). Furthermore, Ishiwata (2011) stresses that, despite endeavors to introduce more multicultural policies locally as well as on national level, the attempts remain limited, as they are based and framed by the exclusivity and exceptionality based on the *Nihonjinron* which shape Japanese identity, and paradoxically—the presence of the foreign minority does not reduce the ethno-cultural hierarchies, rather the foreign other highlights hierarchized distinctions between Japanese and foreigners, thus building persistence for the dichotomy.

6 An Analysis of the Gendered Hospitality Industry—"Ryokan Is the Embodiment of Japanese Culture"

In this second empirical chapter, the data related to Japan's hospitality sector and conservative gender roles will be analyzed. This chapter is divided into two sections. First, it will explore the traditional Japanese hospitality culture (おもてなし omotenashi) as the embodied proxy of "authentic Japaneseness" and how the national nostalgia-imbued views of *omotenashi* reinforce the persistent narrative of Japan's uniqueness left by the Nihonjinron discourse while at the same time examining how *omotenashi* is used as a conscious strategy for branding "authentic Japanese culture". This section will also discuss how foreign women in Japan's service sector are forced into a peculiar position to represent the "authentic Japanese" culture" to guests despite being "othered" as foreigners by the society (See Befu 2001, 82; Burgess 2012; Willis and Murphy-Shigematsu 2008, 6). Next, the service sector and especially the traditional Japanese inn (旅館 ryokan) as a gendered domain representing and enhancing traditional gender roles will be contemplated. To be specific, this section will look at the hospitality sector, women's precarious positions are affected by the underlying rigid gender role expectations which dictate what type of work is suitable for them and what fields they choose to work in, unknowingly strengthening the deep gender gap and binary that exists in Japanese society.

6.1 Omotenashi as the embodiment of Japanese culture

As previously noted, nostalgia is related to values such as authenticity and tradition (Grainge 1999, 621). In addition, national nostalgia is deployed as a preserving factor during major changes perceived as threats to the nation (e.g., Behler et al. 2021; Elgenius and Rydgren 2022; Grainge 1999, 621). Japan's modernization set the foundation for abundant economic growth, but also became associated with a sense of loss of tradition and surging nostalgia for the past (Yoshida 2001, 363). *Nihonjinron* is closely related to the nationalistic argument on "nativeness" and attempts to define and reclaim authentic and traditional Japanese culture. Discourses of *Nihonjinron* were also present during Japan's rapid internationalization in the 1980s when Japan's former Prime Minister Nakasone gave his proposal for "*Furusato* Japan" plan which was launched to re-vitalize and reshape the traditional and native aspects of local and regional cultures and enable the Japanese to appreciate and learn positive qualities of their

own culture (Graburn, Ertl and Tierney 2008; Valaskivi 2013). Goldstein-Gidoni affirms that Japanese official authorities have relentlessly entangled themselves in protecting and reproducing "traditional' Japanese cultural properties" such as traditional artisanal techniques, locations or even people and gone to great lengths ensuring that "traditional and authentic Japan" is accessible to the foreigners (Goldstein-Gidoni 2005, 159). More recently, "Cool Japan" has been employed as a measure to brand these cultural properties and revitalize the local lifestyles of dying regions (Valaskivi 2013).

Following the above logic, traditional and "unique Japanese culture" is preserved and found at grassroot level events, locations or even customs. According to interview data and my own fieldwork experience, I perceive that *ryokan* represents this type of establishment: one that upholds traditional Japanese customs, values, and spirit. Some scholars concur with the government's undisputed narrative that *ryokans* advocate the "authentic Japanese culture and mindset" (Alalsheikh and Sato 2015, 153, Karakawa 2019, 10). The services of *ryokans* are derived from *omotenashi*, usually translated as "Japanese-style hospitality", which originates from Japanese tea ceremony practices but has evolved to cover the hospitality provided in the whole service sector (Alalsheikh and Sato 2015, 123). *Omotenashi* became a media buzzword after a biracial newscaster and announcer Takigawa Christel²¹ held her Olympic bid speech at the International Olympic Committee's General Assembly in Buenos Aires. She presented *omotenashi* as an integral part of Japanese culture and even claims it to be an "embodiment of Japaneseness" since time immemorial:

We will offer you a unique welcome. In Japanese, I can describe it in one unique word: "omotenashi". It means a spirit of selfless hospitality... One that dates back to our ancestors... Yet it is ingrained in Japan's ultra-modern culture. "Omotenashi" explains why Japanese people take care of each other... and our guests (ANNnewsCH, 7 Sept. 2023).

Takigawa's speech created the "omotenashi boom" in the media and even the Japanese government started to advertise and promote omotenashi more widely in their campaigns to attract foreign tourists to Japan to experience the world-class hospitality (Yamada, March 2012; Schreiber, 3 Feb. 2018). According to Travel and Tourism Competitiveness Index by The World Economic Forum, Japan ranked 1st in 2021 (The World Economic Forum 2022b).

²¹ Takigawa Christel is the current wife of former prime minister Koizumi Jun'ichirō's eldest son Koizumi Shinjirō. Despite Japan's insistence on the notion of monoethnicity inside the nation itself as seen in the *Nihonjinron* discourses, choosing a biracial newscaster as Tokyo Olympic bid speech presenter signifies to present a more "multicultural and multiethnic" society outwards.

Omotenashi is reported to include high-quality service, which is influenced by Japanese culture, history, lifestyle as well as nature (Morishita 2015, 157). Ryokan is noted to be one of the principal and major providers of omotenashi (Japan National Tourism Organization). The term "omotenashi" is hard to define and better understood if experienced in person (Morishita 2015, 157), highlighting its unique qualities and "only in Japan" experience. My informant Enkhtuya seems to share her opinion with above notions, going as far as to declare that traditional Japanese-style inn is the embodiment of Japanese culture because of the hospitality served there and to truly understand Japan one has to experience ryokan:

Ryokan is a unique thing of Japan, in other countries there is no such thing as ryokan. (...) Also, while working at the front, [I need to present] Japanese hospitality (omontenashi). (...) Ryokan is Japanese culture's embodiment (塊 katamari). If you think about a country like Japan and want to understand it, its history and so on, I think they are all inside the term "ryokan". (Enkhtuya, 29, Mongolia)

Omotenashi is perceived to physically manifest in the form of extremely respectful language (敬語 keigo), bowing, and attentiveness in anticipating customer's needs before they personally realize them (White, 5 Nov. 2018). Fundamentally omotenashi presented at ryokan consists of four distinct characteristics: treating guests, offering feasts or receptions, understanding the behavior or attitude of people, and the treatment or procedure of certain activities (Morishita 2016, 157). These practices are also embodied in the etymology of the term *omotenashi* as it means to "truthfully connect with others" or "to establish better human relations" (Morishita 2016, 157). This mindset is present when we compare Western style hotel facilities with Japanese inns. Compared with the materialistic hospitality provided by modern Western style hotel facilities, ryokans with hot springs have employed omotenashi to entice customers to experience the traditional Japanese atmosphere and culture, thus going for more spiritual and emotional approach (Alalsheikh 2016, 62; Morishita 2015, 157). Omotenashi is regarded as the dominant service concept in Japan because it invites visitors to undergo spiritually enriching experience while separating themselves from mundane pressures and worries (Alalsheikh 2016, 62; Morishita 2015, 157). This theory echoes the government's employed rhetoric of inviting guests to experience omotenashi in person. However, it is noted that foreign customers usually do not demand impossibly extensive service to feel content whereas Japanese customers who are familiar with *omotenashi* treatment expect fancy service

²² This narrative is similar to the "Cool Japan" cultural policy and *Nihonjinron*, check Chapter 5 for more.

and are quick to write one-star reviews (Shinchi, 26 Feb. 2023). Lina feels this struggle as she tries to balance giving her best service and dealing with angry complaints:

Yes, it must be perfect. When I see complaints on the [information] board, even though [the service] is not always perfect, [in the end] it comes to the guest how tolerating they are, whether they accept that mistake or will just complain about it. (Lina, 31, Lithuania)

The main concept of *ryokan* is to let customers immerse themselves in unique atmosphere while experiencing ritualized and traditional services which follow the old set of traditions. The beauty is perceived to be found in little things that usually go unnoticed in the fast-paced consumerist modern society (Alalsheikh 2016, 62). Services differ from *ryokan* to *ryokan*, but common traditional experiences include sleeping on *futon* (布団 mattress) on the floor of a *tatami* room (畳 traditional bamboo mat), soaking in *onsen* (温泉 mineral hot spring) and enjoying traditional local cuisine in traditional Japanese garments (Choi, Meng, and Lee 2018). These services create a unique and exotic experience for foreign customers unfamiliar with Japanese culture (Choi, Meng, and Lee 2018). Other traditional attributes are listed as removing shoes indoors (Choi, Meng, and Lee 2018). However, as the staff might not share common language or share cultural expectations with their guests, guessing customer's needs becomes challenging (Shinchi, 26 Feb. 2023). Enkthuya shares this experience and narrates how the host must carefully consider the interaction because despite the goal to present Japanese authentic culture and hospitality, *omotenashi* also includes the aim to make the customer feel at ease:

Guests do not understand a lot of Japanese things, [and then] come to a *ryokan*. Because they are foreigners, things that are taken for granted inside Japan aren't obvious to them. (...) "Do like this, don't do like that", they don't come to the *ryokan* to be preached. They come to have a good time, and if that is not allowed, it makes them feel bad. In situations like that you need to be careful how you speak (to customers). (Enkhtuya, 29, Mongolia)

Japanese language is regarded to have one of the most developed honorific systems (Brown 2008, 1). Honorific speech is stressed as an integral immutable part of Japanese language and culture (O'Neill 2015). Japanese respectful language (*keigo*) is also distinguished as the core feature of service culture in Japan. Respectful speech is established as means to show respect to the other party, usually higher in the hierarchy (O'Neill 2015). In the context of a *ryokan* and the service sector, the respectful language shows the host's reverence towards the customer. As this kind of respectful language is commonly used in the service sector, non-native workers need to make extra efforts to meet the traditional hospitality expectations

imbued in the expected use of language. Meiling recalled how she had no choice but to learn Japanese as she had to instruct the customers in Japanese: "Also, because I work in a *ryokan*, I use pretty much Japanese [always], so it is very important. That's why I gradually became good at speaking Japanese." (Meiling, 22, Taiwan).

The use of *keigo* reinforced how the customer takes top priority (Alalsheikh and Sato 2015, 140). There is a proverb in Japan that states "Customer is a God" (お客様は神様です okyakusama wa kamisama desu), underlining the importance of showing respect and hospitality (EU-Japan Centre for Industrial Cooperation). It is narrated that even the word "okyakusama" (お客様 honored customer) in Japanese includes the meaning of veneration and extreme respect. Sometimes *ryokan* go beyond the hotel setting with their hospitality. As an example, guests might be offered services which originally did not belong to their accommodation plan (Alalsheikh and Sato 2015, 160). When I was working at the *ryokan*, my sister who came to visit me received this kind of utmost service as she was offered a breakfast which wasn't originally included in her accommodation plan. Maria also noticed this detail while staying in Japan:

[They are] pretty much treating the customer as a supreme human being (laughs). You get respectful language and lot of bowing and the body language is very... not restricted, but, you know, very meaningful, predetermined gestures. (...) And there is also a heavy atmosphere of professionalism to it, customer service people in Finland can talk to customers very casually whereas in Japan it's very important to remember to speak in respectful language. (Maria, 25, Finland)

Amy agrees and shares a view that Japan has a unique customer service culture and mentality which cannot be found in other countries. However, she also shares how putting customer on a pedestal is at odds with her own values:

Japan has a unique customer service attitude and unique way of doing service. If it was Malaysia, it wouldn't be that polite. That kind of polite dealing with customers is decreasing everywhere else than in Japan. (...) But I'm not going that far and use respectful language myself (尊敬語 sonkeigo) while serving customers (laughs). (...) I think customers and staff should have an equal relationship. That's why if there is a customer with a really bad attitude, my attitude also worsens (laughs). (Amy, 21, Malaysia)

It should be noted that *omotenashi* is distinct from western hotel chains, where the hierarchical gap between staff and customer is immediately noticeable. In addition, despite the earlier statement that respectful speech in Japanese hospitality creates hierarchy, it is argued that the face-to-face hospitality presented at *ryokan* does not reveal top-down

hierarchies as the host and the guest are engaged in more intimate face-to-face service, thus putting them on equal ground (Alalsheikh and Sato 2015, 155; Morishita 2015, 157). *Omotenashi* is reckoned to be the embodiment of high-quality service based on Japanese customs achieved via interaction between service providers and customers (Morishita 2015, 157). Enkhtuya agrees and shares her motivation, at the same time illustrating what kind of interactions she shares with guests:

There are often foreign visitors who want to learn [characteristics of Japanese culture]. I am in contact with these guests, and it is important to present [Japanese culture to them]. "This is how you eat *sashimi*; this is how you wear a *kimono*; this is how you wear a *yukata*". When I explain things like that to customers, I like to do it to customers who don't speak Japanese (and do not understand Japanese culture well). (Enkhtuya, 29, Mongolia)

Moreover, *omotenashi* is not always standardized ritualized service based on the traditional values but oftentimes the hosts are required to have a keen eye to spot any problems the guests might encounter. The host's responsibility is to keep the guest relaxed and provide them an unforgettable and pleasant stay. To achieve this goal, staff members are required to develop communication skills and be adept in realizing guests' needs and comfort levels through joint conversations (Morishita 2015, 158). Joint conversations are not only related to the presentation of authentic Japanese culture, but they also involve small talk with guests. For example, during my *ryokan* internship, my co-workers also instructed me and taught me to talk with our customers, asking whether the food is delicious amongst other things. It is also worth mentioning that while *ryokans* embody traditional values and some of them are designed to be minimalistic, they are not modest accommodations like the description "traditional inn" might suggest; some famous *ryokans* are distinctly high-class facilities with luxurious interiors surrounded by nature and located picturesque locations (Alalsheikh 2016, 62). The promise of a luxurious and memorable experience, as Meiling's comment implies, seems to put a lot of pressure on foreign workers:

[The work] is not complicated, but it's pretty tough. Our *ryokan* is quite expensive, so rich people come to our *ryokan* with the desire to receive top-level service. That's why I have to do my best to make people feel top-levelness, whether it's the way I speak, explain the dishes, or the way I serve. (Meiling, 22, Taiwan)

Miura (2012, 93) argues that an essential part of the competitiveness of Japanese companies is the fact that their competition is focused on quality rather than to the price of the commodity itself. Regardless of the international or domestic market, the striving for perfectness in quality is also visible in the Japanese hospitality industry. Meiling (22 years, Taiwan) ponders on this issue while contrasting it with her home country: "Japanese way of working, right. Compared to Taiwan it is different. Taiwan's way of working is speed, whereas the Japanese way of working relies on small and detailed things, so the way of thinking is a little bit different." In a similar manner, other ethnically Chinese immigrants associate Japanese economic success with strict and serious work culture (Liu-Farrer 2011, 1). Similar to the constant scolding detailed in Sociologist Liu-Farrer's study of labor migration from China to Japan, my informant Meiling shared her own experience of discipline in the service industry:

Even [though I can speak] in Japanese, I don't know everything about Japan. People [at work] always say: "This is Japanese culture, you have to know it, this is (Japanese) tradition, you have to know it", but everything is a new thing [for me]. (...) I cannot understand everything, I feel confused, [I feel that I am] not really good at the job because you [are] always scolding me, that [it] is my fault, it makes me unconfident. (Meiling, 22, Taiwan)

She is expected to know Japanese culture thoroughly despite coming from a different cultural background, and the penalty for lack of knowledge is heavy because there is no room for mistakes. The unconditional aim for perfectness and quality service at *ryokan* caused Meiling to quit and want to change to another industry:

I want to go back to my [home] country. I do not like what they did to me... I try my best in this company but they [are] always scolding me, no compromises for me so I quit. (...) [I am] not running away, I just changed my mind. I want to work in another industry, not the service industry. (Meiling, 22, Taiwan)

This is also a direct manifestation of the situation where much like how pop cultural commodities and cultural policies such as "Cool Japan" are used to define Japan's special cultural traits and the borders of Japanese culture to "outsiders", advertising and worldwide promotion of *omotenashi* and traditional Japanese culture are likewise methods for the same kind of identity fortification and preservation against occurring internationalization. However, at the same time this highlights how foreign female workers are pushed into a peculiar situation when they are expected to promote and preserve Japanese culture, forcing them to present "authentic, homogenous and unique Japaneseness" while not sharing the nostalgic past. This attribute becomes even more prominent in the multiculturalizing hospitality industry which increasingly needs to draw on the labor of foreign workers due to Japan's tourist boom (Nguyen 2020; Sugiura, 17 2019) and Japan's self-essentialization emphasizes the borders between different ethnicities and cultures.

In the Travel and Tourism Competitiveness Index, it is especially the selfless treatment of customers, one of the unique qualities of Japanese hospitality referred to above, that puts Japan to first place in the "Treatment of customers" category (Consulate General of Japan in New York 2015). Treatment of customers is based on anticipating an exceeding customer's wishes (Shinchi, 26 Feb. 2023), which creates pressure on service workers, especially ones like Meiling who feel that they do not understand Japanese culture as a non-native and thus do not possess the ability to anticipate customer's needs the same way native Japanese can. However, the reverse is also true; foreign customers might not understand or recognize the specialties of Japanese-style hospitality. Even though Japan's economic recovery puts a high value on tourism, there has not been much indication of *omotenashi* being a major point of attraction for foreign guests. When the respect for and expectations of omotenashi are lacking, this high-class service cannot be reflected in the accommodation price. As hinted earlier, the COVID-19 pandemic, tight border controls and restrictions on tourism hit the hospitality industry hardest and made many lose their jobs. In order to halt the infections and with the reduced workforce, companies have had to readjust their service strategies. It is reported that despite one ryokan discontinuing the carrying of customers' luggage due to the risk of infection, customer satisfaction did not decrease. Despite the government's attempts to frame omotenashi as a unique defining factor of Japaneseness and Japanese culture, in the face of the reality of a shrinking population and a four-decade-high ratio of new jobs to applicants, companies are forced to ponder whether excessive omotenashi practices are worth to uphold (White, 5 Nov. 2018).

6.2 Service Sector as a Gendered Domain

Before the impact of the COVID-19 pandemic, Japan's employment rates steadily rose between 2013 and 2019 due to the women and elderly people based upward trend in labor force participation (Kotera and Schmittmann 2022, 5). Women's presence and agency in the workforce were especially supported by the leading Liberal Democratic Party to make up for the looming threat of an aging society, shrinking population and the future reduction in the volume of the workforce (e.g., Demelius 2020; Kingston 2019; Nakamatsu 2014). According to the Statistics Bureau, the total labor force amounted to 67.10 million in 2022 which is around 60.3% of all people over 15 years (Statistics Bureau 2022, 126). Japan's biggest industry was the tertiary industry which employed a whopping 73.4% of all eligible workers

and provided 1.8% of Japan's GDP (Statistics Bureau 2022, 30–32). In comparison to the statistics from before the pandemic, accommodation and food services currently account for 50.9 % of the volume of the tertiary industry despite the decrease in GDP and employment rate (Statistics Bureau 2022, 128). Of all the workers working in the accommodation and food services sector, 62% are women (Statistics Bureau 2022, 128). Thus, it can be deduced from governmental data that female workers are overrepresented in the hospitality sector.

In light of the Japanese government's data, the current biggest occupational category for women is reported to be service work, which is around 68.5% of all occupations they are holding (Statistics Bureau 2022, 128–130). As for the occupations women are employed as service workers, women tend to work predominantly in female-dominated occupations like elderly care, nursing, and teaching. Department stores and retail also employ a substantial number of female workers (Yoshida 2011, 219). Moreover, these aforementioned fields and occupations are considered low-wage employment without strong financial security (Cook 2018, 130; Gottfried in Vosko, MacDonald and Campbell 2009, 79; Osawa, Kim and Kingston 2013). Minjun who worked for three years for a *ryokan* in the *Kansai* area raised the issue of low salary during our interviews in autumn 2021: "The working hours were really long but the salary was very low." (Minjun, 27, South Korea)

I have kept in contact with Minjun, and I was surprised when, in early 2023, she wanted to have a sudden in-depth discussion about the very low salaries and the exploitation of the service sector. Out of the blue, the owner of the *ryokan* (社長 *shachō*) where she used to work contacted her to ask her to come back to work now that the pandemic had calmed down. She told me that she declined the offer as she had already stopped considering working in Japan. In the previous interview in fall 2021 she had mentioned on multiple occasions her dissatisfaction with low wages. In 2023, she asked me "But the salary was too low, right, you thought the same?" to which I had to answer "No, I do not know the amount of the salary as my internship at the *ryokan* was an unpaid internship". I proceeded to ask her about their monthly salary and at that time she told me:

"Salaries at the *ryokan*s are already low, but it was 190.000 yen per month in that *ryokan*.²³ It is the same for everyone, not just for foreign workers. That's why many people have to work in two jobs, a full-time job at a *ryokan* and a part-time job elsewhere. It is very exploitative.²⁴ That's also why many workers quit

²³ About 1400 USD as of May 4th, 2023.

²⁴ She felt it is very exploitative and used word "black business" (すごくブラック企業 sugoku burakku kigyou) to describe the relation of long working hours and low salary.

working at the *ryokan* quite fast, you cannot live with that salary." (Minjun, 27, South Korea)

She added that only she and Enkhtuya received higher salaries (250.000 yen)²⁵ due to their special type of work (marketing and translation to foreign customers and travel agencies). She also mentioned that at the time of her departure, the boss was even willing to do salary negotiations because "if [she] quits it will be very troublesome for [the *ryokan*]".

As discussed earlier, longing for a past that individuals have not experienced personally has been employed in the form of collective national nostalgia by conservative political actors as a way to unify the nation during uncertain times among societal changes (e.g., Elgenius and Rydgren 2022; Grainge 1999, 621; Smeekes and Jetten 2019). To preserve Japanese identity, the Japanese government has also given into national nostalgia rhetoric and utilized Japan's culture as a medium to brand-nationalize the aspects of "authentic" Japanese culture to attract visitors to Japan. One of the employed concepts, *omotenashi*, is closely related to the notion of care, and the care itself in Japan is heavily reliant on women's gender roles and uncompensated care in the public (labor market) and private (household) spheres as explored in Chapter 3. Up until now, the breadwinner husband model together with the lingering effects of ideologies such as ryōsai kenbo ("Good Wife, Wise Mother") and the government's insistence on safekeeping the patriarchally constructed societal structures still reproduce and influence perceptions of suitable domains for men and women. The way the labor market is structured, and the gendered expectations of society not only affect a person's choice of the field of employment but also the type of work. In Japan, due to the needs of social institutions, nurturant work which involves care is understood as innately feminine (Morris-Suzuki 2015b). In the case of service work, the reproductive work performed there is often seen as the female domain because of the connected "feminine nature and qualities" (Morris-Suzuki 2015b; Parreñas 2012). Minjun admits that during her career, she has experienced the gender expectation-based discrimination in the labor market:

Because I am a woman myself, I think women have fewer working opportunities than men. (...) There's work that needs power which women can't do, so if we talk about that kind of heavy work, women can't choose those jobs. When I am searching for a job, machines, automobiles, those industries (cannot apply to those fields) ... Because I experienced first-hand that men are more desired [there] than women, I think job hunting chances are worse. (Minjun, 27, South Korea)

-

²⁵ About 1900 USD as of May 4th, 2023.

Parreñas explains that men take part in institutional service work as janitors, teachers, or chefs (Parreñas 2012, 272). However, in the rare cases where men's produced work is also nurturant (perceived as feminine), the work of men is perceived instead as a breach of constructed gender order (Parreñas 2012, 272). For example, men who work in nurturant jobs like kindergarten teachers are shunned (Osaki, "Japan's male babysitters."). Nevertheless, the evidence shows that men predominantly perform "non-nurturant reproductive work" in the service sector like janitor, gardener, and food preparation (Parreñas 2012, 272). The same applies to hospitality which is seen as an innately nurturing job requiring feminine capabilities. Minjun talks about gender stereotypes and prejudice:

There is prejudice. If you are a man, you need to be strong. If you are a woman, you must be delicate. Because you are a man, you must be able to carry this much or if you are a woman, you should be able to do it more beautifully and so on – in Japan it is the same. (Minjun, 27, South Korea)

Alongside with the fact that nurturing work is seen as feminine, females are principally identified as preservers of traditional culture (Mosedale 2011, 128). In the context of *ryokan* and hospitality, women situated in the old-fashioned atmosphere serving others are argued to embody and reproduce something the contemporary society lacks (Yoshida 2001, 363), creating a connection to feel nostalgic to times gone by. Opposed to the men who are also allowed to work at hotels, most of the staff members at *ryokan* are women (Yoshida 2001, 362). Most of the service is done face-to-face is led by the *ryokan*'s *omotenashi* manager, *okami*, who together with female workers are perceived to be an "epitome of politeness" promoting the authentic traditional culture in the rapidly changing Japanese society (Yoshida 2001, 361). In addition to women's role as preserver of culture, this indicates that there is an underlying stereotype of women as excellent servers, making women superior to men in polite customer service because of their innate caring skills. Meiling agrees and speculates about the roots of these kinds of stereotypes:

[Because of] the old traditional thinking, they think girls and women have to serve and men to do some heavy work. (...) Men can do other type of work (at *ryokan*), but not onee-san (serve customers personally). (...). (Meiling, 22, Taiwan)

The aforementioned qualities have been influenced by the "Good Wife, Wise Mother" ideology that has been popular among male politicians who have supported the definition of women responsible for household management and child nurturing (Goldstein-Gidoni 1999). Even though the characterization has its roots in the Western model of ideal Christian woman, the concept has seized a considerable hold on the official discourses about women in Japan

(Ueno 1993 in Goldstein-Gidoni 1999). Parreñas argues that the majority of work done by foreign (migrant) women is non-relational and thus not fitting the definition of traditional care work (Parreñas 2012, 272). Foreign workers conduct reproductive work which involves a broader set of activities like washing clothes, sweeping floors, socializing, and preparing food (Parreñas 2012, 272). Furthermore, the care presented at *ryokan* very much resembles the household chores women are expected to provide at home, thus *ryokan* can be viewed as an extension of women's household domain. To support this fact, the domestic services of *ryokan* like serving food and cleaning are argued to establish a home-like atmosphere (Alalsheikh 2016, 62; Alalsheikh and Sato 2015, 126). It is often claimed that women have more caring skills and a natural aptitude for caring, hence they are declared to be superior in work which requires face-to-face interactions like those demanded in the hospitality industry (Mosedale 2011, 128). This creates a very gendered division of labor revolving around the government advocated gender roles. Minjun provides more insight into that stereotype by pointing out how people feel that women are easier to approach:

(...) if something is different (in the service sector), it's that work that requires power is done by men, for example carrying customers' luggage. And if a customer has a problem, if they have to ask in person, customers feel more relieved if it's a woman, so they end up going to a woman. (...) If it's small children, it is not a rare case either, but if it's a woman, they feel safer. (Minjun, 27, South Korea)

Meiling similarly voices her opinion:

They need girls to serve you dinner, lunch, breakfast, it is more comfortable like that. If you come to a hotel and you see a lot of men, it is not a good feeling, right? You see a lot of women; you feel more relaxed and comfortable. (Meiling, 22, Taiwan)

In many cases, in these fields that are seen as "feminine jobs", female workers are surrounded by mostly female colleagues (Yoshida 2011, 219). The hospitality industry belongs to the tertiary industry and is a prime example of service work dominated by women. In the case of my fieldwork and interviews, the same trend was indicated by the results. Enkhtuya who has worked nine years at a *ryokan* admits after discussion that despite the work itself not being easier for a man or woman, it is a fact that there are more female workers at service sector and at their *ryokan* compared to men: "Men are also working at the *ryokan* (...) But maybe there are a little bit more women, I also think that too. (...) Fundamentally more women working at hotel industry, at service sector." (Enkhtuya, 29, Mongolia)

The above statistical numbers prove that there is a clear tendency for women to be driven in certain fields due to their combability with traditional gender roles. As noted earlier, Japanese labor markets are conservatively built upon lifetime employment and the male breadwinner model (Kotera and Schmittmann 2022, 4–5; Ueno, 2021). The statistics seem to be supporting the fact that the Japanese labor market's gender segregation is blatantly apparent as many companies have separate employment tracks for male and female workers (Assmann 2014, 8; Ueno 2021, 13; Yoshida 2011, 219). Men follow the career track and receive the benefits of lifetime employment while women are forced into the clerical track with fewer options for promotion and are expected to quit work after having children (Cook 2013, 130; Ueno 2021, 13; Yoshida 2011, 219). Numerous other aspects like insufficient childcare support arrangements, long working hours, and dual-track system make many married women find it easier to return to work life as part-time workers or take up non-career track positions rather than a full-time position because of the obstacles based on the artificially constructed gendered labor system (Kotera and Schmittmann 2022, 4-5). Around 70 percent of nonregular workers are women (Kotera and Schmittmann 2022, 4–5; see also Broadbent, 2003). The life course of many women is still ruled by the constraints of labor markets built upon the traditional expectations and gender roles that the Japanese government tirelessly promotes (e.g., Marshall 2017; Miura 2012; Ueno 2021). My informant Amy who studies international law and equality issues in Japan is rather conscious of the labor market issues and the prevailing constructions of a breadwinner salaryman husband. She proposes reasons that force women to choose the service sector and part-time employment over their careers:

I think there are more women [in the service sector]. And if we ask why, to be honest, it's because women are almost always housewives, they work [part-time] as married housewives. If they marry, birth to children, full-time work is quite impossible for them. They have to raise children, in Japan men still (...) work full-time and housewives only go to work from two to three times a week. (...) Childrearing (子育て kosodate) is the cause of why there are more women in the service sector... and why they work in hospitality industry. (Amy, 21, Malaysia)

Omotenashi, which acts as the base of serving customers in the hospitality sector is closely connected to traditional Japanese tea ceremony practices (Takeda et al. 2016). During the tea ceremony, omotenashi is applied to show care and attend to the guests with utmost hospitality (Takeda et al. 2016). It is emphasized how the tea ceremony host must attend to the ceremony attendant thoroughly and ensure through and through an enjoyable experience (Takeda et al. 2016) which can also be applied to the case of a *ryokan*. In Japan, women are also often presented as bearers of tradition while men are seen as agents of the public sphere (Morris-

Suzuki 2015b). The gendered, nurturant and gentle stereotype associated with women was evident in many of my informants' answers. Meiling talks about the gendered nature of servicing and how women are perceived to innately be more careful, thus conducting their service top-notch:

In the service industry, there [are] a lot of women do[ing] this [type of] job because this job needs a careful person to do it. If you do not care about that, you don't notice small things. If [you] need to serve a plate, a girl thinks "how can I make the customer [feel] more relaxed", so they do it more carefully. So, girls [are] more suitable to do this job. If boys do it, maybe they (customers) think boys [are] disgusting if they do it, they think that man is not a man, [that it is] women's job. (Meiling, 22, Taiwan)

Traditional *ryokans* usually create a classy atmosphere not only through professional serving and traditional crafts, but also clothing their workers in traditional attires like kimono and hakama. According to kokusaika principle, the Japanese government has appealed to companies to make Japanese culture and for example souvenirs more exotic and appealing to tourists, and one way to achieve this is argued to be wearing an ethnic dress (Flowers 2012). Similarly, in a *ryokan*'s context, this ethnic dress can be perceived to be the kimono. This Japanese desire is speculated to deny the ethnic other at the expense of boosting tourism and profit (Flowers 2012). Goldstein-Gidoni emphasizes that in modern Japan, women clad in kimonos have been transformed into symbols of Japaneseness (Goldstein-Godoni 1999; see also Ashikari 2003, 72; Assmann, 2008, 360). The agenda has been deliberate as the kimono itself has been framed as a feature of Japanese uniqueness (Assmann 2008, 362; Goldstein-Gidoni 1999). Kimono is endorsed as a cultural distinction dividing the Japanese and Western (Goldstein-Godoni 1999). Furthermore, it is argued that the modern Japanese identity is built on the clear distinction between Western and Japanese culture (Assmann 2008, 370; Befu 2001), it is also recognized to be intimately related to reproducing traditional cultural roles and distinction for genders (Goldstein-Gidoni 1999). One representation of this is how during university graduation men are allowed to wear Western attire whereas young women are seen as proper models of femininity only while wearing a kimono (Goldstein-Gidoni 1999; see also Valk 2018; Foreman 2011 in Bardsley and Miller 2011). This kind of mindset is also observed in the traditional ryokan setting as during my internship, I saw that women working in face-to-face interaction at the front were clad in kimonos while serving the customers. Foreign workers clad in kimonos brought customers food or sometimes if cleaning was required, the kimono was mixed with hakama pants to give more mobility but not remove from the "authentic Japanese atmosphere". In short, the household sphere-based gendered

division of labor and utilization of traditional Japanese items like kimono not only serve as a distinguishing force dividing Japanese from foreigners (as kimono is "unique" to Japan) but seem to also indicate that the government attempts to negotiate and frame Japanese culture by means of authenticity and uniquity. In their opinion authentic Japaneseness is found "in the past traditions", and this nostalgia is triggered in response to the threat to the continuation of "Japanese national identity due to the foreign cultural contact at home and abroad.

In addition, the conventional gendered division of labor also made women more vulnerable to the impacts of the coronavirus pandemic (Fukai, Ichimura and Kawata 2021; Kobayashi et al. 2021; Kotera and Schmittmann 2022). Japanese economy and labor force participation rates shrank for the first time since the steady growth of 2013, and accommodations, drinking and eating services especially bore the brunt of the impact due to the restrictions on non-essential and non-urgent long-distance travel, domestic commuting, and the push for social distancing (Fukai, Ichimura and Kawata 2021; Takahashi 2021, 13). Overall, around 870.000 people became unemployed (Ishibashi, 17 Sept. 2020), and as a result of these government imbued pandemic restrictions, it was recorded that workers in accommodation and food services declined by 250.000 (Takahashi 2021, 13). Women whose presence in the accommodation and food services are prominent are regarded to be the most vulnerable to unemployment because of the pandemic as they usually are employed in very contact-intensive sectors with no option for remote work, unlike the option full-time company salarymen have (Fukai, Ichimura and Kawata 2021; Kobayashi et al. 2021; Kotera and Schmittmann 2022, 2).

The numbers of infected persons are still rising, so it's a little scary all the same. (...) when I am going to work, I use the train, not every train but many of them are crowded with people returning home (...) There are also many people at the workplace, it's the scariest because you do not know who the carrier is (...). There are also the matter of families and people who live together, you might get infected, and you might infect others, I worry about that. (Enkhtuya, 29, Mongolia)

Many scholars have proclaimed that COVID-19 pandemic has affected men and women differently; while men suffered job-related worries like work-style changes or economic insecurity, women were reported to suffer from stress factors unrelated to work such as increased living costs and limited social interactions (Kobayashi et al. 2021). The case of the coronavirus pandemic is also special as previous economic recessions have caused impact in suicide among men, however, for the first time female suicide rates rose (Kobayashi et al. 2021). Women, young people, and non-regular workers are also vulnerable to economic

hardships due to lower average salary (Kobayashi et al. 2021). They also experience the increasing demand for household support as they are responsible for taking care of children and elderly relatives (Kobayashi et al. 2021). Minjun had just quit her *ryokan* job and was in the middle of job hunting in Japan when the pandemic hit Japan. She felt stressed and the situation eventually made her return to South Korea a little bit earlier than she had first intended:²⁶

I wasn't in the country for that long when the pandemic started (...) but similarly to other places in the world there were lots of effects to life. I had just quit working at *ryokan* and was searching for a job (...). It caused worries, of course; I had to change jobs fast, because if you cannot earn money, you cannot live... There can be circumstances where you have to return to your home country [because you don't have enough money]. (Minjun, 27, South Korea).

Enkhtuya recalls feeling a little bit depressed and lonely because she was unable to meet her family during the pandemic: "Sad feelings... Well, this year (because of corona), I haven't been able to meet my family, so that is lonely." (Enkhtuya, 29, Mongolia)

In addition to women, foreign residents (who are much more vulnerable because of their status as outsiders) seem to share enhanced stressors and vulnerability. Japan's immigration policy is described as a "no-immigration policy" (Burgess 2022; Chung 2010; Flowers 2012). Japanese policymakers have vehemently denied Japan being an immigration country, opting to avoid the term "migrant" and instead referring to "entrant" or "foreign worker" (Burgess, 2022). The whole immigration system is based on the assumption that foreigners won't stay long or settle down, rather it is expected that after their usefulness they will be discarded (Burgess, 2022). Based on the wide support for the government's hardline on border controls in response to the coronavirus pandemic, it cannot be denied that the "island-nation mentality" hasn't disappeared; there is an indication of stable support for isolationism and protectionism (Burgess 2022). Chris Burgess examined that numerous foreign sojourners have experienced discriminative behavior in Japan during the COVID-19 pandemic (Burgess, 2022). For example, they were vulnerable to reduced wages or unemployment as well as inadequate access to support (Burgess, 2022). It was pointed out that Japanese-style multiculturalism had a significant role in worsening the pandemic's impacts because Japanese-style multiculturalism itself is a non-integrative policy that does not support empowering foreigners nor aid them in obtaining social capital and becoming responsible and

-

²⁶ Initially Minjun had thought that she would return home before she turned 30.

independent members of Japanese society (Burgess, 2022). Pre-pandemic Japan made advancements towards opening up from its isolationist island nation mindset which has complicated their handling with foreign migrants, but all the progress seems to have halted or regressed (Burgess, 2022). Burgess also hypothesizes that these struggles of providing support and the weakening yen may have cost Japan its attractiveness as a destination to work and live among non-Japanese (Burgess, 2022).

In the interviews, none of my informants addressed the gender disparities found in the Japanese labor market directly. Therefore, it is safe to argue that different gender expectations – like men need to be strong and for that reason women as less strong individuals cannot apply to certain type of work – is a representation of gender disparity. The pandemic countermeasures and tight border control issued by the Japanese government have been proven to have had a negative impact on foreign labor (Fukai, Ichimura and Kawata 2021; Kobayashi et al. 2021; Kotera and Schmittmann 2022). None of the informants had not needed official help, thus indicating that the municipality's efforts to alleviate problems among residents were insufficient and many did not seem to know what kind of support they could request, making municipal services seem unapproachable. The majority of the study subjects cited worries about salaries and infecting others which are common findings on the effects of the COVID-19 pandemic in Japan. However, as established earlier, these kinds of problems among foreigners are not an uncommon occurrence, rather COVID-19 just heightened the persistent struggles foreigners share in their everyday life.

7 Conclusion—"Women and Men Do Not Have the Same Expectations"

In this master's thesis, I have demonstrated how Japan's national nostalgia based political narrative manifests in the country's hospitality industry by constructing an extremely gendered domain, affecting foreign female workers via enhanced traditional gender roles and how these same narratives about "uniqueness" and "homogeneity" serve as a major obstacle hindering Japan's aspirations from realizing tabunka kyōsei society. As a data collection method, I conducted semi-structured online interviews among foreign female workers who were working in Japan's service sector together with my own internship-turned-fieldwork experience at a traditional ryokan in Kansai area. As for data analysis, I utilized the thematic analysis method based on the interview materials to interpret my data. I applied the lens of "national nostalgia" (Smeekes, Wildschut and Sedikides 2021) as this study's framework together with the researcher's positionality (Ali 2014; Ali 2015; Narayan 2003) to evaluate the data, and special emphasis was put on the concepts of uniqueness, hospitality, and care work. While analyzing the results, I found a correlation between Japan's collective identity building in response to the threat of change, the strict gender binary and the low integration of foreign workers. These findings show that despite nationalistic narratives underlying Japan as a monoethnic, homogenous and unique society similar to Nihonjinron have transformed into new forms of nationalism since their peak years, remnants of those ideologies are still relevant to contemporary Japanese society via collective national nostalgia. Furthermore, the ideologies are constantly reproducing and enhancing Japan's own self-image as a "unique country and culture" which obstructs multicultural society building by reinforcing the ingroup and outgroup borders based on ethnicity.

This research was able to provide several contributions. Firstly, there is an abundance of previous research focusing on foreign female workers on entertainment and sex industry such as hostess clubs (e.g., Aoyama 2009; Kamise 2013) and of foreign nurses (e.g., Efendi et al. 2022; Świtek 2016), but very little research is done from the point of view of the Japanese hospitality sector. My study managed to fill this gap in the existing body of literature and provides an opportunity to build upon this topic in the future. Moreover, even though these women share some common realities and problems while residing in Japan, the employees in the hospitality industry have different contexts from those of entertainment workers. For example, they do not bear the same stigmas, but struggle with strict work expectations,

responsibilities, and societal inclusion on other fronts. As Ali (2014) affirms, compared to men, women must balance various expectations related to home and work. This aspect makes women's experiences, especially foreign ones, a fruitful group to study in a multicultural setting. In addition, Japanese *ryokan* and *omotenashi* culture as an extremely gendered domain reproducing traditional values and nation-branding is not much researched in the West.

Secondly, there are few existing studies conducted on the influence of national nostalgia narrative in an East Asian context. The preponderance of existing studies present the national nostalgia narrative that the populist radical right-wing political employ in an American and European context (e.g., Behler et al. 2021; Elgenius and Rydgren 2017; Elgenius and Rydgren 2022; Smeekes, Wildschut and Sedikides 2021; Smeekes and Jetten 2019). My findings suggest that Japan's right-wing leaning leading Liberal Democratic Party's political rhetoric has affinity to a same kind of narrative of "homogeneity" and "monoethnicity" of the past to defend the ethnic and cultural borders of Japan. Hence, I contributed to the national nostalgia scholarship by offering a glimpse of how this same narrative is utilized in a Japanese context to define "Japanese collective identity" and the characteristics that belong to it, and at the same time resulting in augmented and reinforced borders between the Japanese ingroup and the foreign outgroup.

Thirdly, despite *Nihonjinron* ideology being obsolete and debunked by several academics, there seems to be a research gap in presenting how *Nihonjinron* still significantly impacts Japanese society by setting boundaries between Japanese and foreigners, making it still relevant to discuss in a contemporary context. As Befu (2001, 44) noted, the absence of women in *Nihonjinron* serves as a way to challenge the "hegemonic" Japanese identity and turn the scope to the matter of multiculturalizing Japan which the national government has been turning a blind eye to. *Nihonjinron* ideology has transformed several times and is so embedded in the contemporary Japanese narrative advocated by the Japanese government that various fragments of the ideology about homogenous and "racially monoethnic" society with unique culture and customs are found in many facets of the society. Even the *tabunka kyōsei* policies aimed at the inclusion of foreign residents highlight the cultural borders of Japanese and non-Japanese, resulting in the exclusion by inclusion (Burgess 2007; Burgess 2012; Chapman 2006). The close affinity of *Nihonjinron*, national nostalgia narrative and *tabunka kyōsei* prove that they are just the newest manifestation of *Nihonjinron*. Therefore, I have provided the female perspective to *Nihonjinron* which has been lacking previously and

presented how the ideology still manifests in different contexts to strengthen Japanese identity and othering the foreign presence, obstructing Japan's transformation to a multicultural society.

For the abovementioned reasons, my research data provides a nuanced picture of the foreign female workers, their experiences and how traditional gender roles are embodied in Japan's multiculturalizing hospitality sector. In addition, I was able to present some of the multifaceted causality relations between the Japanese national government's agenda, the gendered labor market, and how this narrative about society's "uniqueness" correlates with the resistance to society's multiculturalization and what impacts they have among Japan's increasing foreign population. The effects of national nostalgia rhetoric on Japanese national identity, traditional gender roles and foreign integration could be studied further in the future.

At a rhetorical level, the Japanese government harnesses Japan's national nostalgia to reimagine Japanese identity through shared features, a common homogenous past and by framing Japanese culture as "unique". This "uniqueness" is narrated to be the result of Japan being historically an island nation where foreign influences have not been reached due to its geopolitical isolation. However, this narrative contradicts Japan's multiculturalist past and the current reality of Japan fostering foreign immigrants and minorities, painting a picture of a homogenic society. In turn, this strengthens the strict borders between "ethnically Japanese" and the "foreign other", producing difficulties for immigration and implementation of a multicultural society. It is also noted that the multicultural policies sharing cultural distinctions highlight the differences between cultures, furthering the deep divide between the native and immigrant populations. Hence, a narrative similar to *Nihonjinron* continues to shape perceptions of the ethnic other and divide the society, blinding the Japanese government of its multiculturalist reality.

In regard to the extent to which Japan's national nostalgia towards the "homogenous and unique culture" affects Japanese society, I found it had a distinctive correlation with interactions and obstacles for multiculturalism in Japan. For instance, one finding indicated the importance of the Japanese language and networking with Japanese nationals, so one could argue that Japanese society is built on a power structure between foreigners and the ethnic Japanese. None of my informants had major troubles with the Japanese language nor considerable hardships. However, none of the members of my informant group had long-term plans to stay in Japan. Despite the need to network with Japanese nationals to acquire a

guarantor for housing, they tended to stay among people of similar backgrounds (i.e., other foreigners). Alas, despite being fairly privileged and not facing major troubles, my findings showed that my informants were not sufficiently integrated into society. All of my interlocutors underlined the need to know how to speak the Japanese language, and academic research supports this fact of the language barrier drawing distinctions between cultures (See Phillips 2007). The language was stressed to be the key to understanding Japanese "unique" culture and behavior. Another relevant point was that most of the informants did not know or had a superficial knowledge of the *tabunka kyōsei* policies and how they were implemented on a local level. When I asked them about what kinds of services municipalities provide to foreign residents, all of them voiced that they had not required assistance and listed briefly on a general level a couple of services they had heard of being provided for foreigners. This suggests that services which are fashioned after the *tabunka kyōsei* policies to make foreign residents integrate better into the host society are perhaps nonexistent, or at least hard to reach to the foreign population and inadequately advertised.

Furthermore, Japanese internal self-identity is found to be fluctuating between a positivenegative binary axis following the historical and geopolitical changes in the economy; during economically and politically prosperous times, these factors provided stronger national selfesteem, letting cultural nationalism to prevail and vice-versa (Befu 2001; Nederveen Pieterse and Kim 2012, 169). Postwar Japan set to mold its image into a more peace-loving and harmonious society through the employment of soft power (e.g., Befu 2001; Iwabuchi 2015; Valaskivi 2013). With the aid of Japan's soft power, this attempt to improve the image of Japan via cultural mediums and commodities seems to have succeeded as the research participants had very positive images of Japan and Japanese culture. Recently Japan has utilized its immense amount of cultural capital in campaigns such as "Cool Japan", which promoted Japan's popular culture. Despite the scarcity of academic inquiries into Japan's success in attracting visitors due to its soft power campaigns, in my research there were strong indicators between Cool Japan and motivation to come and experience Japanese culture in person. Japan has also attempted to define "authentic Japanese culture", and this national nostalgia-based narrative had shaped foreigners' impressions of Japan in a way that "ryokan can only be found in Japan" or "Japan has the unique service culture (omotenashi) which is not found in other parts of the world".

In addition, this same nostalgia-based narrative impacts Japanese society itself by creating expectations for genders grounded on traditional gender roles. As Amy from Malaysia put it:

"[In Japan], women and men do not have the same expectations". I examined this notion based on my own internship experience and the interview data I gathered, the results indicating a similar conclusion of a gendered society on the grounds of exceedingly prominent gender norms. In recent history, Japan's leading elite have nudged society towards gendered labor by supporting societal constructs such as the breadwinner model which cemented perceptions of woman's role in the household and advocated them as caregivers and providers of welfare through their uncompensated work. Since then, the notion of care has been inextricably linked with femininity and suitable gender roles, influencing women to contribute care work for society in the "feminine fields" both inside and outside the bounds of their domestic domain. As a consequence, the occupations related to care are perceived to befit women based on their feminine qualities. One of the questions I raised at the beginning of this study was how gender roles manifest in the hospitality sector. In the context of ryokan and hospitality, as Yoshida (2001, 363) pointed out, women are perceived to embody values that are absent in modern society, therefore especially the traditional ryokan is seen as an extension of the domestic domain due to the fact that the tasks there resemble household chores, reproducing the image of woman as feminine caretaker and household manager. The gendered labor is also visible in the roles of "onee-san" and "okami": as the ryokan aims to make the customer feel at home, ryokan takes advantage of the image of females as homemakers serving guests in a home-like environment. Furthermore, this ideology of "women being superior at caring and servicing" was evident in the stories my informants told me as they narrated that "if you have a problem while staying at a ryokan, the customer feels more relieved to seek woman's help and might feel disgusted if a man is serving them". In short, the gender roles manifest through the perception that the household is a women's sphere, caring is feminine duty and females are superior in giving service due to their innate qualities.

Another key question was how these reproduced gender roles and expectations affect foreign women. My informant group did not acknowledge gender disparities directly; however, society's gender experiences were visible through expectations. As they were working on the field themselves, they did not seem to notice the preference for female labor in the *ryokan* nor the expectation of women being face-to-face care providers for the guests, but when asked they could point out that men always carried the heavy luggage due to their strength and women needed to be more delicate and thoughtful. In addition, foreign women were expected to understand and convey the "authentic" and "unique" Japanese culture which *ryokan*

embodies and preserves. However, this put foreign female workers into a peculiar situation where they have to present authentic Japanese culture to foreign and domestic customers despite the Japanese public narrative othering and not integrating them into members of society. Additionally, it was acknowledged that, as a woman, it was acknowledged that women could not work in every profession and that Japanese society innately holds societal constructions which force women to choose to work in flexible positions like the ones the hospitality sector provides. On top of the strict and demanding work due to top-level service, the COVID-19 pandemic heightened the vulnerable positions of working women. The latest studies have found that female-dominated fields in Japan like the hospitality sector were hit the hardest, and the lack of tourists made several individuals lose their employment. Principally all the interviewed females were worried about the effects of the coronavirus, and one of them had experienced the stress of finding a good job during the COVID-19 pandemic.

Taking all these points into consideration, Japan's gendered hospitality sector and foreign women's experiences in the field reveal how Japanese hospitality is constructed on traditional gender norms. Furthermore, women are the ones expected to serve and provide care work, and this notion also affects foreign female workers through the perceived "feminine work", suitable gender roles and indirect expectations that foreign females must fulfill as women. What's more, these gendered societal norms and gendered expectations are constantly reproducing and enhancing conservative gender roles in Japan. My findings indicate that national nostalgia is indeed employed in the "narrative of Japaneseness" and influences it by shaping the definitions of "authentic" and "unique" Japanese culture. Likewise, this narrative of a "unique" and "monoethnic" Japan highlights the distinctions of ethnicities and cultures compared to Japanese culture, reinforcing the borders between Japanese and non-Japanese residents. From these findings, I conclude that these facts prevent foreigners' integration and realization of a truly multicultural Japan.

I acknowledge certain limitations in this study. The experience of conducting this research revealed several aspects which could have been executed differently; the interview questions could have covered fewer topics, concentrating only on hospitality workers, or alternatively, I could have included a wider age range or more diverse legal statuses, to mention a few. One prominent change in the execution of this research—which I briefly mentioned in Chapter 4—is the fact that I could have conducted a second fieldwork at the *ryokan* as a service employee together with my informants instead of simply utilizing interview materials, which would have offered a more comprehensive set of data. Secondly, as most of the data was gathered in

Japanese among non-native speakers of the Japanese language and translated by another nonnative speaker (yours truly), this thesis might include some miscommunication and misconceptions which derive from the lacking language skills. Thirdly, this thesis touches upon only specified parts of both gendered labor and hospitality sector along with the national nostalgia rhetoric, further research is recommended to build an even more comprehensive and multifaceted picture of this reality in Japan. In particular, the female-dominant ryokan could have been compared to male dominant service spheres such as the professional culinary art domain (AP, 14Aug. 2015), which would have bestowed a richer comparative picture of the evident male–female binary in the hospitality sector. Lastly, as established in this study, women are associated as "domestic care workers" and their care work is taken for granted, making them and their work rather invisible whereas men's work input as, for example, fulltime salarymen is visible to the general public. These visibility and invisibility dynamics could be tested in future studies to better understand the gendered work domains of Japan. However, despite these posed limitations, I firmly believe that this study has achieved its goal in shedding light on the extent Japan's national nostalgia imbued narrative of "authentic Japaneseness" prevents integration and increasing multiculturalization in Japan. In addition, it also presents how the gender roles manifest and affect the multiculturalizing hospitality sector, and the various realities and expectations the gendered domains induce on foreign female workers.

References

- Abe, Yukiko. 2011. "The Equal Employment Opportunity Law and Labor Force Behavior of Women in Japan." *Journal of the Japanese and International Economies*, 25(1): 39–55.
- Abe, Yukiko. 2013. "Regional Variations in Labor Force Behavior of Women in Japan." Japan and the World Economy, 28: 112–124.
- Adachi, Tomoko. 2013. "Occupational Gender Stereotypes: Is the Ratio of Women to Men a Powerful Determinant?" *Psychological Reports*, 112(2): 640–650.
- Adams, William. 2015. "Conducting Semi-Structured Interviews." *In Handbook of Practical Program Evaluation*, edited by Joseph S. Wholey, Harry P. Harty, and Kathryn E. Newcomer, 492–505. San Francisco: Jossey-Bass. Doi: 10.1002/9781119171386.
- Agyeiwaah, Elizabeth, Wantanee Suntikul, and Li Yee Shan Carmen. 2019. "Cool Japan': Anime, Soft Power and Hong Kong Generation Y Travel to Japan." *Journal of China Tourism Research* 15(2): 127–148.
- Akaha, Tsuneo. 2005. "Soft Power' in Japan's Security Policy: Implications for Alliance with the United States." *Pacific Focus*, 20(1): 59–91.
- Alalsheikh, Abdulelah. 2016. "An analysis on the Japanese Omotenashi from the servicedominant logic perspective." Kwansei Gakuin University Repository. https://core.ac.uk/download/pdf/186667074.pdf, accessed May 22, 2023.
- Alaslsheikh, Abdulelah, and Yoshinobu Sato. 2015. "Characteristics of the Hospitality, Omotenashi in the Traditional Japanese Inn: A Case Study of Kagaya." Kwansei Gakuin Repository. https://core.ac.uk/reader/143638136, accessed May 22, 2023.
- Ali, Rabia. 2014. "Empowerment beyond resistance: Cultural ways of negotiating power relations." *Women's Studies International Forum*, 45: 119–126.
- Ali, Rabia. 2015. "Rethinking representation: negotiating positionality, power and space in the field." *Gender, Place and Culture: A Journal of Feminist Geography*, 22(6):783–800.
- Anderson, Benedict. 1991. *Imagined Communities: Reflections on the Origin and Spread of Nationalism*. London: Verso.
- ANNnewsCH. "Takigawa Kurisuterusan no purezenteeshon IOC Sōkai" [Takigawa Christel's Presentation at the IOC Gathering]. Youtube video. September 7, 2013. https://www.youtube.com/watch?v=6hggygKWwhg, accessed, May 22, 2023.

- Aoyama, Kaoru. 2009. *Thai Migrant Sex Workers; From Modernisation to Globalisation*. Basingstoke: Palgrave Macmillan.
- AP. "Female sushi chefs muscling in one of Japan's last bastions of male dominance." *South China Morning* Post. August 14, 2015. https://www.scmp.com/news/asia/east-asia/article/1849273/female-sushi-chefs-muscling-one-japans-last-bastions-male, accessed May 14, 2023.
- Aronsson, Anne Stefanie. 2015. Career Women in Contemporary Japan: Pursuing Identities, Fashioning Lives. New York: Routledge.
- Ashikari, Mikiko. 2003. "The memory of the women's white faces: Japaneseness and the ideal image of women." *Japan Forum*, 15(1): 55–79.
- Assmann, Stephanie. 2008. "Between Tradition and Innovation: The Reinvention of the Kimono in Japanese Consumer Culture." *Fashion Theory*, 12(3): 359–376.
- Assmann, Stephanie 2014. "Gender Equality in Japan: The Equal Employment Opportunity Law Revisited." *The Asia-Pacific Journal Japan Focus*, 12(45): 2. https://apjjf.org/2014/12/45/Stephanie-Assmann/4211.html
- Bardsley, Jan, and Laura Miller. 2011. "The Perfect Woman: Geisha, Etiquette, and the World of Japanese Traditional Arts." In Manners and Mischief, 67–79. Oakland: University of California Press.
- Bauder, Harald. 2011. *Immigration dialectic: imagining community, economy, and nation*. Toronto: University of Toronto Press.
- Befu, Harumi. 2001. *Hegemony of Homogeneity: An Anthropological Analysis of Nihonjinron*. Melbourne: Trans Pacific Press.
- Behler, Anna Maria C., Athena Cairo, Jeffrey D. Green, and Calvin Hall. 2021. "Making America Great Again? National Nostalgia's Effect on Outgroup Perceptions." Frontiers in Psychology 12. Doi: 10.3389/fpsyg.2021.555667.
- Bélanger, Danièle, Kayoko Ueno, Khuat Thu Hong, and Emiko Ochiai. 2011. "From Foreign Trainees to Unauthorized Workers: Vietnamese Migrant Workers in Japan." *Asian and Pacific Migration Journal* APMJ, 20(1): 31–53.
- Bishop, Beverley. 2002. "Globalization and Women's Labour Activism in Japan." *EJCJS*. http://www.japanesestudies.org.uk/articles/Bishop.html, accessed May 22, 2023.
- Bishop, Beverley. 2005. *Globalization and women in the Japanese workforce*. London: Routledge.
- Braun, Virginia, and Victoria Clarke. 2006. "Using Thematic Analysis in Psychology." *Qualitative Research in Psychology*, 3(2): 77–101.

- Brinton, Mary C, and Nobuko Nagase. 2017. "The Gender Division of Labor and Second Births: Labor Market Institutions and Fertility in Japan." *Demographic Research*, 36(36): 339–370.
- Brinton, Mary C., Eunmi Mun and Ekaterina Hertog. 2021. "Singlehood in Contemporary Japan: Rating, Dating, and Waiting for a Good Match." *Demographic Research*, 44: 239–276.
- Broadbent, Kaye. 2003. Women's Employment in Japan the Experience of Part-Time Workers. London: Routledge Curzon.
- Brown, Lucien. 2008. "Contrasts between Korean and Japanese honorifics." *Rivista Degli Studi Orientali*, 81(1/4): 369–385.
- Bukh, Alexander. 2014. "Revisiting Japan's Cultural Diplomacy: A Critique of the Agent-Level Approach to Japan's Soft Power." *Asian Perspective*, 38(3): 461–485.
- Burgess, Chris. 2007. "Multicultural Japan? Discourse and the 'Myth' of Homogeneity." *The Asia Pacific Journal*, 5(3). https://apjjf.org/-Chris-Burgess/2389/article.html
- Burgess, Chris. 2010. "The 'Illusion' of Homogeneous Japan and National Character:

 Discourse as a Tool to Transcend the 'Myth' vs. 'Reality' Binary." *The Asia-Pacific Journal*, 9(1). https://apjjf.org/-Chris-Burgess/3310/article.html
- Burgess, Chris. 2012. "Maintaining Identities—Discourses of Homogeneity in a Rapidly Globalizing Japan." *Electronic Journal of Contemporary Japanese Studies*, May 29, 2012. http://www.japanesestudies.org.uk/articles/Burgess.html?pagewanted=all
- Burgess, Chris. 2022. "COVID-19 border policies strengthen Japan's insular mindset." *East Asia Forum*, October 12, 2022. https://www.eastasiaforum.org/2022/10/12/covid-19-border-policies-strengthen-japans-insular-mindset/
- Carroll, Tessa. 2010. "Local Government Websites in Japan: International, Multicultural, Multilingual?" *Japanese Studies*, 30(3): 373–392.
- Chapman, David. 2006. "Discourses of Multicultural Coexistence (*Tabunka Kyosei*) and the 'Old-Comer' Korean Residents of Japan." *Asian Ethnicity*, 7(1): 89–102. Doi:10.1080/14631360500498593.
- Chapman, David. 2020. "Gender and the Koseki." In *The Routledge Companion to Gender and Japanese Culture*, edited by Jennifer Coates, Lucy Fraser, Mark Pendleto, 83–91. London: Routledge.
- Chiavacci, David, and Carola Hommerich. 2017. Social Inequality in Post-Growth Japan: Transformation during Economic and Demographic Stagnation. London: Routledge.

- Choi, Kyuhwan, Bo Meng, and Timothy J. Lee. 2018. "An Investigation into the Segmentation of Japanese Traditional '*Ryokan*' Hotels Using Selection Attributes." *Journal of Vacation Marketing*, 24(4): 324–339.
- Chung, Erin Aeran. 2010. "Korea and Japan's multicultural models for immigrant incorporation." *Korea Observer*, 41(4): 649–676.
- Consulate General of Japan in New York. "A Tourism Boom." *Japan Information Center*, Summer 2015. https://www.ny.us.emb-japan.go.jp/en/c/2015/07-Jul/japaninfo-2015-07/01.html
- Cook, Emma E. 2018. "Power, Intimacy, and Irregular Employment in Japan." In *Intimate Japan*, edited by Allison Alexy, Emma E. Cook, 129–147. Honolulu: University of Hawaii Press.
- Cooper-White, Pamela. 2014. "Intersubjectivity." In Encyclopedia of Psychology and Religion, edited by David A. Leeming, 882–886. Boston: Springer.
- Dalton, Emma. 2015. Women and Politics in Contemporary Japan. Abingdon: Routledge.
- Dasgupta, Romit. 2011. "Emotional Spaces and Places of Salaryman Anxiety in Tokyo Sonata." *Japanese Studies*, 31(3): 373–386.
- Dasgupta, Romit. 2015. "Re-Imagining the 'post-Bubble' Family in Tokyo Sonata and Hush." In Configurations of Family in Contemporary Japan, edited by Tomoko Aoyama, Laura Dales, and Romit Dasgupta, 27–38. New York: Routledge.
- Demelius, Yoko. 2020. "Multiculturalism in a 'homogeneous' society from the perspectives of an intercultural event in Japan." *Asian Anthropology*, 19(3): 161–180.
- Demelius, Yoko. 2021. "Thinking through Community Spirit: *Zainichi* Koreans in Post-Korean Wave Japanese Communities." *Japanese Studies*, 41(1): 93–112.
- Ducanes, Geoffrey, and Manolo Abella. 2008. "Labour shortage responses in Japan, Korea, Singapore, Hong Kong, and Malaysia: a review and evaluation." *International Labor Organization*, Working Paper, 2. https://www.ilo.org/asia/publications/WCMS_099166/lang--en/index.htm
- Efendi, Ferry, Ching Min Chen, Anna Kurniati, Yuni Sufyanti Arief, and Reiko Ogawa. 2022. "The Course of Broken Dreams: The Expectations and Realities of the Life of Indonesian Nurses as Care Workers in Japan." *Collegian: The Australian Journal of Nursing Practice, Scholarship and Research*, 29(5): 680–687.
- Ehara, Yumiko. 2013. "Gender Studies in Sociology in Post-War Japan." *International Journal of Japanese Sociology* IJJS, 22(1): 94–103.

- Elgenius, Gabriella and Jens Rydgren. 2017. "The Sweden Democrats and the Ethno-Nationalist Rhetoric of Decay and Betrayal." *Sociologisk Forskning*, 54(4): 353–358.
- Elgenius, Gabriella and Jens Rydgren. 2022. "Nationalism and the Politics of Nostalgia." *Sociological Forum*, 37(S1): 1230–1243.
- Emerson, Robert Wall. 2015. "Convenience Sampling, Random Sampling, and Snowball Sampling: How Does Sampling Affect the Validity of Research?" *Journal of Visual Impairment & Blindness*, 109(2): 164–168.
- Etikan, Ilker, Sulaiman Abubakar Musa, and Rukayya Sunusi Alkassim. 2016. "Comparison of Convenience Sampling and Purposive Sampling." *American Journal of Theoretical and Applied Statistics*, 5(1): 1–4. Doi: 10.11648/j.ajtas.20160501.11.
- Eto, Mikiko. 2005. "Women's Movements in Japan: The Intersection Between Everyday Life and Politics." *Japan Forum*, 17(3): 311–333.
- EU-Japan Centre for Industrial Cooperation. "Geography & Culture." *EU Business in Japan*. July 2022. https://www.eubusinessinjapan.eu/about-japan/background/geography-culture
- EU-Japan Centre for Industrial Cooperation. "The Customer is God." *EU Business in Japan*. https://www.eubusinessinjapan.eu/about-japan/business-culture/customer-god
- European Institute for Gender Equality. "EIGE-2021 Gender Equality Index 2021 Report: Health." *EIGE 2021*. https://eige.europa.eu/publications/gender-equality-index-2021-report/gender-differences-household-chores
- Ferrarini, Benno. 2012. "International trade, domestic market potential and income in developing Asia." *Journal of the Asia Pacific Economy*, 17(1): 127–137.
- Finnish Institute for Health and Welfare. "Work and income." March 12, 2023. https://thl.fi/en/web/gender-equality/gender-equality-in-finland/work-and-income
- Flowers, Petrice. 2012. "From *Kokusaika* to *Tabunka Kyōsei*: Global Norms, Discourses of Difference, and Multiculturalism in Japan." *Critical Asian Studies*, 44(4): 515–542.
- Fukai, Taiyo, Hidehiko Ichimura, and Keisuke Kawata. 2021. "Describing the impacts of COVID-19 on the labor market in Japan until June 2020." *The Japanese Economic Review*, 72: 439–470. Doi:10.1007/s42973-021-00081-z
- Gibbs, Graham R. 2007. Analyzing Qualitative Data. Los Angeles: Sage.
- Gibson, William J., and Andrew Brown. 2009. *Working with Qualitative Data*. Los Angeles: Sage.
- Goldstein-Gidoni, Ofra. 1999. "Kimono and the Construction of Gendered and Cultural Identities." *Ethnology*, 38(4): 351–370.

- Goldstein-Gidoni, Ofra. 2005. "The Production and Consumption of 'Japanese Culture' in the Global Cultural Market." *Journal of Consumer Culture*, 5(2): 155–179.
- Goldstein-Gidoni, Ofra. 2019. "The Japanese Corporate Family: The Marital Gender Contract Facing New Challenges." *Journal of Family Issues*, 40(7): 835–864.
- Gottlieb, Nanette. 2005. *Language and society in Japan*. Cambridge: Cambridge University Press.
- Graburn, Nelson H, John Ertl, and R. Kenji Tierney. 2008. *Multiculturalism in the New Japan: Crossing the Boundaries Within*. New York: Berghahn Books.
- Grainge, Paul. 1999. "Reclaiming heritage: Colourization, culture wars and the politics of nostalgia." *Cultural Studies*,13(4): 621–638.
- Green, David. 2021. "Charting *Tabunka Kyōsei*: An Assessment of Municipal-Level 'Multicultural Coexistence' and Immigrant Integration Efforts in Japan." *Journal of Asian Sociology*, 50(2): 401–429.
- Haines, David W., Keiko Yamanaka, and Shinji Yamashita. 2012. "Wind over Water:

 Migration in an East Asian Context". Foundations in Asia Pacific Studies, New York:

 Berghahn Books.
- Hambleton, Alexandra. 2011. "Reinforcing Identities? Non-Japanese Residents, Television and Cultural Nationalism in Japan." *Contemporary Japan*, 23(1): 27–47.
- Hammine, Madoka, and Nathanael Rudolph. 2022. "Perceiving and problematizing 'invisibility' in English language education and criticality: a duoethnographic dialogue." *Asian Englishes*, 1–15.
- Hara, Hiromi. 2022. "The Gender Wage Gap in Japan—the Glass Ceiling Phenomenon." *CESifo Forum*, 23(2): 36–40.
- Hatase, Mariko, and Yoichi Matsubayashi. 2019. "Does Government Promote or Hinder Capital Accumulation? Evidence from Japan's High-Growth Era." *Structural Change and Economic Dynamics*, 49: 245–265.
- Hirano, Yuko O., Kunio Tsubota, and Shun Ohno. 2020. "Factors associated with the recruitment of foreign nurses in Japan: a nationwide study of hospitals." *Human Resources for Health*, 18(1): 88–88.
- Ikkatai, Yuko, Atsushi Inoue, Kei Kano, Azusa Minamizaki, Euan McKay, and Hiromi M. Yokoyama. 2019. "Parental Egalitarian Attitudes Towards Gender Roles Affect Agreement on Girls Taking STEM Fields at University in Japan." *International Journal of Science Education*, 41(16): 2254–2270.

- Ishibashi, Mari. "Women bear brunt of Japan's pandemic job losses." *Nikkei Asia*. September 17, 2020. https://asia.nikkei.com/Spotlight/Datawatch/Women-bear-brunt-of-Japan-s-pandemic-job-losses
- Ishiwata, Eric. 2011. "'Probably Impossible': Multiculturalism and Pluralisation in Present-Day Japan." *Journal of Ethnic and Migration Studies*, 37(10), 1605–1626.
- Itoh, Mayumi. 1996. "Economic Myths Explained: Japan's Abiding Sakoku Mentality." *Orbis*, 40(2): 235–245.
- Iwabuchi, Koichi. 2015. "Pop-culture diplomacy in Japan: soft power, nation branding and the question of 'international cultural exchange'." *International Journal of Cultural Policy*, 21(4): 419–432.
- Iwabuchi, Koichi. 2010. "Tabunka shakai nihon ni okeru 'bunka' no toi" [The Question of "Culture" in Japan as a Multicultural Society]. In *Tabunka shakai no 'bunka' o tou* [Questioning "Culture" in a Multicultural Society]. Tokyo: Seikyūsha, 9–34.
- Iwabuchi, Koichi, Hyun Mee Kim, and Hsiao-Chuan Hsia. 2016. *Multiculturalism in East Asia: A Transnational Exploration of Japan, South Korea and Taiwan*. New York: Rowman & Littlefield International, Ltd.
- Japan National Tourism Organization. "Omotenashi." *Japan National Tourism Organization*. https://www.japan.travel/en/au/experience/culture/omotenashi/
- Jeong, Jaehyeon. 2022. "*Nihonjin* after 3.11: The Construction of Japaneseness in Times of National Crisis." *Continuum*, 1–18.
- JIJI. "Over 80% in Japan believe women burdened by childcare." *The Japan Times*. March 14, 2023. https://www.japantimes.co.jp/news/2023/03/14/national/social-issues/japan-child-care-burdensome-polls/
- Joseph, John Earl. 2006. Language and politics. Edinburgh: Edinburgh University Press.
- Kamise, Yumiko. 2013. "Occupational Stigma and Coping Strategies of Women Engaged in the Commercial Sex Industry: A Study on the Perception of 'Kyaba-Cula Hostesses' in Japan." *Sex Roles*, 69(1-2): 42–57.
- Kang, Jean S. 2017. "Evaluating Labor Force Participation of Women in Japan and Korea: Developments and Future Prospects." *Asian Journal of Women's Studies*, 23(3): 294–320.
- Karakawa, Ryuchi. 2019. "The Effects of Japanese Ryokan Attributes on Perceived Values and Purchase Intention." University of Central Florida: Electronic Theses and Dissertations. https://core.ac.uk/download/pdf/236312196.pdf

- Kawaguchi, Daiji, and Soohyung Lee. 2017. "Brides for sale: Cross-border marriages and female immigration." *Economic Inquiry*, 55(2): 633–654.
- Kim, Hyuk-Rae, and Ingyu Oh. 2011. "Migration and Multicultural Contention in East Asia." *Journal of Ethnic and Migration Studies*, 37(10): 1563–1581.
- Kim, Viktoriya, and Philip Streich 2020. "*Tabunka Kyōsei* without immigration policy: The role of centers for international exchange and their challenges." *Contemporary Japan*, 32(2): 174–196.
- Kingston, Jeff. 2019. Critical Issues in Contemporary Japan. New York: Routledge.
- Kobayashi, Tomoyuki, Masaharu Maeda, Yui Takebayashi, and Hideki Sato. 2021. "Traditional Gender Differences Create Gaps in the Effect of Covid-19 on Psychological Distress of Japanese Workers." *International Journal of Environmental Research and Public Health*, 18(16). Doi:10.3390/ijerph18168656.
- Kodama, Naomi, and Izumi Yokoyama. 2013. "Women's Labor Supply and Taxation—
 Analysis of the Current Situation Using Data—." *Policy Research Institute, Ministry of Finance, Japan, Public Policy Review*, 14(2): 267–300.

 https://www.mof.go.jp/english/pri/publication/pp_review/fy2017/ppr14_02_03.pdf
- Kohara, Miki, and Bipasha Maity. 2021. "The Impact of Work-Life Balance Policies on the Time Allocation and Fertility Preference of Japanese Women." *Journal of the Japanese and International Economies*, 60: 1–14.
- Kondoh, Kenji. 2020. "A Paradoxical Immigration Restriction Policy for Unskilled Illegal Immigrants." *Asia-Pacific Journal of Regional Science*, 4(2): 479–497.
- Kotera, Shinya, and Jochen M. Schmittmann. 2022. "The Japanese Labor Market During the COVID-19 Pandemic." *International Monetary Fund*. May 13, 2022. https://www.imf.org/en/Publications/WP/Issues/2022/05/13/The-Japanese-Labor-Market-During-the-COVID-19-Pandemic-517840
- Kowner, and Walter Demel. 2015. Race and Racism in Modern East Asia: Interactions, Nationalism, Gender and Lineage. Leiden: Brill.
- Koyama, Shizuko. 2013. *Ryōsai Kenbo: The Educational Ideal of "Good Wife, Wise Mother"* in Modern Japan. Boston: Brill.
- Krishnan, Ramesh. "Where do Highly-Skilled Foreigners Work in Japan? An Industry Overview." *JPort Journal*. November 28, 2019. https://jportjournal.com/article/987
- Kumamoto Nichinichi Shinbun. "Shimin teigi ni gaikokujin tsuika, hantai iken sattō

 Kumamoto-shi jichi jōrei kaisei-an ni 2400-ken 'senkyo-ken ataeru' gokai mo netto de
 kakusan [Foreigners added to the definition of "citizen", flood of dissenting opinions

- Kumamoto City Autonomy Ordinance Amendment Proposed 2,400 cases "Give voting rights" Misunderstanding spreads on the Internet]. *Kumamoto Nichinichi Shinbun*. January 20, 2023. https://kumanichi.com/articles/921575
- KYODO. "Just two women picked as Kishida reshuffles Cabinet." *The Japan Times*. August 11, 2022. html. https://www.japantimes.co.jp/news/2022/08/11/national/politics-diplomacy/cabinet-two-women/
- KYODO. "Fewer women elected in Lower House poll despite empowerment law." *The Japan Times*. November 1, 2021. https://www.japantimes.co.jp/news/2021/11/01/national/politics-diplomacy/electionwomen/
- Lee, Kristen Schultz. 2010. "Gender, Care work, and the complexity of family membership in Japan." *Gender & Society*, 24(5): 647–671.
- Lee, Soo-im, Stephen Murphy-Shigematsu, and Harumi Befu. 2006. "Japan's Diversity Dilemmas: Ethnicity, Citizenship, and Education." New York: iUniverse.
- Liddicoat, Anthony J. 2007. "Internationalising Japan: Nihonjinron and the Intercultural in Japanese Language-in-education Policy." *Journal of Multicultural Discourses*. 2(1): 32–46.
- Lie, John 2001. Multiethnic Japan. Harvard University Press. Doi:10.4159/9780674040175
- Linton, April. 2009. "Language politics and policy in the United States: implications for the immigration debate: Multilingualism and Language Politics: New Challenges."

 International Journal of the Sociology of Language, 199: 9–37.
- Liu-Farrer, Gracia. 2011. Labour migration from China to Japan: International students, transnational migrants. London: Routledge.
- Liu-Farrer, Gracia. 2020. *Immigrant Japan: Mobility and Belonging in an Ethno-Nationalist Society*. Ithaca: Cornell University Press.
- Lu, Hangyan and Warren A Hodge. 2019. "Toward Multi-Dimensional and Developmental Notion of Researcher Positionality." *Qualitative Research Journal*, 19(3): 225–235.
- Mackie, Vera C. 2003. Feminism in modern Japan: citizenship, embodiment, and sexuality. Cambridge: Cambridge University Press.
- Margolis, Eric. "Japan Doesn't Want to Become Another Casualty of English." *Foreign Policy*. May 26, 2020. https://foreignpolicy.com/2020/05/26/japan-doesnt-want-to-become-another-casualty-of-english/
- Marshall, Robert. 2017. "Gender inequality and family formation in Japan." *Asian Anthropology*, 16(4): 261–278.

- Matsui, Kathy, Hiromi Suzuki, Kazunori Tatebe and Tsumugi Akiba. 2014. "Womenomics 4.0: Time to Walk the Talk." *Goldman Sachs*. https://www.gspublishing.com/content/research/en/reports/2014/05/29/163af1fc-35c0-47f6-b6bc-c3a9e378e020.pdf
- McKinley, Jim. 2017. "Overcoming Problematic Positionality and Researcher Objectivity." In *Doing Research in Applied Linguistics*, edited by Jim McKinley and Heath Rose, 37–46. Abingdon: Routledge.
- Ministry of Foreign Affairs of Japan. "Highly skilled professional visa." May 15, 2023. https://www.mofa.go.jp/j info/visit/visa/long/visa16.html
- Ministry of Health, Labor and Welfare. "Summary of notification status of foreigner employment status (as of the end of October 2017)." January 2018. https://www.mhlw.go.jp/stf/houdou/0000192073.html
- Miura, Junpei. "Survey: 40% of working wives try to earn less for tax reasons". *Asahi Shimbun*. December 14, 2022. https://www.asahi.com/ajw/articles/14792427>
- Miura, Mari. 2012. Welfare through Work: Conservative Ideas, Partisan Dynamics, and Social Protection in Japan. Ithaca: Cornell University Press.
- Mizukoshi, Kosuke, Florian Kohlbacher, and Christoph Schimkowsky. 2016. "Japan's Ikumen Discourse: Macro and Micro Perspectives on Modern Fatherhood." *Japan Forum*, 28(2): 212–232.
- Merriam-Webster Dictionary. "Sample." *Merriam-Webster.com*. 2023. https://www.merriam-webster.com/dictionary/sample
- Mohanty, Ajit K. 2010. "Languages, Inequality and Marginalization: Implications of the Double Divide in Indian Multilingualism." *International Journal of the Sociology of Language*, 205: 131–154.
- Moore, Jenny. 2012. "A Personal Insight into Researcher Positionality." *Nurse Researcher*, 19(4): 11–14.
- Morishita, Shunichiro. 2016. "Managing *omotenashi* in onsen ryokans: A case study of Kurokawa Onsen in Kyushu, Japan." *Journal of Global Tourism Research*, 1(2): 157–160. https://www.jstage.jst.go.jp/article/jgtr/1/2/1_157/_pdf
- Morita, Liang. 2018. "Does Doing Things the Japanese Way Attract Highly-Skilled Migrants?" *Cogent Social Sciences*, 4(1): 1–12.
- Morita, Liang. 2023. "Japan's Technical Intern Training Program (TITP) to be Abolished." SSRN Electric Journal. Doi:10.2139/ssrn.4417548

- Moriyasu, Ken. "75% of Japanese women not interested in management". *Nikkei Asia*. May 14, 2015. https://asia.nikkei.com/NAR/Articles/75-of-Japanese-women-not-interested-in-management
- Morris-Suzuki, Tessa. 2015a. "Beyond Racism: Semi-Citizenship and Marginality in Modern Japan." *Japanese Studies*, 35(1): 67–84.
- Morris-Suzuki, Tessa. 2015b. *Re-inventing Japan: Nation, Culture, Identity: Nation, Culture, Identity.* New York: Routledge.
- Mosedale, Jan. 2011. *Political Economy of Tourism: A Critical Perspective*. London: Routledge.
- Munia, Rafael. 2015. "The Limits of Multiculturalism In Japan: In Defense Of Interculturalism And Nomadic Assemblage." *Ponto Urbe*, 16.
- Muta, Kazue. 2009. "Senryaku to shite no onna Meiji Taishō no onna no gensetsu o megutte" [Woman as Strategy—Concerning Meiji and Taisho period discourse on women]. In *Seiyakuwari* [Gender Roles], *nihon no feminizumu* 3 [The history of gender roles: Japan's Feminism 3]: 45–70. Iwanami Shoten.
- Nagase, Nobuko. 2018. "Has Abe's Womanomics Worked?" *Asian Economic Policy Review*, 13(1): 68–101. https://onlinelibrary.wiley.com/doi/full/10.1111/aepr.12202
- Nakamatsu, Tomoko. 2014. "Under the Multicultural Flag: Japan's Ambiguous Multicultural Framework and its Local Evaluations and Practices." *Journal of Ethnic and Migration Studies*, 40(1): 137–154.
- Narayan, Kirin. 2003. "How native is a 'native' anthropologist?" In *Feminist Postcolonial Theory: A Reader*, edited by Reina Lewis and Sara Mills, 285–305. Edinburgh: Edinburgh University Press.
- Nemoto, Kumiko 2016. *Too Few Women at the Top: The Persistence of Inequality in Japan.*Ithaca: Cornell University Press.
- Nederveen Pieterse, Jan, and Jongtae Kim. 2012. *Globalization and Development in East Asia*. London: Routledge.
- Nguyen, Anh Thi Ngoc. 2020. "Japan's Inbound Tourism Boom: Lessons for its Post-COVID-19 Revival." *International Monetary Fund*. August 21, 2020. https://www.imf.org/en/Publications/WP/Issues/2020/08/21/Japans-Inbound-Tourism-Boom-Lessons-for-its-Post-COVID-19-Revival-49618
- Noguchi, Sharon. 2007. "Finding Home: Immigrant Life in Japan." *Asia Pacific Journal*, 5(2). https://apjjf.org/-Sharon-Noguchi/2349/article.html

- OECD. 2021. "Gender wage gap." Organization for Economic Co-operation and Development. http://www.oecd.org/gender/data/gender-wage-gap.htm
- Ogasawara, Yuko. 2016. "The Gender Triad: Men, Women, and Corporations." In *Power in Contemporary Japan*, edited by Gill Steel, 167–182. New York: Palgrave Macmillan US.
- Oguma, Eiji. 1995. *Tan'itsu minzoku shinwa no kigen: "Nihonjin" no jigazō no keifu* [The myth of the homogeneous nation: the genealogy of Japan's self image]. Tōkyō: Shin'yōsha.
- Ohtsu Makoto, and Tomio Imanari. 2015. *Inside Japanese business: a narrative history,* 1960-2000. London: Routledge.
- O'Neill, Patrick Geoffrey. 2015. *Japanese Respect Language: When, Why, and How to Use It Successfully: Learn Japanese Grammar, Vocabulary & Polite Phrases With This User-Friendly Guide*. North Clarendon: Tuttle Publishing.
- Osaki, Tomohiro. "Japan's male babysitters battle stigma and stereotypes", *The Japan Times*, November 15, 2020. https://www.japantimes.co.jp/news/2020/11/15/national/socialissues/male-babysitters-japan/
- Osawa, Machiko, Myoung Jung Kim, and Jeff Kingston. 2013. "Precarious Work in Japan." *The American Behavioral Scientist*, 57(3): 309–334.
- Ozturk, Ayse, Eric Joiner, and S. Tamer Cavusgil. 2015. "Delineating Foreign Market Potential: A Tool for International Market Selection." *Thunderbird International Business Review*, 57(2): 119–141.
- Paillard-Borg, Stéphanie and Jessica Holmgren. 2016. "Immigration, Women, and Japan—A Leap Ahead and a Step Behind: A Qualitative Journalistic Approach." *SAGE Open*, 6(4).
- Phillips, Anne 2007. Multiculturalism without Culture. Princeton: Princeton University Press.
- Parreñas, Rhacel Salazar. 2012. "The Reproductive Labour of Migrant Workers." *Global Networks*, 12(2): 269–275.
- Robson, Colin. 2002. Real World Research: A Resource for Social Scientists and Practitioner-Researchers. Oxford: Blackwell Publishing.
- Roth, Joshua Hotaka. 2017. "Responsibility and the Limits of Identification: Fieldwork Among Japanese and Japanese Brazilian Workers in Japan." In *Doing Fieldwork in Japan*, edited by Theodore C. Bestor, Patricia G. Steinhoff and Victoria Lyon Bestor, 335–351. Honolulu: University of Hawaii Press.

- Ryall, Julian. "Japan visa changes could lead to residency for workers from Philippines,
 Indonesia, Vietnam. But will they be welcome?" *South China Morning Post*. April 26,
 2023. https://www.scmp.com/week-asia/people/article/3218328/japan-visa-changescould-lead-residency-workers-philippines-indonesia-vietnam-will-they-be-welcome
- Saldaña, Johnny. 2009. *The Coding Manual for Qualitative Researchers*. Thousand Oaks: SAGE Publications. https://www.sagepub.com/sites/default/files/upm-binaries/24614_01_Saldana_Ch_01.pdf
- Schoppa, Leonard James. 2006. Race for the Exits: The Unraveling of Japan's System of Social Protection. Ithaca: Cornell University Press.
- Schreiber, Mark. "Has Christel Takigawa cursed Japan with her 2020 Olympics *omotenashi* pitch?" *The Japan Times*, February 3, 2018.

 https://www.japantimes.co.jp/news/2018/02/03/national/media-national/christel-takigawa-cursed-japan-2020-olympics-omotenashi-pitch/
- Sechiyama, Kaku. 2013. *Patriarchy in East Asia a Comparative Sociology of Gender*. Leiden: Brill.
- Shani, Giorgio. 2019. "Consuming the Nihonjinron." *Nations and Nationalism*, 25(4): 1119–1121.
- Shinchi Arihiro, "Nihon ga uri ni suru 'omotenashi bunka', naze gaikokujinkankōkyaku ni hibikanai no ka" [Why doesn't Japan's "hospitality culture" appeal to foreign tourists?]." Bengoshi Nyuushu [Lawyer News]. February 26, 2023. http://bengo4.com/c 18/n 15681/
- Shigematsu, Setsu. 2012. Scream from the Shadows: The Women's Liberation Movement in Japan. Minneapolis: University of Minnesota Press.
- Shiobara, Yoshikazu. 2020. "Genealogy of *tabunka kyōsei*: A Critical Analysis of the Reformation of the Multicultural Co-living Discourse in Japan." *International Journal of Japanese Sociology*, 29(1): 22–38.
- Shiobara, Yoshikazu, Kohei Kawabata, and Joel Matthews. 2020. *Cultural and Social Division in Contemporary Japan: Rethinking Discourses of Inclusion and Exclusion*. Abingdon: Routledge.
- Smeekes, Anouk, and Maykel Verkuyten. 2015. "The presence of the past: Identity continuity and group dynamics." *European Review of Social Psychology*, 26(1): 162–202.
- Smeekes, Anouk, and Jolanda Jetten 2019. "Longing for One's Home Country: National Nostalgia and Acculturation Among Immigrants and Natives." *International Journal of Intercultural Relations*, 69: 131–150.

- Smeekes, Anouk, Tim Wildschut, and Constantine Sedikides. 2021. "Longing for the 'good Old Days' of Our Country: National Nostalgia as a New Master frame of Populist Radical Right Parties." *Journal of Theoretical Social Psychology*, 5(2): 90–102.
- Sony Life Insurance Company. "Josei no katsuyaku ni kansuru ishiki chōsa 2022." [Awareness survey on women's empowerment 2022]. November 1, 2022. https://www.sonylife.co.jp/company/news/2022/nr_221101.html#sec1
- Statistics Bureau of Japan. 2021a. "Basic Complete Tabulation on Population and Households of the 2020 Population Census of Japan."

 https://www.stat.go.jp/english/info/news/20211228.html
- Statistics Bureau of Japan. 2021b. "Survey on Time Use and Leisure Activities" https://www.stat.go.jp/english/data/shakai/2021/pdf/timeuse-a2021.pdf
- Statistics Bureau of Japan 2022. "Statistical Handbook of Japan 2022." https://www.stat.go.jp/english/data/handbook/index.html
- Steinberg, Chad, and Masato Nakane. 2012. "Can Women Save Japan?" Washington, DC: International Monetary Fund.
- Suen, Lee-Jen Wu, Hui-Man Huang, and Hao-Hsien Lee. 2014. "A comparison of convenience sampling and purposive sampling." *Hu li za zhi*, 61(3): 105–111.
- Sugimoto, Yoshio. 1999. "Making Sense of Nihonjinron." *Thesis Eleven*, 57(1): 81–96.
- Sugimoto, Yoshio. 2014. *An Introduction to Japanese Society*. Cambridge: Cambridge University Press.
- Sugiura, Eri. "Japan gets more than it bargained for with tourist boom", *Nikkei Asia*. April 17, 2019. https://asia.nikkei.com/Spotlight/The-Big-Story/Japan-gets-more-than-it-bargained-for-with-tourist-boom
- Świtek, Beata. 2016. *Reluctant Intimacies: Japanese Eldercare in Indonesian Hands*. New York: Berghahn Books.
- Taira, Kazufumi. 2019. "A Systematic Form of Japanization in Okinawa: Japanese Identity Construction through a Japanization Discourse in Textbooks and a Textbook Guide." *International Studies in Sociology of Education*, 28(2): 110–126.
- Takahashi, Koji 2021. "Accommodation and Food Services Workers amid the COVID-19 Crisis in Japan." *Japan Labour Issues*, 5(33):13–20. https://www.jil.go.jp/english/jli/documents/2021/033-04.pdf
- Takahashi Yoshiaki. 2020. "Are Foreign Residents of Japan Happy? Clues from a Local Government Survey." *Asia-Pacific Review*, 27(2): 130–142.

- Takeda, Tomoya, Yuki Miyamoto, Xiaodan Lu, Kayo Okuhira, Noriyuki Kida and Tomoko Ota. 2016. "Eye Movement Analysis for Expert and Non-Expert in Japanese Traditional Culture of Tea Ceremony—From the View Point of Japanese Hospitality, 'Omotenashi.'" In Digital Human Modeling: Applications in Health, Safety, Ergonomics and Risk Management, edited by Vincent G. Duffy, 596–603. Cham: Springer International Publishing.
- Tauchi, Kosuke. "Japan proposes to end problem-filled technical intern program" *Asahi Shimbun*, April 10, 2023. https://www.asahi.com/ajw/articles/14882268
- Tilastokeskus. "Kotitöiden ja hoivavastuun jakautuminen." *Terveyden ja hyvinvoinnin laitos*, November 17, 2022. https://thl.fi/fi/web/sukupuolten-tasa-arvo/tasa-arvon-tila/perheet-ja-vanhemmuus/kotitoiden-ja-hoivavastuun-jakautuminen
- Tipton, Elise K. 2015. Modern Japan: A Social and Political History. Florence: Routledge.
- Townsend, Susan C. 2013. "The 'miracle' of car ownership in Japan's 'Era of High Growth', 1955-73." *Business History*, 55(3): 498–523.
- Tsuda, Takeyuki. 1999. "The Permanence of 'Temporary' Migration: The 'Structural Embeddedness' of Japanese-Brazilian Immigrant Workers in Japan." *The Journal of Asian Studies*, 58(3): 687–722.
- Tsutsui, Junya. 2020. "Work and Family in Japanese Society." Singapore: Springer Singapore.
- Ueno, Chizuko. 2021. "Why Do Japanese Women Suffer from the Low Status?: The Impact of Neo-Liberalist Reform on Gender." *Japanese Political Economy*, 47(1): 9–26.
- Valaskivi, Katja. 2013. "A brand new future? Cool Japan and the social imaginary of the branded nation." *Japan Forum*, 25(4): 485–504.
- Valk, Julie 2018. "From Duty to Fashion: The Changing Role of the Kimono in the Twenty-First Century." *Fashion Theory*, 22(3): 309–340.
- Vogt, Gabriele 2015. "Foreign Workers in Japan." In *The SAGE Handbook of Modern Japanese Studies*, edited by James Babb, 567–582. Doi:10.4135/9781473909908.n32
- Vosko, Leah F., Martha MacDonald, and Iain Campbell. 2009. *Gender and the Contours of Precarious Employment*. New York: Routledge.
- White, Linda. 2018. Gender and the Koseki in Contemporary Japan. London: Routledge.
- White, Stanley. "Japan invests in service industry, reshaping its legendary hospitality" *Reuters*, November 5, 2018. https://www.reuters.com/article/us-japan-economy-capex-services-idUKKCN1N90RH

- Willis, David Blake, and Stephen Murphy-Shigematsu. 2008. *Transcultural Japan at the Borderlands of Race, Gender, and Identity*. London: Routledge.
- The World Bank. 2020. "Fertility rate, total (births per woman)." https://data.worldbank.org/indicator/SP.DYN.TFRT.IN
- The World Economic Forum. 2022. "Global Gender Gap Report 2022." https://www.weforum.org/reports/global-gender-gap-report-2022/in-full/1-benchmarking-gender-gaps-2022/
- The World Economic Forum. 2022. "Interactive Data and Economy Profiles." *Travel & Tourism Development Index 2021: Rebuilding for a Sustainable and Resilient Future*. May 24, 2022. https://www.weforum.org/reports/travel-and-tourism-development-index-2021/explore-the-data
- Yamada, Masaki. "Learning the Spirit of *Omotenashi*." *Public Relations Office*. March 2012. https://www.gov-online.go.jp/eng/publicity/book/hlj/html/201203/201203_04.html
- Yamaguchi, Kazuo. 2019. "Japan's Gender Gap." Finance & Development, 56(1): 26–29.
- Yamamoto, Kana. 2015. "The myth of '*Nihonjinron*', homogeneity of Japan and its influence on the society." *CERS Working Paper*. https://cers.leeds.ac.uk/wp-content/uploads/sites/97/2016/04/The-myth-of-%E2%80%9CNihonjinron%E2%80%9D-homogeneity-of-Japan-and-its-influence-on-the-society-Kana-Yamamoto.pdf
- Yoshida, Akiko. 2011. "No chance for romance: Corporate culture, gendered work, and increased singlehood in Japan." *Contemporary Japan*, 23(2): 213–234.
- Yoshida, Mitsuhiro. 2001. "Joking, Gender, Power, and Professionalism Among Japanese Inn Workers." *Ethnology*, *XL*(4): 361–369.
- Yu, Wei-Hsin. 2009. *Gendered Trajectories: Women, Work, and Social Change in Japan and Taiwan*. Stanford: Stanford University Press.
- Zhou, Yanfei. 2015. "Career Interruption of Japanese Women: Why Is It So Hard to Balance Work and Childcare?" *The Japan Institute for Labour Policy and Training*. https://www.jil.go.jp/english/JLR/documents/2015/JLR46_zhou.pdf
- United Nations. 2017. "World Population Aging."

 https://www.un.org/en/development/desa/population/publications/pdf/ageing/WPA201

 7 Highlights.pdf, accessed December 1, 2020.

Appendices

Appendix 1 Consent Form for Interview and Research Cooperation

Consent Form for Interview and Research Cooperation

This interview will be conducted as follows. If you understand the purpose and content of the interview and are willing to participate to the interview, please sign the consent form below.

1. The Purpose of the interview

This research will be carried out as part of the thesis assignment for the master's degree in East Asian Studies at the University of Turku. The purpose of this study is to better understand the daily lives and experiences of foreign women working in Japan's service sector. The interview includes questions about work, entering Japan, the hospitality business, daily life, etc., but mainly what kind of experience you had as a woman and as a foreigner in Japan. The insights gained from this study will benefit the conditions and rights of foreign workers in Japan in the future. For this reason, the contributions of the participants are highly valued. We apologize for any inconvenience and appreciate your cooperation.

2. Interview method

In this survey, we will conduct at least two 60-minute interviews. The content of the interview will be recorded with ZOOM, and the content will be transcribed and analyzed. Your permission to record is asked before the interview. The interview will be conducted online by ZOOM, but if it does not work for you, it is possible to use other methods. The link for the online meeting is sent to you before the interview via email. Interviews will be conducted in Japanese or English, depending on the wishes of the information provider. If the interview is conducted in Japanese, the interview is more likely to be recorded in consideration of the researcher's Japanese proficiency. The interview is a structured interview and contains open-ended questions. The informant can decline to answer any question and skip the answer at his or her own will.

We are planning to hold only one round of interviews, but there may be questions that the researcher would like to ask again during the analysis process. In that case, we may ask for additional interviews. The interview period is scheduled to end on December 31st, 2022.

3. Participation and withdrawal from interviews

Participation in this interview is voluntary. You may refuse the interview or withdraw your participation in the interview at any time. In addition, the collaborator (interviewed person) will not be treated disadvantageously in situations where they refuse to take part in the interview, withdraw participation or refuse to answer any question.

4. Data management

The collected data and personal information will be managed with due care and will not be used for purposes other than writing the master's thesis and related assignments. Pseudonyms will be used in the thesis and presentations so that the individuals who participated to this study cannot be identified. Everyone's rights are respected.

5. Publication of interview results

Based on the data obtained from the interview, the researcher writes a master's degree thesis and submit it to the person in charge of the degree. The thesis will be made available online with restrictions and all participants will be anonymized.

check the transcript of the interview or the recording of the intervie

If you wish, you can check the transcript of the interview, or the recording of the interview at any point you wish.
Transcription of interview content and audio confirmation (preferred / not wished):
In addition, regarding the completed master's thesis, we will explain the contents of the it verbally if the participant wishes, so please let us know your wish.
b) Summary report of master's thesis (preferred / not wished)
How to send materials if you "prefer" any of the above $a) - b$)
□ E-mail:
□ Mail: ¬¬
6. Person in charge and contact information
The Researcher:
Email:

I understand the above and agree to cooperate with the interview and agree to the handling of data.
Date:
Collaborator's name:
Signature (can be waved):
Contact address (email or home address):
Research Conductor:
Affiliation:
Contact address:

Appendix 2 Interview Questions

Background information

- May I confirm your name?
- How old are you?
- What is your nationality?
- What is your education?
- What is your current profession?
- Are you married?
- How long have you stayed in Japan?

Immigration process

- Why did you decide to come to Japan?
- How did you enter the country? Was the procedure easy or difficult?
- Are there any memorable events regarding the immigration procedure?
- How long did it take to get a visa?
- Have you had to renew your visa?
- If you have had to renew the visa, how was your visit to the Immigration Bureau?
- Did anyone help you when you were trying to enter the country or get a visa?

Language

- Did you speak Japanese before coming to Japan? If so, were you already fluent?
- What did you think of your Japanese skills when you interacted with people?
- Where did you use Japanese language? How often?
- Do you think your language skills have helped you in Japan? In what ways did you find it useful?
- -Are there any language-related stories you would like to share? Do you have any other good or difficult language-related experiences in Japan?
- Have someone commented on your Japanese ability in Japan?

Service sector, hospitality business and working life

- -If I can ask you, what kind of work permit or contract do you have?
- How did you find the job?
- How was the application process/ job interview?
- What do you think of working at Japan's service sector?
- What were your usual tasks in your job?
- What did you like about your work?
- What was the most difficult aspect in your work? What was the thing you hated the most?
- -How long did you work at your job?
- Did you acquire any new skills?
- What did you learn from your work experience in Japan?
- What do you think of Japanese working style or work culture?
- Would the job you did be different in your home country compared to Japan? How so?
- Did you think you were well paid for your work?
- Was it normal to overwork in your job? Did you overwork yourself in your work?
- Would you recommend this job to other foreign workers?
- Have you done any other kind of work in Japan in the past?
- Do you have any memorable memories to share when you worked in Japan?

Life in Japan

- How was your life in Japan?
- Before you came to Japan, what did you expect your life in Japan to be like?
- How have your opinions about life in Japan changed since you started living in Japan?
- Did you have any problems while living in Japan? If you had, how did you solve them?
- Were you facing problems on a regular or irregular basis?
- Was your life in Japan different from the life in your hometown?
- Are there any events that left an impression on you after you came to Japan?

Leisure time

- -How did you spend your free time in Japan?
- Did you have any hobbies?
- -How did you spend your time in your home country? Did your leisure time activities differ from those back at home?

Community

- What kind of relationships you formed while in Japan?
- What is the nationality of those people you have befriended?
- How often and how did you keep in contact with your family abroad?
- What did your family think of you coming to Japan?
- How was your neighbourhood? What kind of neighbourhood was it?
- Did there live other foreign residents in the neighbourhood?
- How did the neighbourhood interact with its residents? What about the foreign nationals?
- What kind of services the municipality/local governments provided to its residents?
- How does the municipality help the foreign residents? For example, do they provide information in English or other languages, some language courses etc.?
- Have you taken part in any community or neighbourhood activity in the past? If you have, what kind of activity or event you have participated in? (Cleaning day etc.)

About women

- Were there a lot of women working at your job?
- Were there more women working than men?
- Do you think service sector in Japan have more women than men working there?
- Do you think it is easier for woman to perform or do service jobs?
- What do you think of women's working opportunities in Japan?
- Do you think there are certain expectations for women and men?

Miscellaneous:

- What kind of perceptions you had about Japan (as a country) before coming to Japan?
- Did they change after living there?
- Did you have anyone you could rely on if you needed help?
- Did you ever feel lonely? Why or why not?
- Did you ever feel powerless in Japan?

Future:

- What do you want to do in the future?
- Do you plan to return to Japan to work?